#### Central Agreement

between

## Caterpillar Inc.

and the

**UAW** 

March 1, 2023

Article 1 - P	Purpose	
1.1	Purpose	1
1.2	Asexual Clause	2
Article 2 - R	Recognition	
2.1	Bargaining Units	2
2.2	Scope of Agreement	4
Article 3 - U	Jnion Security	
3.1	Union Shop	4
3.2	Authorization for Dues Check-Off	5
3.3	Revoking Check-Off Authorization	7
Article 4 - R	Representation	
4.1	Union Representation	8
4.2	Abuse of Privileges	9
4.3	Pass for Union Representatives	9
4.4	Lists of Representatives	10
4.5	International Union Visit to Plant	10
4.6	Discussion of Grievance Disposition	11
4.7	Business Units	11
Article 5 - C	Grievance Procedure	
5.1	Definition of Grievance	12
5.2	Identification of Issue	12
5.3	No-Precedent Settlements	13
5.4	Limited Liability: Back Pay Offsets	14
5.5	Disciplinary Action	15
5.6	Written Warnings	16
5.7	Medical Disputes	16
5.8	Final Step Answers	17

Article	6 - Ar	bitration	
	6.1	Exchange of Statements:	
		Referral to Arbitration	17
	6.2	Permanent Arbitrator	18
	6.3	Order of Grievances Arbitrated	19
	6.4	Hearing Dates: Briefs	20
	6.5	Hearing Dockets	20
	6.6	Submissions	25
	6.7	Authority and Role of the Arbitrator	26
	6.8	Arbitration Expenses	28
	6.9	Expedited Arbitration	28
Article	7 - Ho	ours of Work and Overtime	
	7.1	Regular Weekly Schedule of Work	29
	7.2	Starting Times: Lunch: Brief Snacks	30
	7.3	Irregular Shifts and Schedules	31
	7.4	Night Shift Premium	32
	7.5	Report-In Pay	32
	7.6	Call-In Pay	32
	7.7	Time and One-Half	33
	7.8	Double Time: Holidays	33
	7.9	Overtime Determination	38
	7.10	Pyramiding Prohibited	38
	7.11	Overtime Allocation	39
	7.12	Notice for Required Overtime	39
	7.13	Required Overtime	39
	7.14	Notice to Union Representatives	
		of Overtime	41
	7.15	Paydays	41
	7.16	Definition of "Regular Scheduled"	41
Article	8 - He	ealth and Safety	
	8.1	Safety and Medical	42
	8.2	Local Union Safety Committees	43

	8.3	Safety Complaint Procedure	46
	8.4	Report of Physical Examination	49
	8.5	Excessive Employee Exposure	50
	8.6	Results of Environment Test	50
	8.7	Request for Central Committee	
		Investigation	50
	8.8	Safety Glasses	51
	8.9	Special Protective Garments: Devices	53
	8.10	Metatarsal Protection	53
	8.11	OSHA Walk-Around Pay	53
	8.12	Central Safety Committee	54
	8.13	Functions of the Central Safety	
		Committee	55
Artic	le 9 - Va	acation, Vacation Pay, Separation Pay	
	9.1	Vacation Payment	57
	9.2	Vacation Entitlements: Periods	57
	9.3	Vacation Period Exceptions	58
	9.4	Definition of "Base Period"	59
	9.5	Amount of Vacation Payment	60
	9.6	Separation Payment	61
	9.7	Dates of Vacation Payment	61
	9.8	Definition of "Service"	61
Artic	le 10 - S	Skilled Trades Groups	
	10.1	Skilled Trades Groups: Seniority	61
	10.2	Applicability of Articles 12 and 13	62
Artic	le 11 - S	Seniority	
	11.1	Probationary Period	62
	11.2	•	63
	11.3	Accumulation of Seniority	64
	11.4	Accumulation During Layoff	65
	11.5	Seniority Lists	65

	11.6	Waiver of Seniority Rules	66
	11.7	Steward Notification	66
A4: -1-	10 I-	h Onnaine	
Article		bb Openings  Floring of Markethale	
	12.1	Election of Methods	66
	12.2	$\varepsilon$	66
	12.3	Notice of Job Opening	67
	12.4	Job Bid: Eligibility	67
	12.5	Job Bid Restrictions-Recall/New Hire	68
	12.6	Job Bid Restrictions - Demotion	69
	12.7	Job Bid Restrictions Waiver	69
	12.8	$\varepsilon$	70
	12.9	Job Bid	70
	12.10	Post/Bid Procedure	71
	12.11	Reassignment	74
	12.12	Canvass	
	12.13	New Hire	74
	12.14	Notification of Openings Filled	75
	12.15	Preferential Hire	75
	12.16	Definition: Facilities	77
Article	213 - Re	eduction In Force and Layoff	
	13.1	Definition of Layoff	77
	13.2	Temporary Layoff	78
	13.3	Notice of Layoff	78
	13.4	Definitions	78
	13.5	Procedure for Reduction in Force	79
	13.6	Temporary Layoff - Inventory	84
	13.7	Union Representatives	84
	13.8	Layoff Payments	85
Article	214 - Le	eaves of Absence	
	14.1	Medical Leave of Absence	85
	14.2	Return from Leave	86

	14.3	Failure to Return from Leave	87
	14.4	Personal Leave	87
	14.5	Verification of Leave	87
	14.6	Military Leave	88
	14.7	Peace Corps Leave	88
	14.8	Pregnancy Leave	88
	14.9	Union, Governmental Leaves	89
	14.10		89
	14.11	Family Leave	90
Article	e 15 - Sa	alary Plan	
	15.1	Paid Absence Allowance	91
	15.2	Holiday Pay	97
	15.3	Jury and Witness Service	100
	15.4	Temporary Military Service	102
	15.5	Bereavement	104
	15.6	Parental Leave	105
	15.7	General	106
Article	e 16 - G	eneral	
	16.1	Supervision Working	107
	16.2	Discrimination Prohibition	109
	16.3	Savings Clause	109
	16.4	Bulletin Boards	110
	16.5	No-Strike: No-Lockout	110
	16.6	Moving Allowance	111
	16.7	Addresses of Employees	112
	16.8	Required Physical Examinations	112
	16.9	Occupational Injury: Pay for Shift	113
	16.10	Contract Work	113
	16.11	Off-Shift Classes	134

Articl	le 17 - T	raining	
	17.1	Definitions	134
	17.2	Plant Joint Apprenticeship	
		Committee	136
	17.3	Central Joint Apprenticeship	
		Committee	140
	17.4	Application for Apprenticeship	142
	17.5	Eligibility Requirements	142
	17.6	Entry into an Apprenticeship	143
	17.7	Terms of Apprenticeship	144
	17.8	Apprentice Reductions and Returns	144
	17.9	Wages	146
	17.10	Supervision of Apprentices	146
	17.11	Tools	146
	17.12	Placement Upon Graduation	147
	17.13	Apprentice Programs Not Listed	148
	17.14	Training Programs	
Articl	le 18 - W	Vages	
	18.1	Rates of Pay	148
	18.2	Temporary Assignment	150
	18.3	Negotiation of Job Classifications	151
	18.4	Purpose of Job Descriptions	155
Articl	le 19 - D	Ouration	
	19.1	Duration	154
Lette	rs of Agi	reement	
		1 Arbitrator Selection	157
		Appendix to LOA 1	158
	LOA 2	2 Reslotting of Jobs	161
		3 Waiver of Time Limits	162
		4 Union Education Scholarships	163
		5 Employee Involvement	164

LOA 6 Equal Employment Opportunity	
Committee	165
LOA 7 Governmental Agency Rulings	170
LOA 8 Protective Clothing	170
LOA 9 Notice of Contracted Work	171
LOA 10 Reinstatement of Grievances	172
LOA 11 Returns to the Unit	174
LOA 12 Lockout, Hazardous Material	176
LOA 13 Training Stations	179
LOA 14 Arbitration Scheduling	180
LOA 15 Contract Work	182
LOA 16 Vacation Use	183
LOA 17 Training Stations - Students	
& Teachers	183
LOA 18 Grievance Prevention and	
Resolution	184
LOA 19 Prior Arbitration - Section 12.1	191
LOA 20 Retention Rate	191
LOA 21 Rewards and Recognition Programs	194
LOA 22 Medical Restrictions	
Laid-off Employees	195
LOA 23 Caterpillar Job Security Program	196
LOA 24 Plant Closing Restriction	209
LOA 25 Mediation	210
LOA 26 Lump Sum Payments	213
LOA 27 Supplemental Employees	213
LOA 28 Temporary Assignments	218
LOA 29 Employee Commitment to Business Success	219
LOA 30 Voluntary Separation Package	219
LOA 31 Hiring Bonus	220
LOA 32 Substance Abuse Program	221
LOA 33 Team Lead and Non-Traditional Roles	

# Appendix A Basic Hourly Rates of Pay 226 Supplementary Base Rate Schedule Local 974 230 Supplementary Base Rate Schedule Local 751 233 Supplementary Base Rate Schedule Local 2096 234

235

Memorandum of Agreement – Employees Hired Prior To January 10, 2005

#### AGREEMENT Between

Caterpillar Inc. and International Union, United Automobile, Aerospace and Agricultural Implement Workers of America and its affiliated Local Unions Nos. 751, 1872, 974, and 2096.

#### **PREAMBLE**

This Agreement entered into and concluded at Peoria, Illinois, on this 1<sup>st</sup> of March, 2023 by and between Caterpillar Inc., hereinafter designated and referred to as the "Company" but only with respect to the facilities owned or operated by the Company at locations covered by the bargaining units specified in Article 2 and the International Union, United Automobile, Aerospace and Agricultural Implement Workers of America and its affiliated Local Unions Nos. 751, 1872, 974, and 2096 hereinafter designated and referred to as the "Union," through their duly authorized representatives, do hereby agree as follows:

## ARTICLE 1 Purpose

- (1.1) The purpose of this Agreement is to establish harmonious relations between the parties and to facilitate orderly adjustment of grievances, complaints and disputes, which may arise from time to time between the Company and the Union. This Agreement is entered into in consideration of the mutual performance thereof in good faith by the parties.
- (1.2) The parties agree that wherever words such as "he," "him," "his," "Committeeman," or similar words appear in this Agreement, appended Letters of Agreement, Exhibits, Local Agreements, or Agreements and Plans relating to Pensions, Group Insurance, and Tax Deferred Retirement Plan, it is understood that such words are considered asexual and refer to females and males equally.

## ARTICLE 2 Recognition

(2.1) The Company recognizes the International Union, United Automobile, Aerospace and Agricultural Implement Workers of America (hereinafter sometimes called International Union) and its respective Local Unions 751, 1872, 974, and 2096 as the exclusive representatives severally for the respective bargaining units, hereinafter described, for the purpose of collective bargaining with respect to rates of pay, wages, hours of employment, and other conditions of employment.

The respective bargaining units to which the recognition of the International and the respective Local Unions extends are:

- (i) Local 974 and International Union the Peoria Unit as described in NLRB Certification dated June 10, 1948, Case No. 13-RC-175. The geographical limits of the above unit shall only include the Peoria Area.
- (ii) Local 751 and International Union the Decatur Unit as described in NLRB Certification dated September 26, 1955, Case No. 13-RC-4494 and Case No. 13-RC-4508.
- (iii) Local 1872 and International Union the York Unit as described in NLRB Certification

- dated April 5, 1954, Case No. 4-RC-2199. The geographical limits of the above unit shall only include the York Distribution Center.
- (iv) Local 2096 and International Union the Pontiac Unit as described in NLRB Certification dated February 16, 1979, Case No. 33-RC-2393.
- (v) Other Locals of the International that now or hereafter may represent employees at other facilities of the Company may with the consent of the Company and the International, become parties to this Agreement during its term; in such event the several Local Unions signatory hereto agree that the Preamble hereof and this Section 2.1 may be amended without further concurrence by them to reflect such joinder.
- (2.2) The provisions of this Central Agreement and appended Letters of Agreement (together with the appropriate Local Agreement when made and ratified) shall comprise a complete agreement for, and be individually applied to, each bargaining unit specified in Section 2.1 of this Agreement. This Central Agreement even though not a complete agreement as to all terms and conditions of employment at the several bargaining units, shall be effective and binding on the Union and the Company on the first Monday following the date on which the Company receives satisfactory notice from the International Union that it has been ratified by a majority of the employees represented by the Union voting on ratification. No Local Agreement shall change, alter or detract from this Agreement, but rather shall only supplement this Agreement to the extent not inconsistent with it. If there is a conflict or inconsistency between the provisions of this Agreement and those of any Local Agreement, the provisions of this Agreement shall be controlling.

## **ARTICLE 3** Union Security

- (3.1) Each employee shall, on and after the thirtieth day following his first day of employment in a bargaining unit, be required as a condition of employment in that bargaining unit to be and remain a member of the Union; provided, that this shall not apply (a) to any employee to whom such membership is not available on the same terms and conditions generally applicable to other members, nor (b) to any employee to whom membership in the Union is denied, or whose membership in the Union is terminated, for any reason other than his failure to tender the initiation fee and periodic dues uniformly required as a condition of acquiring or retaining such membership, nor (c) until the thirtieth day following the date on which this Agreement goes into effect, to any person who on the date this Agreement goes into effect, is an employee (whether or not actively working) but is not a member of the Union in good standing in accordance with its Constitution and Bylaws.
- (3.2) For the duration of this Agreement, any member of the Union desiring to have the Company check off the Union initiation fee, if any, and his regular monthly Union membership dues, may sign, secure any other signature necessary, and complete an "Authorization for Dues Check-Off" which will authorize the Company to deduct such dues from the employee's earnings. Such dues, regardless of source, shall be remitted via electronic funds transfer to the Financial Secretary of the respective Local Union. However, the Company shall not deduct initiation fees from the earnings of employees in the bargaining unit represented by Local 974.

The Union will provide the necessary "Authorization for Dues Check-Off" forms which shall authorize the Company to make such deductions and the Union will secure the necessary signature or signatures on the forms and deliver the properly completed and signed forms to the Company.

Such "Authorization for Dues Check-Off" is to be in the form agreed upon by the parties. Such "Authorization for Dues Check-Off" will be placed in effect by the Company during the next month following delivery to the Company, if such assignment is delivered not later than the day of the month specified in the respective Local Agreement. Otherwise, the assignment will be placed into effect the second month following.

Deductions to be made in accordance with this Section shall be from the employee's first pay distribution of each calendar month provided an employee has earnings in that pay period. If an employee has insufficient earnings in the first pay the deduction shall be made from a subsequent pay for weeks during that same month. If the employee has insufficient net earnings in any of these pays the amount of the dues shall not be carried into the succeeding month, but the Company shall report to the respective Local Union that the employee had no dues deducted for that month, having received insufficient earnings in any of these pays. Initiation fees shall be carried until collected.

A total of the deductions made in accordance with this Section 3.2 shall be paid to the Financial Secretary of the respective Local Union no later than five regularly scheduled workdays (excluding Saturdays, Sundays, holidays as listed in Section 7.8 and vacation periods as described in Section 9.2) after the end of the week in which pay is distributed which reflect the deductions and efforts will be made to make such payments by Tuesday of the week following the week in which pay is distributed which reflect the deductions.

The Union agrees that it will indemnify the Company and hold it blameless for any damages, which may be assessed as a result of the operation of this Section.

During any period when there is not a written agreement in effect between the Company and the Union which permits or provides for the deduction of union dues, the check-off of union dues will be suspended for such period only. At the end of any such period, the Company will continue to rely on the existing Authorization for Dues Check-Off forms on file to check-off dues.

(3.3) A "Check-Off Authorization" shall be irrevocable by an employee in the Union Locals 751, 1872, 974, and 2096 (1) for a period of one year from the date of delivery of such authorization to the Company or each successive period of one year or (2) until the termination of this Agreement, whichever occurs sooner, unless written notice is given by any such employee to the Company and the Union not more than twenty days and not less than ten days prior to the expiration of (a) any such one year period or (b) this Agreement, whichever occurs sooner.

A "Check-Off Authorization" of an employee who separates from a bargaining unit after the effective date of this Agreement, and who thereafter returns to that bargaining unit, shall be automatically reinstated upon such return, except when the employee's return is subsequent to the termination of this Agreement.

## **ARTICLE 4 Representation**

(4.1) Beginning with the commencement of the terms of office for Union Representatives who are elected subsequent to the effective date of the Agreement or June 1, 2023, whichever occurs earlier and in order to provide a system of Union Representation for the processing and settlement of grievances, the representation structure for each Local and the duties and privileges of such representatives shall be set forth in the appropriate Local Supplement. The number of Company-paid part-time (as necessary) committeemen for each Final Step Grievance Committee shall be as

Committee Commit	teemen
East Peoria Transmissions (Bldgs KK, HH (West) & Bldg H Electric Drive)	) 3
East Peoria Undercarriage (Bldgs BB, EE, GG & HH (East))	2
East Peoria Prime Product (Bldgs LL, NN & SS)	5
Decatur	5
Mapleton – CMO	3
Mapleton – Seals	2
Morton	3
Pontiac	3
Technical Center and Peoria Proving Ground	2
Maintenance Components	2
York	3

(4.2) Union Representatives will handle grievances with least possible interference with production and efficient operations. If in any instance the Company feels that such privileges are being abused, it shall so notify the Local Union, in writing, and in the event the abuse is not then corrected, or an understanding not reached, the Company may suspend the privileges of the Union Representative involved. Any dispute arising therefrom may then be taken up under the grievance procedure.

Number of

The Company agrees that the provisions of Sections 4.2 or 4.3 will not be administered in such a manner that Union Representatives will be denied the privileges granted Union Representatives in the performance of the functions permitted such representatives under this Agreement or the appropriate Local Agreement.

- (4.3) If, in the handling of a grievance in accordance with the provisions of this Agreement, or the appropriate Local Agreement, it is necessary for an employee or a Union Representative to leave his line, unit or immediate working area, the employee or Union Representative shall secure a pass from his immediate Supervisor. The Supervisor will issue the pass without undue delay. Such pass shall indicate the reason for the issuance of the pass and the area or areas, which the Union Representative is authorized to enter. The Union Representative will present the pass to the Supervisor of the area for which the pass was issued and indicate the employees he wishes to contact.
- (4.4) Each Local Union shall provide the Company with a list of all Union Representatives, members of Grievance Committees, members of the Bargaining Committee, and officers of the Local Union. Each Local Union shall notify the Company of any changes in this list as promptly as possible. The Company agrees to provide each Local Union with a list of supervisory employees, by departments and shifts, who are authorized and designated to handle grievances under the local grievance procedure. The Company shall notify the Local Union of any changes in this list as promptly as possible.
- (4.5) Upon request to a Business Unit Labor Relations Department, International Representatives of the Union (not to exceed two) shall be granted permission to visit a business unit during working hours for the purpose of investigating any specific grievance which is pending the Final Step of the grievance procedure, or which is scheduled to be heard at an arbitration hearing, and the proper investigation of which requires entry into that business unit. The Company will acknowledge the request and will set a time, which is mutually agreeable for such visit. A member

of the Final Step Grievance Committee and the Local Union President and Chairman of the Local Bargaining Committee may accompany the International Representative(s) during such visit.

Management representatives may accompany the Union Representative(s) during such visit. Such visits shall be of reasonable duration, and during these visits the International Representative(s) may interview the aggrieved employee or employees, provided the interviews do not materially interfere with production and efficient operation. The International Representative(s) shall be subject to all business unit rules and regulations while engaged in such visit.

- (4.6) The appropriate Committeeman will be permitted to discuss the final disposition of the grievance in the grievance procedure with the aggrieved employee.
- (4.7) Throughout both the Central Agreement and various Local Agreements, there are references to various facilities, locations, business units, divisions, and plants. Unless specifically provided otherwise, those terms shall refer to the following:

Decatur

York

East Peoria – Transmissions (Bldgs KK, HH (West) & Bldg H Electric Drive)

East Peoria – Undercarriage (Bldgs BB, EE, GG HH (East)) & Maintenance Components

East Peoria – Prime Product (Bldgs LL, NN & SS)

Mapleton - CMO

Mapleton – Seals

Technical Center and Peoria Proving Ground

Morton

Pontiac

For the purposes of Letter of Agreement 24, all of Mapleton (CMO/Seals) is one facility and all of East Peoria is one facility.

#### ARTICLE 5 Grievance Procedure

- (5.1) A grievance is a claim of an employee or the Union that his or their rights including interpretation, application or alleged violation of any provisions of this Agreement, working conditions not covered by this Agreement, appended Letters or Memorandums of Agreement or appropriate Local Agreement, have been violated.
- (5.2) It is mutually desired that grievances be satisfactorily settled as quickly as possible. The parties agree that frivolous and/or repetitive grievances shall not be filed. Information concerning repetitive grievances provided to supervisors will also be provided to the appropriate Grievance Committee Chairman.

If the Company refuses to accept a grievance, which it deems to be frivolous and/or repetitive, the appropriate Committeeman may request an explanation from the business unit Labor Relations Manager concerning the reason(s) for such determination. If, following such explanation, the Committeeman still believes that such grievance is not frivolous or repetitive; the grievance may be presented, in writing, directly to the Final Step of the local grievance procedure.

The filing and processing of multiple grievances on the same issue or incident by employees or

Union Representatives does not facilitate the resolution of grievances nor fulfill the intent of Section 4.2 or Section 16.5 of this Agreement. In order to minimize duplication of effort by Union Representatives, and to expedite the processing of grievances when more than one employee is aggrieved over the same issue or incident, a single group grievance on behalf of these employees will be presented to the appropriate step of the local grievance procedure.

When a grievance arises, the employee shall identify the issue by indicating the specific action or non-action on the part of the Company which prompts the grievance whereupon the Supervisor shall, without undue delay, send for the Union's First Step Grievance Representative and an earnest effort will be made to settle the grievance in accordance with procedures provided in the Local Agreements.

The parties jointly recognize that it is desirable to resolve, if possible, any question of fact relevant to any grievance at as early a Grievance Step as is reasonably possible and will cooperate with each other in an effort to do so; such cooperation shall extend, in appropriate cases, to joint investigation to establish the relevant facts.

- (5.3) Termination of any grievance in the First Step or, in locations having a Three Step grievance procedure, the Second Step of the local grievance procedure, either by the Company granting relief in whole or in part or by the Union withdrawing or dropping the grievance, shall not constitute a precedent for the settlement of any future grievance in any Step of the grievance procedure or in support of either party's position in arbitration; however, any such "no precedent" termination shall not constitute an eradication of the events which led to the Company action or non-action of which the grievance complained.
- (5.4) No claims against the Company, including claims for back pay, by an employee covered by this Agreement, or by the Union, shall be retroactive to any period more than 12 months prior to the date the grievance was first filed in writing. In the case of a discharged employee or other cases of accruing liability (as defined in Section 6.3 of this Agreement), all claims for back pay shall be limited to the amount of wages the employee would otherwise have earned from his employment with the Company, less the following:
  - 1. Any Unemployment Compensation, which the employee is, not obligated to repay or which he is obligated to repay but has not repaid nor authorized the Company to repay on his behalf.
  - 2. Compensation for personal services other than the amount of compensation he was receiving from any other employment, which he had at the time he last worked for the Company and which he would have continued to receive had he continued to work for the Company during the period covered by the claim.

Wages for total hours worked each week in other employment in excess of the total number of hours the employee would have worked for the Company during each corresponding week of the period covered by the claim shall not be deducted.

(5.5) Employees shall be given disciplinary layoff, suspended from employment, or discharged only for just cause. In imposing disciplinary layoff, suspension or discharge on a current charge, the Company will not take into account any prior infractions (including falsification of employment application) which occurred more than three years previously nor, in the case of attendance infractions, any absence paid under Section 15.1 for which the Company was notified 30 minutes prior to the start of the employee's shift. In the event an employee has been given disciplinary layoff,

suspended from employment, or discharged, his Supervisor will send for the employee's Committeeman to discuss, for such time as may reasonably be necessary, the case with the employee. In the event the employee's Committeeman is not available, the appropriate Union representative will be called.

Within a reasonable time (not to exceed 5 working days) after disciplinary layoff, suspension from employment or discharge has been imposed, either party may request that a disciplinary hearing be conducted by the appropriate Final Step Company representative. At the disciplinary hearing, the employee shall have the right to be present and the appropriate Committeeman shall have the right to represent him. If the parties mutually agree, witnesses may be jointly interviewed so the relevant facts can be ascertained. Employees at work who participate in a disciplinary hearing will not lose pay for regularly scheduled hours spent in such hearing.

Grievances involving disciplinary layoff, suspension or discharge may be presented, in writing, directly to the Final Step of the local grievance procedure. The Company must be notified of a claim of wrongful disciplinary layoff, suspension or discharge within fifteen calendar days after same occurs, and the case shall be taken up promptly and diligent efforts made to dispose of it.

- (5.6) An employee who receives a warning will, where the act resulting in the warning would constitute cause for disciplinary action if continued, be given a written notice setting forth his name, identification number, date of the warning, and the reason for the warning.
- (5.7) If, in the Final Step of the local grievance procedure, the Union considers as unsatisfactory the Company's decision on a grievance which has as its basic issue a medical disagreement wherein the findings of the Company's physician or physicians are in conflict with the findings of the employee's personal physician, the medical question shall be submitted to a third physician mutually agreed to by the parties within five working days following the Company's decision on such grievance. The medical opinion of the third physician after examination of the employee and consultation with the other two physicians shall resolve such conflict. A copy of the third physician's findings shall be supplied to both the Company and the Union. The expense of the third physician shall be paid one-half by the Company and one-half by the Union, each of whom will be billed separately by the third physician for their share of such expense.

Each Local Union may, at its option, develop with the Company a list of mutually agreeable physicians including specialists, and/or medical facilities to which such medical disputes may be referred.

(5.8) The Company and the Union agree that, no later than the 60<sup>th</sup> day after a grievance is date stamped into Final Step of the grievance procedure, at each business unit of the Central Agreement, the Company will answer the grievance. The 60 day period for grievances to be answered may be extended for a specific grievance by mutual agreement of the parties.

#### ARTICLE 6 Arbitration

(6.1) Except for those grievances referred to in Section 6.9 below, if the Union is of the opinion that the Company's Final Step decision on a grievance involving the interpretation, application or violation of any provision of this Agreement, appended Letters of Agreement or appropriate Local Agreement is unsatisfactory and that the grievance merits consideration for possible arbitration, the appropriate Local of the Union shall prepare a "Union Statement of Unresolved Grievance" and shall

refer the grievance to the appropriate Regional Director of the International Union, or his designated representative, who shall review such grievance and all relevant facts to determine whether it warrants further consideration. If the Regional Director or his designated representative determines that the grievance warrants further consideration, he shall deliver the "Union Statement of Unresolved Grievance" to the UAW-Agricultural Implement Department. It shall be the final responsibility of the UAW-Agricultural Implement Department to determine whether such grievance shall be referred to arbitration as hereafter provided.

If the UAW-Agricultural Implement Department is of the opinion that such a grievance merits consideration for possible arbitration, the UAW-Agricultural Implement Department shall within, but not beyond, 60 calendar days of the date of the Company's answer in the Final Step of the grievance procedure, prepare and deliver a written "Notice of Appeal to Arbitration" to the Company's Corporate Labor Relations Office. If such notice is not given within such time, the grievance shall be deemed closed.

- (6.2) The Company and the International Union (acting for itself and the appropriate Local Unions) shall, pursuant to the procedures heretofore separately agreed upon, select a Permanent Arbitrator, and if necessary any successor Permanent Arbitrators, subject, however, to his or their removal by death, incapacity, resignation, or request of the Company or the International Union. Should the Company or the International Union desire to terminate the appointment of a Permanent Arbitrator, the one so desiring shall give written notice to the other and to the Permanent Arbitrator terminating the latter's authority as of the date specified in such notice after which he shall have no authority to receive additional cases or to hold hearings (other than to complete any hearings in which partial evidence has been heard but not concluded); the Arbitrator shall, however, complete any hearings in which partial evidence has been heard and render decision thereon and on any pending cases in which hearings have been completed. Any case in which the Arbitrator is unable or unwilling to render a decision shall go over to the successor Arbitrator for hearing or rehearing and decision. If a party gives notice to terminate the Arbitrator's authority it shall bear the Arbitrator's termination fee and any cancellation fees for future scheduled hearings.
- (6.3) Grievances referred to arbitration shall be heard in the chronological order of their referral, provided, however, that grievances involving discharges and other cases of similar accruing liabilities shall be heard in the chronological order of their referral prior to the hearing of any other grievance. The phrase "grievances involving discharge and other cases of similar accruing liabilities" shall be interpreted to mean discharges, improper layoff, failure to recall an employee, or some other circumstance wherein an employee is not at work and disputes involving the suspension of the privileges of a Union Representative under the provisions of Section 4.2 of this Agreement.

By mutual agreement of the Company and the International Union (acting for itself and the appropriate Local Union), specific grievances may be heard out of chronological order.

(6.4) The dates for hearings of grievances referred to arbitration shall be established by mutual agreement of the Permanent Arbitrator, the Agricultural Implement Department of the International Union and Corporate Labor Relations Office. A transcript will be prepared for each hearing by a court reporter. The parties may file post-hearing briefs with the Arbitrator within such period as he may specify. However, either party may, at its option and without establishing precedent for future action, waive the filing of a post-hearing brief, in which event, notice of intent to waive such filing shall be given to the Arbitrator and the other party within five calendar days after the hearing or receipt of the transcript, whichever is appropriate.

(6.5)

a. A tentative docket of those cases to be considered at a given hearing shall be furnished to the Corporate Labor Relations Office and the Permanent Arbitrator in the numerical order for scheduling at least eighty calendar days prior to the date of the hearing. This tentative docket may list more cases than the number to be scheduled for the hearing.

At least sixty-five calendar days prior to a particular hearing, representatives from the UAW-Agricultural Implement Department and the Corporate Labor Relations Office will meet to attempt to resolve any grievance referred to it by determining whether or not the facts of the case have resulted in a violation of the provisions cited. The Regional Director of the UAW (or his designated representatives), the Bargaining Chairman and/or the Chairman of the Final Step Grievance Committee, and the Labor Relations Manager (or his designated representatives) may attend such meetings.

If unresolved grievances remain following this meeting and the UAW-Agricultural Implement Department wishes to process such grievances to arbitration, the UAW-Agricultural Implement Department shall forward to the appropriate Chairman of the Final Step Grievance Committee and Labor Relations Manager a "Notice of Statement of Unresolved Grievances Due". This written notice shall be forwarded within ten days after such meeting, and shall specify those grievances, which remain unresolved from the tentative docket. If no such Notice is delivered within such time, all grievances on the tentative docket shall be deemed closed. If any grievance is bypassed in the chronological order, as listed in such Notice, such bypassed grievance(s) shall be deemed closed.

At least thirty calendar days prior to the hearing, the appropriate Local Bargaining Chairman and Corporate Labor Relations shall meet to simultaneously exchange "Statements of Unresolved Grievances" for not more than the following maximum number of grievances in chronological order (as defined in Section 6.3).

Length of Hearing	Number of Grievances
1 day	2
2 days	4
3 days	6

If no Statements of Unresolved Grievance are provided for simultaneous exchange by the Union then all grievances listed on the "Notice of Statement of Unresolved Grievances Due" shall be deemed to be closed. If less than the above maximum number of statements are provided for simultaneous exchange by the Union, then all other grievances listed in such Notice shall be deemed closed. If in the simultaneous exchange of statements, any grievances listed in the Notice are bypassed by the Union, then all such bypassed grievances shall be deemed closed.

The "Union Statement of Unresolved Grievance" shall contain (i) the specific action or non-action claimed to constitute a violation of this Agreement, appended Letters of Agreement or appropriate Local Agreement and all of the facts relied upon in support of such claim, (ii) the specific section(s) of such Agreement or Agreements claimed to be violated, together with a statement of the reason(s) why such action or non-action is claimed to constitute a violation of such section(s), and (iii) the specific relief requested. The "Company Statement of Unresolved Grievance" shall contain (i) the Company's understanding of the issue involved, (ii) all of the facts relied upon by the Company in defense against the grievance,

and (iii) the Company's position on the issue.

Within ten (10) days of such exchange, either party may elect to file with the other a "Notice of Additional Facts". This "Notice of Additional Facts" is intended solely to allow the filing party to counter facts documented on the other party's "Statement of Unresolved Grievance" which were not addressed on the filing party's "Statement of Unresolved Grievance." Such "Notice of Additional Facts" must identify the specific facts being countered.

It is the purpose and intent of the "Statements of Unresolved Grievance" and "Notices of Additional Facts" to assure that there will be full discussion and consideration of the grievance on the basis of a full disclosure of the relevant facts and positions. If such grievance is pursued to arbitration and if, in arbitration, either party shall attempt to deviate from or supplement the contents of such statements, and/or notices, the Arbitrator shall either (1) obtain the concurrence of the other party to admit the deviation or supplement, or (2) restrict the evidence and testimony to the issues, alleged violations, facts, and positions contained in the original statements and/or notices.

- b. Additions to the docket of those cases referred to in Section 6.9 below which are to be considered at a given hearing shall be furnished to the Corporate Labor Relations Office in the chronological order of their referral at least thirty calendar days prior to the date of the hearing.
- c. Thirty calendar days prior to a particular hearing, the docket for that hearing shall be frozen by taking the following number of unresolved cases from the top of the tentative docket described in (a) and (b) above:

Number of Cases
2
4
6

Within thirty calendar days prior to a hearing, substitutions in the docket may only be made by mutual agreement of the parties.

(6.6) Not later than the day of the arbitration hearing, the parties will meet for the purpose of preparing and signing a submission to arbitration, which will include a statement of the issue or issues and the Arbitrator shall decide only the issue or issues set forth in such submission. However, if the parties fail to agree upon and sign such submission by the hearing date, and if the Union Statement of Unresolved Grievance sets forth (i) the specific action or non-action claimed to constitute a violation of this Agreement, appended Letters of Agreement or appropriate Local Agreement, (ii) the specific Section or Sections of such Agreement or Agreements claimed to be violated together with a statement of the reason or reasons why the action or non-action specified in (i) is claimed to constitute a violation of such Section or Sections, and (iii) the relief requested, the grievance shall be presented to arbitration notwithstanding the failure to agree upon a submission. In such latter event, the Arbitrator shall determine whether the Union Statement of Unresolved Grievance reasonably complies with the requirements of the foregoing sentence. If it does not comply, he shall refer the matter back to the appropriate Local Union. In such event, the Local Union will prepare and deliver a new Statement of Unresolved Grievance. The grievance shall then be processed in the same manner as though such were the first Union Statement written and delivered. If the Union Statement of Unresolved Grievance does comply, he shall proceed to a decision on the

merits, but shall determine only whether or not the action or non-action specified in the Union Statement of Unresolved Grievance violates the Section or Sections of the Agreement or Agreements so specified and, if such violation has occurred, the relief to be granted, which shall not be broader than the relief so requested.

(6.7) The decision of the Arbitrator shall be final and binding upon all parties to the dispute. However, the Arbitrator shall have no power to change, alter, detract from, or add to the provisions of this Agreement, the appended Letters of Agreement, and/or the appropriate Local Agreement but shall have the power only to apply and interpret the provisions of such Agreements or Letters in reaching his decision; the Arbitrator shall have no power to rule on any dispute arising under Section 16.2 of the Decatur, Section 6.7 of the Peoria, Section 6.6 of the Pontiac Local Agreements, or Section 18.3 hereof.

It shall be incumbent upon the Arbitrator to recognize that he has been engaged by the parties for the sole purpose of adjudicating disputes that are properly before him and that his conduct of the hearing should be governed accordingly. The parties have agreed that the Arbitrator should conduct an orderly hearing and that the Arbitrator should not intrude into either party's presentation so as to prevent that party from putting forward its case in the manner and sequence determined by that party, subject, of course, to reasonable rules of arbitration procedure and evidence.

The parties assume the sole responsibility for developing the theory, arguments and proof of their respective positions in the arbitration procedure. The Arbitrator shall respect their judgment as to the content of their presentation and shall not engage in developing additional or alternative theories, arguments or proof supportive of either party's position. The Arbitrator shall neither accept as fact, nor require the other party to refute, either party's charges, allegations or statements that are not supported by proper evidence.

The provisions of this Article 6 do not authorize the Arbitrator to serve as a mediator, conciliator or conduct an independent investigation of facts unless the parties so mutually agree. It is the desire of both parties that the Arbitrator serve the parties in the adjudication role described above and for the purpose for which the arbitration process was originally developed - a means by which a truly impartial third party could adjudicate disputes between two parties to a labor agreement over the interpretation, application or violation of that agreement.

The parties wish to make clear that the arbitrator has the authority to decide alleged violations of Sections 8(a)(1), and 8(a)(3) of the National Labor Relations Act. When issuing a decision on the alleged contract and unfair labor practice violation(s), the arbitrator shall identify the issue(s), and explain why the facts presented at the hearing either do or do not support the alleged violation(s).

- (6.8) Expense of the arbitration, including the Arbitrator's fees and expenses and stenographic expenses, shall be borne equally by both parties subject to the special provisions applicable in the event one of the parties terminates the Arbitrator's authority.
- (6.9) The foregoing provisions of this Central Agreement are amended only to the extent necessary to provide an expedited arbitration procedure for grievances wherein the parties mutually agree in the Final Step of the grievance procedure that the sole issue is whether there was "just cause" for either discharging an employee, or giving an employee a disciplinary layoff.

If the appropriate Local of the Union is of the opinion that such a "just cause" grievance merits consideration for possible expedited arbitration, the appropriate Local of the Union shall within, but

not beyond, 60 calendar days of the Company's answer in the Final Step of the grievance procedure, prepare and deliver a written "Notice of Appeal to Expedited Arbitration" to the appropriate Labor Relations Manager with copies to the UAW-Agricultural Implement Department and the Company's Corporate Labor Relations Office. If such notice is not given within such time, the grievance shall be deemed closed.

Such grievances shall be heard in the chronological order of their referral, provided, however, that grievances involving discharges shall be heard in the chronological order of their referral prior to the hearing of any other grievance in expedited arbitration. Grievances referred under this Section 6.9 will be heard by the panel arbitrator.

The parties agree to waive the right to file a post-hearing brief in such expedited arbitration. The Arbitrator shall render an award within 60 calendar days of receipt of the transcript. The Arbitrator will limit his opinion to that which is necessary to briefly explain his award.

In the event a grievance scheduled for expedited arbitration is resolved at least thirty days prior to the date of the hearing, another such just cause grievance may be substituted by mutual agreement of the parties.

#### ARTICLE 7 Hours of Work and Overtime

- (7.1) The regular weekly schedule of work of employees on the regular first and second shifts shall consist of five days Monday through Friday, inclusive, and of employees on the regular third shift shall consist of five days Sunday night through Thursday night, inclusive. For all purposes of this Agreement and each Local Agreement, except the preceding sentence, and notwithstanding any inconsistent provision elsewhere, in the case of employees scheduled to work on the regular third shift, Sunday night through Thursday night, the day of work shall be considered to be the calendar day upon which the shift is scheduled to end, and in the case of employees scheduled to work on irregular schedules who are scheduled to work on Sundays rather than Mondays, because of the fact that the regular weekly schedule of the regular third shift begins on Sunday instead of Monday, the day of work shall be considered to be the calendar day on which their shift is scheduled to end. Due to the diversity of alternate work schedules, alternate work schedule shift designations will be determined by the Company.
  - (7.2) The regular shifts will be as follows:
  - a. The regular first shift will start between 6:00 a.m. and 8:00 a.m., the exact starting times to be determined by the Company.
  - b. The regular second shift will start between 2:00 p.m. and 4:00 p.m.
  - c. The regular third shift will start between 10:00 p.m. and 12:00 midnight.

Employees on these shifts will be scheduled to work eight hours, including a 20-minute lunch period to be considered and paid for as time worked.

An employee who is scheduled to work four or more hours overtime before or beyond his regular shift will be allowed time for a brief snack, not to exceed twenty minutes, before the start or end of his regular shift, whichever is applicable. The twenty-minute brief snack period will also apply to

any employee whose regularly scheduled shift is twelve hours or more.

- (7.3) It is recognized that because of operation requirements, employees may be assigned to a schedule other than a regular shift as above defined. Exceptions to the regular shift are:
  - 1. Irregular schedules consisting of five eight-hour shifts, not necessarily coinciding with the regular shift, during the seven calendar days Monday through Sunday.
  - 2. Alternate work schedules consisting of all schedules other than those defined as regular or irregular.

When a new alternate work schedule is initiated within a work group, all of the initial assignments to such schedule will be made on a voluntary basis. However, in the event that insufficient qualified volunteers are available to meet initial staffing requirements, the Company shall have the right to require incumbents to work the alternate work schedule and to assign employees to any unfilled positions as otherwise permitted under this Agreement provided, however, the foregoing provisions of this sentence shall not be applicable to alternate work schedules which involve shifts of more than twelve hours.

Employees assigned to such irregular schedules or alternate work schedules, and who feel that such assignments are not based upon real and practical considerations, may seek recourse under the grievance procedure.

(7.4) Beginning on the first Monday following ratification of the 2023 Central Agreement, the Company will pay a nightshift premium of 75 cents per hour for all hours worked by an employee whose regular schedule begins between the hours of 12:00 noon and 2:00 a.m., inclusive. Beginning with the first full pay period in March 2025, the Company will pay a nightshift premium of \$1.00 per hour for all hours worked by an employee whose regular schedule begins between the hours of 12:00 noon and 2:00 a.m., inclusive.

An employee who is regularly assigned to and works a shift whose starting time entitles the employee to receive a nightshift premium for all hours worked on such shift and who is temporarily assigned to the regular first shift for a period of time (1) to receive training which the Company determines is necessary, (2) to work during each of the "vacation periods" set forth in Section 9.2 of this Agreement, or (3) to work less than a week on the regular first shift shall continue to receive a night shift premium for all hours worked during such periods.

An employee who has a regular weekly schedule which consists exclusively of 3 twelve hour days will receive an additional 4 hours pay per week at straight time.

- (7.5) When there is no work, or not enough work, and an employee is permitted to report for work on his regular shift without being notified not to report, then the employee shall receive a minimum of four hours' pay at whatever rate is applicable, unless the lack of work is caused by a labor dispute within the plant or is brought about by a condition beyond the control of the Company.
- (7.6) An employee called to work outside of his regular schedule shall receive a minimum of four hours' pay at whatever rate is applicable. The charge an employee received for overtime allocation purposes will be compatible with the rate the employee is paid.
  - (7.7) An overtime premium shall be paid as follows:

- A. Work in excess of the employee's regularly scheduled hours on any regularly scheduled workday shall be paid at the rate of time and one-half.
- B. Work performed on days outside of an employee's regular workweek schedule:
  - 1. For the first such day shall be paid at the rate of time and one-half.
  - 2. For the second such day, regardless of whether the employee worked such first available day, shall be paid at the rate of double time.
  - 3. For the third or fourth such day shall be paid at the rate of time and one-half.
- C. Work performed on recognized holidays shall be paid at the rate of double time.
- (7.8) When the following holidays (with the exception of December 24 and December 31 of any year) fall on a Sunday, Monday will be observed as the holiday.

The following are recognized holidays:

A. At all Business Units except as noted in B of this Section:

April 7, 2023 (Fri.)

May 29, 2023 (Mon.)

July 4, 2023 (Tues.)

September 4, 2023 (Mon.)

November 23, 2023 (Thurs.)

November 24, 2023 (Fri.)

December 24, 2023 (Sun.)

December 25, 2023 (Mon.)

December 31, 2023 (Sun.)

January 1, 2024 (Mon.)

January 15, 2024 (Mon.)

March 29, 2024 (Fri.)

May 27, 2024 (Mon.)

July 4, 2024 (Thurs.)

September 2, 2024 (Mon.)

November 28, 2024 (Thurs.)

November 29, 2024 (Fri.)

December 24, 2024 (Tues.)

December 25, 2024 (Wed.)

December 31, 2024 (Tues.)

January 1, 2025 (Wed.)

January 20, 2025 (Mon.)

April 18, 2025 (Fri.)

May 26, 2025 (Mon.)

July 4, 2025 (Fri.)

September 1, 2025 (Mon.)

November 27, 2025 (Thurs.)

November 28, 2025 (Fri.)

December 24, 2025 (Wed.)

December 25, 2025 (Thurs.)

December 31, 2025 (Wed.)

January 1, 2026 (Thurs.)

January 19, 2026 (Mon.)

April 3, 2026 (Fri.)

May 25, 2026 (Mon.)

July 4, 2026 (Sat.) Observed on July 3, 2026 (Fri.)

September 7, 2026 (Mon.)

November 26, 2026 (Thurs.)

November 27, 2026 (Fri.)

December 24, 2026 (Thurs.)

December 25, 2026 (Fri.)

December 31, 2026 (Thurs.)

January 1, 2027 (Fri.)

January 18, 2027 (Mon.)

March 26, 2027 (Fri.)

May 31, 2027 (Mon.)

July 4, 2027 (Sun.) Observed on July 5, 2027 (Mon.)

September 6, 2027 (Mon.)

November 25, 2027 (Thurs.)

November 26, 2027 (Fri.)

December 24, 2027 (Fri.)

December 25, 2027 (Sat.)

December 31, 2027 (Fri.)

January 1, 2028 (Sat.)

January 17, 2028 (Mon.)

April 14, 2028 (Fri.)

May 29, 2028 (Mon.)

July 4, 2028 (Tues.)

September 4, 2028 (Mon.)

November 23, 2028 (Thurs.)

November 24, 2028 (Fri.)

December 24, 2028 (Sun.)

December 25, 2028 (Mon.)

December 31, 2028 (Sun.)

January 1, 2029 (Mon.)

January 15, 2029 (Mon.)

#### B. At Caterpillar Logistics Services facilities:

May 29, 2023 (Mon.)

July 4, 2023 (Tues.)

September 4, 2023 (Mon.)

November 23, 2023 (Thurs.)

November 24, 2023 (Fri.)

December 24, 2023 (Sun.)

December 25, 2023 (Mon.)

December 31, 2023 (Sun.)

```
January 1, 2024 (Mon.)
May 27, 2024 (Mon.)
July 4, 2024 (Thurs.)
September 2, 2024 (Mon.)
November 28, 2024 (Thurs.)
November 29, 2024 (Fri.)
December 24, 2024 (Tues.)
December 25, 2024 (Wed.)
December 31, 2024 (Tues.)
January 1, 2025 (Wed.)
May 26, 2025 (Mon.)
July 4, 2025 (Fri.)
September 1, 2025 (Mon.)
November 27, 2025 (Thurs.)
November 28, 2025 (Fri.)
December 24, 2025 (Wed.)
December 25, 2025 (Thurs.)
December 31, 2025 (Wed.)
January 1, 2026 (Thurs.)
May 25, 2026 (Mon.)
July 4, 2026 (Sat.) Observed on July 3, 2026 (Fri.)
September 7, 2026 (Mon.)
November 26, 2026 (Thurs.)
November 27, 2026 (Fri.)
December 24, 2026 (Thurs.)
December 25, 2026 (Fri.)
December 31, 2026 (Thurs.)
January 1, 2027 (Fri.)
May 31, 2027 (Mon.)
July 4, 2027 (Sun.) Observed on July 5, 2027 (Mon.)
September 6, 2027 (Mon.)
November 25, 2027 (Thurs.)
November 26, 2027 (Fri.)
December 24, 2027 (Fri.)
December 25, 2027 (Sat.)
December 31, 2027 (Fri.)
January 1, 2028 (Sat.)
May 29, 2028 (Mon.)
July 4, 2028 (Tues.)
September 4, 2028 (Mon.)
November 23, 2028 (Thurs.)
November 24, 2028 (Fri.)
December 24, 2028 (Sun.)
December 25, 2028 (Mon.)
December 31, 2028 (Sun.)
```

January 1, 2029 (Mon.)

At the Caterpillar Logistics Services facilities in addition to the foregoing holidays, each employee shall be entitled to two (2) additional holidays during each twelve-month period starting

on the first Monday in the first full pay period in April and ending on the last day of the last pay period prior to the first Monday in the first full pay period in April the year following. Such holidays shall be scheduled to be taken at such times as will cause the least inconvenience in the area of operations in which these employees work and at times to be approved by their Supervisor.

- (7.9) Except as provided in Section 7.1, for purposes of computing overtime under Sections 7.7 and 7.8 above, the day of work shall be considered the calendar day upon which the employee is authorized to start work. No employee on a regular schedule as defined in Section 7.1 shall be required to take time off from his regular workweek in order to work a Saturday or Sunday.
- (7.10) Overtime shall not be pyramided or paid twice for the same hours worked nor shall an employee receive payment for the same day under more than one of the provisions of Sections 15.1, 15.2, 15.3, 15.4, 15.5, or 15.6.
- (7.11) All overtime will be allocated in accordance with the provisions of the applicable Local Agreements.
- (7.12) An employee required to work overtime on a regularly scheduled workday shall be given notice at least the workday prior to the day upon which the overtime is to be worked. An employee required to work overtime on a day not regularly scheduled shall be given notice at least two workdays prior to the day upon which the overtime is to be worked. An employee may be required to work overtime only in the event he is properly notified as herein provided.
- (7.13) An employee not desiring to work overtime, however, shall not be required to do so, if he promptly notifies his Supervisor, and if within the group normally sharing the overtime, there can be found another employee available who would be willing to work such overtime. An employee so notifying his Supervisor shall, in any event, be excused from such overtime assignment, if his request to be so excused is based on compelling personal reasons (such as personal illness or injury in his immediate family).

An employee shall not be required to work on any day other than the employee's regularly scheduled workdays during any week if the employee has worked all of the employee's regularly scheduled workdays plus at least one day outside of the employee's regular schedule in each of the previous two consecutive weeks. For purposes of the preceding sentence, holidays, other than those observed on Saturday, shall be deemed to be days worked.

An employee assigned to a regular weekly schedule of work as defined in Section 7.1 will not be required to work a Saturday, or an employee assigned to an alternate schedule of work as defined in Section 7.3 will not be required to work a Saturday, which falls outside their normal work schedule that immediately follows, or precedes, any of the holidays specified in Section 7.8 falling on a Friday or Monday, except when necessary to protect the physical plant and/or equipment or unless the employee has been notified at least 14 calendar days earlier that his services are needed for repairs that the Company has scheduled during such holiday weekend. This paragraph is not applicable to employees scheduled for emergency parts shipments.

An employee assigned to a regular weekly schedule of work as defined in Section 7.1 will not be required to work over ten (10) hours in any day during such week. An employee assigned to an alternate schedule of work as defined in Section 7.3 will not be required to work over two (2) hours in addition to the employee's regularly scheduled hours on any day during such week provided, however, such employee will not be required to work more than twelve (12) hours on any day if

such employee's regularly scheduled hours are less than twelve (12) hours on such day. An employee assigned to an alternate schedule of work as defined in Section 7.3 will not be required to work any additional hours on any day during such week if such employee's regularly scheduled hours are twelve (12) or more hours on such day. No employee, except those assigned to an irregular work schedule or an alternate work schedule, will be required to work on a Sunday, and no employee, except those assigned to an irregular work schedule, will be required to work on a holiday, except when necessary work is required to protect the physical plant and/or equipment.

- (7.14) The Company will advise the appropriate Union representative of overtime being worked by each employee in accordance with the respective Local Agreements.
- (7.15) All employees shall have their pay distributed by electronic funds transfer to a financial institution of their choosing which accepts such funds transfer. If the use of electronic funds transfer creates a demonstrated hardship on any employee, the appropriate Committeeman may discuss the issue with the business unit Labor Relations Manager. The parties will make a good-faith effort to resolve the issue.

Employees will be paid on a biweekly basis. Payday will be on Friday or on the next banking day preceding Friday if Friday is a banking holiday. The Company will distribute nonnegotiable pay stubs to all employees.

(7.16) Wherever in this Agreement, or the Local Agreements supplemental to this Agreement, the terms "regular scheduled workday," "normal scheduled days of work," "regularly scheduled hours," and all such similar terms are used, reference is intended to be the employee's usual and customary workweek and daily work hours.

## ARTICLE 8 Health and Safety

(8.1) The Company, the Union and employees will cooperate toward the prevention of accidents and furtherance of a safety program. The Company will continue in its efforts to protect and promote the health of all employees. The Company will endeavor to maintain a clean, properly lighted, heated, and ventilated factory with approved safety devices and will provide medical services at those plants where such is practical, or, alternatively will make arrangements for medical services to be provided by a licensed physician or other licensed health care provider at a reasonably convenient location.

During 2023 negotiations, the Company and the Union reaffirmed their common goals for a safer, more productive workplace for our employees and its members. The parties are mutually aware of the need for continued interest in the health and safety of employees covered by this Agreement. The parties jointly recognize that the elimination or minimization of unsafe or unhealthy acts and conditions in the workplace is of mutual benefit and, as such, should not be pursued in an adversarial environment. The parties further recognize the mutual advantage of correcting any deficiencies and resolving any disputes at the earliest opportunity. Therefore, the parties agree to place renewed attention, emphasis, and effort into the use of the local safety complaint procedure. To this end, the Union, is committed to work diligently towards enhancing the health and safety of its members and is equally committed to working with the Company to achieve this goal. In order that it may assist its members, the Union will encourage them to use these procedures before considering referral of the matter to any governmental agency. The Company agrees to continue to ensure that Supervisors at all levels give proper priority to health and safety issues and that such Supervisors are informed

of the facilities and techniques available to respond to health and safety issues. The parties are committed to the goal of reducing health and safety hazards and creating a safe environment for the benefit of all.

(8.2) Each plant shall have a Safety Committee, composed of management and Union representatives, in accordance with the provisions of applicable Local Agreements. The appropriate Safety Representative will provide the business unit Labor Relations Manager written notice of all Alternate Safety Committee members and shall notify the business unit Labor Relations Manager of any changes to said notice. Changes will not be made more than four (4) times per calendar year for each safety committee, and such changes will not be effective until five (5) business days from the receipt of said written notice.

When it is necessary for a safety committee member to be absent, such safety committee member will be replaced by the appropriate alternate safety committee member. When both the Safety Committee member and their designated alternate are absent, representation, if required, will be provided by an adjacent safety committee member.

It shall be the duty of the Safety Committee to meet on a monthly basis and at such other times as mutually agreed without loss of pay for regularly scheduled hours. A copy of the minutes of such meetings will be provided to members of the Safety Committee. Such minutes shall include: the date of the meeting, names of the individuals present, a brief statement of items discussed and the consensus or disposition, if any, reached on those items.

The Company will provide monthly, to the plant Safety Committee, one current copy of that Business Unit's OSHA Form #300, #301 and man-hours worked. The Company will provide the Director of the Plant Safety Committee, or the alternate in the Director's absence, with a copy of the OSHA Form #301 for any recordable incident included in the Business Unit's OSHA Form #300.

The functions and objectives of the Safety Committee will be:

- a. To encourage the observation of safety rules and the furtherance of the safety program.
- b. To review recordable, serious or unusual injuries and illnesses experienced within the plant and recommend possible corrective measures where appropriate.
- c. To review significant developments of mutual interest in the industrial health and safety fields and consider the applicability of such developments to the plant.
- d. To review new manufacturing equipment and major process changes where employee health or safety may be affected and make appropriate recommendation.
- e. To review established and proposed safety procedures for recognized hazardous materials (metalworking fluid programs; mist, fluid, and particulate ventilation; and foundry environmental air programs) and physical hazards (noise, heat and radiation) to which employees are exposed and make appropriate recommendations.
- f. To review significant changes in the Company's health and safety programs due to legal requirements or Company-initiated revisions. The Company will supply this information to the Committee in advance of implementation to allow sufficient opportunity to discuss these programs and make appropriate recommendations for improvement.

- g. To review and recommend safety-related improvements in current safety training programs, such as vehicle operator safety training, machine operator safety training, lockout, and hazardous material training.
- h. To periodically review the plant hearing conservation program and discuss issues including sound surveys, design specifications, and make appropriate recommendations for improvement.
- i. To periodically review the facility ergonomics program, progress in reducing musculoskeletal injuries, procedures for reporting ergonomic concerns, corrective actions being taken on high risk jobs, and to make appropriate recommendations for improvement.
- j. To periodically review the plant Safety and Ergonomics Concern Log.
- k. To periodically review the risks at the plant associated with working at elevated heights, review existing fall protection systems, and make appropriate recommendations for improvement.

#### (8.3) Safety Procedure

Stage 1: An employee who believes that a condition has developed which presents a significant threat to his health or safety shall promptly notify his Supervisor of such condition. The Supervisor shall determine, as promptly as possible, whether such condition represents a significant threat to the health or safety of the employee or employees involved and, if indicated, initiate appropriate corrective measures.

Stage 2: If a satisfactory solution to the problem cannot be agreed upon in Stage 1, the employee may request and the Supervisor shall, as soon as possible, but no later than the end of the employee's next regularly scheduled shift, send for the Union Safety Sub-Representative (if such there be) in whose jurisdiction the condition exists for the purpose of conducting a Stage 2 joint investigation of the problem with the Supervisor. Such investigation may also receive the attention of the second-level Supervisor. At plants where Safety Subcommittees have not been established, the Union Safety Representative in whose jurisdiction the condition exists will be called initially as provided herein.

Stage 3: If a satisfactory solution to the problem cannot be agreed upon in the foregoing Stage 2 joint investigation, the investigation may be broadened at that time, if the Union Safety Sub-Representative or Union Safety Representative, as the case may be, so requests of the Supervisor, by the addition of (1) the Director of the Union Safety Committee (in the Peoria area, the Divisional or Plant Union Safety Representative, and in the Decatur Plant the Union Safety Representative, in whose jurisdiction the condition exists), (2) the Superintendent of the area, or his designated representative, and (3) the Company Safety Supervisor or his designated representative. It is understood that the Union representatives specified in (1) above will have designated alternates on the other two shifts who will function in their stead on shifts other than the shifts on which the representatives specified in (1) above work and that the Company will have been notified of these designated alternates. If a satisfactory solution to the problem is not arrived at within 3 working days the employee will be notified of the status of his complaint.

Stage 4: If a satisfactory solution to the problem has not been agreed upon at the conclusion of the Stage 3 joint investigation, a grievance may be filed directly, within five working days, to the

Final Step of the grievance procedure by the member of the Final Step Grievance Committee within whose jurisdiction the alleged unsafe condition exists, and such grievance thereafter may be processed through the grievance procedure and, if not resolved, to arbitration. The Director of the Union Safety Committee will be allowed to attend the Final Step meeting solely for the purpose of discussing a safety grievance that has been processed through the safety complaint procedure and submitted to the Final Step agenda.

Prior to the submission of the written grievance, the above-designated member of the Final Step Grievance Committee may discuss the safety problem with the appropriate Union Safety Representative provided that the Grievance Committeeman notifies the appropriate Final Step Company representative that the safety problem is being referred to the grievance procedure.

Union Safety representatives specified in (1) of Stage 3 of this Section 8.3, if desiring to investigate or discuss a condition of safety other than as provided above, may request their Supervisor to notify the Company Safety Supervisor. The Safety Supervisor, or his designated representative, upon such notification, will meet the Union Safety Representative, without undue delay, at his place of work to discuss the matter involved or, if mutually desired, to jointly investigate the condition in question.

Notwithstanding the above, and in addition to the other provisions of this Section, if a condition or practice comes to the attention of a Union Safety Representative which he believes to constitute a serious hazard which immediately imperils the health or safety of an employee or employees, the Union Safety Representative is authorized to bring such condition or practice to the attention of the Factory or Division Manager or in Local 2096 the Department Manager or, if he not be present, to his designated representative, immediately.

It is understood and agreed that conditions challenged as unsafe, safety complaints, alleged violations of the safety provisions of this Agreement, etc., will not be processed through the First or Second Steps of the grievance procedure.

Nothing in this Section 8.3 shall be construed to restrict the employee's rights under Section 502 of the Labor Management Relations Act of 1947.

(8.4) Whenever a physical examination or laboratory test has been made of an employee by physicians acting for the Company, a report thereof will be given to the employee and/or the personal physician of the employee involved upon the written request of such employee. However, if such examination or test discloses an abnormal condition, the employee will be so advised.

The Company will provide pulmonary function tests for all Mapleton employees, once every two (2) years. For those employees who, because of the nature of their work, are required by the Company to wear respirators, the Company will provide medical evaluations of the respiration functions on at least an annual basis.

Once during the term of the current agreement the Company will make available, on an off-shift basis, appropriate physical tests (as deemed by the plant medical director) to employees in welding classifications and employees in classifications that perform oil quench heat treat operations.

(8.5) Whenever it is determined by Company monitoring or tests that employees have had exposure exceeding the permissible level as set forth in 29 CFR 1910.1000 Air Contaminants, Code of Federal Regulations, such information shall be provided in writing to the Director of the Business

Unit Union Safety Committee or, in the Peoria area, to the appropriate Business Unit Union Safety Representative.

- (8.6) If, as the result of an employee complaint, the Company conducts a test of noise, air contaminants or airflow, the results of such tests shall be explained to the employee involved and to the Director of the Local Union Safety Committee or, in the Peoria area, to the Business Unit Union Safety Representative.
- (8.7) In addition to the other provisions of this Article 8, if a condition comes to the attention of the Director of a Business Unit Union Safety Representative (or in the Peoria area, the Business Unit Union Safety Representative)
  - (i) which involves noise, air contaminants or air flow, and
  - (ii) which he feels constitutes a health or safety hazard to employees, and
  - (iii) about which he has been unable to obtain a satisfactory explanation or response from Company Safety representatives,

such Director prior to the filing of a grievance in the Final Step of the grievance procedure may submit a written request to the Central Committee on Health and Safety that such Committee conduct a joint investigation in accordance with item 3 of Section 8.13. If such an investigation is conducted, the Director of the Business Unit Union Safety Committee and/or the Local Union President or his designated representative will be allowed to meet with the investigating Central Committee members before and after such investigation in order to explain his complaint and to receive an explanation of the Central Committee's findings.

- (8.8) The Company shall be the sole source of safety glasses approved for wear in designated safety glasses areas. Such safety glasses shall be provided by the Company at no cost to the employee subject to the following:
  - a. For nonprescription safety glasses, the Company shall furnish the first pair without cost to the employee and such glasses shall remain the property of the Company. Replacement pairs, if necessary, shall be at the employee's expense except as provided in (c) below.
  - b. For employees requiring them, prescription ground safety glasses shall be provided at no cost to the employee; provided that such safety glasses shall, except as provided in (c) below, be furnished upon receipt by the Company of a revised prescription. However, such replacements due to revised prescriptions will not be furnished more often than once per year. Such glasses shall thereafter be the sole property of the employee. A prescription for ground safety glasses will be accepted by the Company when the prescription is based on an examination made by a qualified eye doctor within the two preceding years, provided that the Company shall in no instance be obligated to pay any part of the cost resulting from a prescription for nonstandard frames, special temples, tinted or any other type of special glasses not required by reason of the employee's work at the Company.
  - c. When the nature of an employee's work results in damage to either nonprescription or prescription-ground safety glasses to the extent that the Company's Safety Supervisor advises replacement, the replacement cost will be borne by the Company.

- d. On either nonprescription or prescription ground safety glasses, if an employee elects not to choose from the selection of standard no-cost frames, the Company will contribute \$14.00 towards the purchase of nonstandard frames from the Company-approved source. If an employee desires pairs of glasses in addition to those provided in (a), (b) or (c) above, the employee may purchase them from the Company at the Company's cost.
- (8.9) When the Company determines that the nature of a job requires the wearing of special protective garments or safety devices, other than safety glasses, the Company will furnish the equipment without cost to the employees and will require the wearing or use of such safety equipment as a condition of employment.
- (8.10) When the Company determines that the nature of the job requires the wearing of safety shoes with steel toes; employees will be required to purchase safety shoes with steel toes. Where such purchase is made at a Company-approved source, the Company will contribute the sum of \$150.00 annually toward the purchase of each pair of such shoes. When the Company determines that the nature of a job requires the wearing of metatarsal protection; employees shall have an option of wearing auxiliary equipment provided by the Company or of purchasing safety shoes with special metatarsal protection. In the event employees elect to purchase safety shoes with metatarsal protection, and where such purchase is made at a Company-approved source, the Company will contribute the sum of \$45.00 toward the purchase of each pair of such shoes.

The Company reserves the right to increase the amount(s) provided for safety shoes with steel toes and / or metatarsal protection within any business unit or all of the business units.

When the nature of an employee's work results in damage to safety shoes to the extent that the Company's Safety Supervisor advises replacement, the replacement cost will be borne by the Company.

(8.11) The Company and the Union recognize the obligation imposed upon the parties to this Agreement by the Occupational Safety and Health Act. When an employee, who has been designated for this purpose as a representative of the employees in the bargaining unit, accompanies an OSHA inspector on an official plant inspection tour at the inspector's request, he shall not lose pay for regularly scheduled hours during such tour. More than one employee may participate as a designated representative but no more than one such designated representative shall act in such capacity at any one time and no more than one such designated representative will be paid for the same hours.

Notwithstanding the previous paragraph, when two OSHA inspectors make simultaneous and separate inspection tours, one designated representative may participate with each OSHA inspector and they shall not lose pay for regularly scheduled hours during such tours.

(8.12) As essential and valuable as the parties agree the local safety complaint procedure is in resolving questions and disputes, the parties recognize the potential advantage of a Central Committee of health and safety professionals. Such a Committee, operating in a non-adversarial environment, could provide a valuable means of communications between the parties on general matters of health and safety. Therefore, the parties agree to the establishment of a Central Committee on Health and Safety. This Committee will consist of three representatives of the International Union and three representatives of the Company. At least one of the members appointed by each party will have had professional training in industrial hygiene and at least one other member appointed by each party will have had experience and/or training in industrial safety.

The Committee shall select a Chairman and a Secretary from among the members serving. When a Company representative is Chairman, a Union representative shall be Secretary and vice versa.

The Committee will meet as frequently and at times as may be mutually agreed upon. The Secretary will keep minutes of each meeting, copies of which will be provided to the Company and Union representatives on the Committee within a reasonable time following each meeting.

- (8.13) The functions and objectives of the Central Committee on Health and Safety will be:
- 1. To encourage the observation of safety rules and the furtherance of the safety program.
- 2. To review serious or unusual injury and illness experience of plants covered by this Agreement and recommend possible corrective measures where appropriate.
- 3. To conduct, at the request of either party, a joint investigation to measure noise, air contaminants or airflow in facilities and areas covered by this Agreement when and where conditions indicate such measurement is necessary. Arrangements for such investigations shall be made in advance and the investigations shall be conducted within a reasonable period of time.
- 4. To review the causes of major accidents or occupational illness clusters, which occur in the work place and fatalities resulting from work-related injuries and make recommendations where appropriate.
- 5. To review established handling procedures, protective measures and emergency procedures for all recognized harmful physical agents or chemicals to which employees covered by this Agreement are exposed and make recommendations where appropriate.
- 6. To review significant developments of mutual interest in the industrial health and safety fields and consider the applicability of such to facilities or areas covered by this Agreement.
- 7. To review significant changes in the Company's health and safety programs due to legal requirements or Company initiated revisions and make recommendations where appropriate.
- 8. To review and recommend changes in employee education and training programs related to health and safety.
- 9. To develop and implement a training program to be conducted annually during the term of this Agreement, consisting of twenty four hours of training over three days (eight hours of which will be of the Union sharing best practices in health and safety) each contract year, specifically designed for local Union Safety Committee Directors (or, in their absence, alternate Directors). No later than 45 days prior to the annual safety training, the parties will share the topic(s) that they intend to present in current year's training session.
- 10. Whenever it is determined by Company monitoring or tests that employees have had exposure exceeding the permissible level as set forth in 29 CFR 1910.1000 Air Contaminants, Code of Federal Regulations, such information shall be provided in writing to the Central Committee on Health and Safety.
- 11. The Company will provide to the Union Central Safety Committee a copy of each facility's

report on OSHA Form #300 and the man-hours worked at each such facility during the period covered by such report.

## ARTICLE 9 Vacation and Separation Pay

- (9.1) Each eligible employee covered by this Agreement shall be paid, in the manner provided herein, for vacation hours used during each "base period".
- (9.2) Each employee having less than ten years of service with the Company at the end of the month, in which the "base period" terminated, shall be eligible for 120 vacation hours. Each employee having ten or more years of service with the Company at the end of the month in which the "base period" terminated, shall be eligible for 160 vacation hours. Each employee having twenty or more years of service with the Company at the end of the month in which the "base period" terminated, shall be eligible for 200 vacation hours.

Except in the event of a national emergency, those departments or plants in which summer shutdowns are contemplated will close the department or plant for up to one week for the purpose of giving employees their vacations. Such vacation period(s) will occur during the period between June 15 and September 15 of each year. Management shall notify employees about the specific dates for their business unit not less than 60 days prior to the first day of a vacation period. In any event, such notification shall be made by May 15 of each year. In any national emergency, deviations from the above-stated intentions will be discussed in advance with the Union.

The Christmas "vacation period" for those departments or plants in which a shutdown may be contemplated will be the seven calendar days from Christmas Day (or the day observed as the Christmas Day holiday) of one year through New Year's Day (or the day observed as the New Year's Day holiday) the year following.

- (9.3) All vacations shall be taken during "vacation periods" except for the following:
- a. Vacations of employees required for urgent work and vacations of employees in certain departments responsible for rendering service to customers and maintenance of plants and properties.
- b. The additional vacation hours of those employees eligible for more than 80 vacation hours.
- c. Vacations of employees required to perform Temporary Military Service (as defined in Section 15.4) during a "vacation period."

Vacation time under the conditions of (a), (b), and (c), above and all other vacation hours if there are no vacation periods shall be taken during the "base period" at such times as will cause the least inconvenience in the departments in which these employees work and at times to be approved by the department Supervisor. Employees may be required to submit requests for vacation days electronically. Employees will be provided confirmation of their electronic vacation request.

Employees whose services are required during "vacation periods" will not be asked to waive vacation entitlement, or any part thereof, in the scheduling of such work.

An employee will not be required to work a vacation period unless he has been so notified at

least 30 calendar days prior to first day of the period he is required to work.

- (9.4) The "base period" during which an employee may take vacation hours is from the first Monday in the first full pay period in January, one year, through the Sunday prior to the first Monday in the first full pay period in January, the year succeeding (January 1, 2024; January 13, 2025; January 12, 2026; January 11, 2027; January 10, 2028; January 8, 2029).
- (9.5) The vacation payment shall be an amount in addition to the employee's regular wages and shall be computed as a percentage of the total amount of such regular wages earned by the employee during the preceding payroll year less required deduction. The percentage to be used in computing an employee's vacation payment shall be as set forth in the schedule below:
  - a. For an employee having less than 10 years of service with the Company at the end of the month in which the "base period" terminates, the applicable percentage shall be six percent.
  - b. For an employee having 10 or more years of service with the Company at the end of the month in which the "base period" terminates, the applicable percentage shall be eight percent.
  - c. For an employee having 20 or more years of service with the Company at the end of the month in which the "base period" terminates, the applicable percentage shall be ten percent.

Pay for vacation time shall be deducted from the employee's vacation payment as vacation time is taken and shall be made at the employee's then current straight-time hourly rate for each authorized vacation hour until the vacation payment is exhausted. Vacation pay shall be included with the employee's regular pay for the pay period in which the vacation time was taken. Payment for any remaining portion of the employee's vacation payment will be made to the employee following the end of the base period and no later than February 15.

- (9.6) If an employee is separated from the payroll during the "base period," such employee will be paid for any remaining vacation payment as a separation payment.
- (9.7) Separation payments will be made at the time the employee is paid his last paychecks, whenever that is possible, but in any event not later than two weeks after the date of separation.
- (9.8) As used in this Article 9, "service" for employees hired prior to December 11, 1970, is the measurement of all periods of full-time employment with the Company, its predecessors and subsidiaries. It includes all periods of authorized leaves as provided hereafter in Article 14 and periods of layoff since his last date of hire. For employees hired on or subsequent to December 11, 1970, "service" is the measurement of all periods of full-time employment with the Company since his last date of hire including all periods of authorized leaves as provided hereafter in Article 14 and all periods of layoff.

#### **ARTICLE 10 Group Seniority**

(10.1) Within each of the bargaining units represented by Locals Nos. 751, 974, and 2096 as set forth in Section 2.1 of this Agreement, there shall be skilled trades groups (hereinafter referred to individually as "group") as set forth in each Exhibit B to the Local Agreements for such bargaining units.

Group seniority shall be measured and applied in accordance with applicable provisions of such Local Agreements.

(10.2) Article 12 and Article 13, except to the extent provided for in a Local Agreement described in Section 10.1, shall not be applicable to an employee covered by this Article 10.

#### **ARTICLE 11 Seniority**

(11.1) Employees shall be regarded as probationary during the first ninety days of work by such employee. No claim of wrongful layoff or discharge of a probationary employee shall be taken up as a grievance, unless such claim is based on alleged personal prejudice or Union activity during such ninety days worked. Such grievances must be in writing and supported by written evidence. The foregoing provision shall not prohibit a probationary employee from filing a grievance alleging a violation of other contractual provisions applicable to such employee.

If a probationary employee is retained beyond the ninetieth day worked, his seniority shall revert back to his last date of hire.

However, an employee who is separated or is on layoff from one business unit covered under this Agreement and hires into another business unit covered under this Agreement shall not be required to serve a probationary period at the new location, provided he has served a full probationary period in any other business unit covered under this Agreement and further provided he applies for work at that new location not later than 60 calendar days following his last day worked at his prior location.

(11.2) Seniority in one of the business units specifically set forth in Section 2.1 of this Agreement shall be measured from an employee's last date of hire into the bargaining unit or the unit which currently constitutes such bargaining unit.

If an employee who has never worked in a bargaining unit as described above transfers into any such bargaining unit, either at his request or the Company's request, his "date of hire" will be considered the date of such transfer. Seniority with respect to any one such bargaining unit is broken for the following reasons only:

- 1. If the employee quits.
- 2. If the employee is discharged and not reinstated.
- 3. If the employee refuses, while laid off, to accept a recall to work (issued by the Company), or fails to report for work in accordance with such recall within ten calendar days after the mailing by certified letter mailed to the employee's last known address. The Company will grant an extension to the ten-day period if, prior to the expiration of the ten-day period, satisfactory reason is given for the employee's inability to immediately report for work.
- 4. If an employee has been laid off for a period greater than his accumulated seniority at time of layoff or three years, whichever is less.
- 5. If an employee fails, unless satisfactory reason is given, to report for work by his fourth scheduled workday following a temporary layoff under the provisions of Section 13.6 of this

- Agreement or any other layoff where the return to work date has been announced prior to the beginning of such layoff and such employee has received written notice of such date.
- 6. If an employee (other than a laid off employee whose seniority has not been broken under the foregoing provisions of this Section 11.2) accepts employment at a location that is not within the geographical limits of the bargaining unit in which he possesses seniority.

The seniority of an employee in a bargaining unit shall be applied as provided in the Local Agreement covering that bargaining unit.

(11.3) An employee who is transferred from a job in a bargaining unit to a job outside of the bargaining unit shall not accumulate seniority during the period he remains in a job with the Company outside of that bargaining unit but within the geographical limits of that bargaining unit.

Upon an offer by the Company to him of the opportunity to return to that bargaining unit, he may exercise the seniority he accumulated prior to his transfer to a job outside of the bargaining unit in accordance with the applicable provisions of this Agreement and the applicable Local Agreement.

- (11.4) If an employee is laid off from a business unit, the employee's seniority shall accumulate during the layoff subject to the provisions of Section 11.2 of this Agreement.
- (11.5) Quarterly seniority lists covering employees in a business unit (active or on leave of absence) shall be provided by the Company to the Local Union representing that business unit. These lists, by IO, NIO or skilled trades groups will give each employee's name, factory number, department number, shift, rate of pay, and seniority date or group seniority date, whichever is applicable, and appropriate job classification code numbers if used. A quarterly list covering supplemental employees in a business unit shall also be provided by the Company to the Local Union representing that business unit. This list will give each employee's name, factory number, department number, shift, rate of pay, and appropriate job classification code numbers if used. Upon the request of the Grievance Committee Chairman, the Company will provide a list of nontraditional assignments. Such requests shall not be made more than once per month.
- (11.6) By written mutual agreement of the parties, the seniority rules herein set forth may be waived in order to provide employment for the disabled.
- (11.7) When an employee enters a department, the Supervisor, upon request of the Steward, will, within one working day after the request, provide the employee's seniority date, and job classification.

# **ARTICLE 12 Job Openings**

- (12.1) When an opening occurs, the Company may elect to fill it by job posting, reassignment, or new hire, subject to the succeeding provisions of this Article.
- (12.2) In the event the Company elects to fill a job opening by job posting, employees who file or have filed an application on a form provided by the Company (hereinafter referred to as a Bid) for such opening will be considered in accordance with the provisions of Sections 12.3 through 12.10. Notwithstanding the foregoing, the job posting procedure will not be used for job openings in Labor Grade 1, by an employee who is being retained at work by virtue of preferential seniority,

or by an employee who is scheduled for layoff and is being retained at work under Section 13.5 (f). On or after January 1, 2019, the Company may, at its sole discretion, implement an electronic bid system.

- (12.3) A notice (designated as "Notice of Job Opening") of the existence of such openings will be posted on bulletin boards reserved for that purpose or electronic displays in the facility in which the opening exists and will be posted by noon on Thursday of one week until noon on Tuesday of the following week. The Notice of Job Opening will:
  - a. identify the job classification, labor grade, shift, department number, and seniority unit of the opening, and
  - b. indicate the effective time and date the posting period begins and ends.
- (12.4) Except as provided in Sections 12.5 and 12.6 below, an employee actively at work within the business unit in which the opening exists will be eligible to file a Bid for such opening provided the employee has not moved in accordance with the job posting procedure in the prior twelve months.

For the purposes of this Section 12.4, employees will be considered "actively at work" during a temporary layoff under Sections 13.2 or 13.6, vacation, or absences under Sections 15.1, 15.3, 15.4 15.5, or 15.6.

- (12.5) Except for openings in the same seniority unit and either in a higher Labor Grade or on the same job but on a different shift, an employee will be ineligible to move in accordance with the job posting procedure for a period of twelve months after:
  - (i) recall; or
  - (ii) new hire.

Notwithstanding this Section 12.5, an employee recalled or hired into an opening in an IO will be afforded one opportunity to move in accordance with the job posting procedure to an opening in a higher Labor Grade in any NIO during such twelve-month period.

Notwithstanding this Section 12.5, an employee recalled into an opening in an NIO will be afforded one opportunity to move in accordance with the job posting procedure to an opening in an equal or higher Labor Grade to a different NIO during such twelve-month period.

- (12.6) An employee who is demoted for lack of skill and ability or who is medically placed as a result of temporary medical restrictions will be considered, for the purpose of bidding eligibility, to have been placed under the appropriate provisions of the job posting procedure and will be subject to the bidding restrictions contained in the appropriate Sections 12.4 and/or 12.5 above.
- (12.7) All bid restrictions contained in Sections 12.4, 12.5 and 12.6 shall be waived if an employee is moved to a different classification as the result of a reduction in force subsequent to the action which caused the restriction. The bid restriction contained in Section 12.4 shall also be waived if an employee is moved to a different shift as a result of a reduction in force subsequent to the action, which caused the restriction.
  - (12.8) The ratio of openings posted under the provisions of Section 12.2 above to openings filled

by recall or new hire at Labor Grade 2 and above shall be at least 1:2 during any calendar year within a business unit.

- (12.9) An employee eligible to file a Bid as provided in Section 12.4 above may do so by completing a Bid form for each desired opening, not to exceed five on file and active at any one time, and submitting it in a manner to be prescribed within each facility. A Bid will be rejected if it:
  - a. is submitted after the expiration time and date of the posting period; or
  - b. exceeds the limitation of five being filed or on file at any one time; or
  - c. does not adequately identify the employee and the job for which the employee is bidding; or
  - d. is not in accordance with the restrictions contained in Sections 12.4, 12.5 and/or 12.6.

A properly submitted Bid shall remain active until the Company selects an employee for such posted opening, until the filling of the opening by job posting is otherwise canceled, or until the employee notifies the Company that the Bid is to be withdrawn. Any withdrawal of a Bid must be accomplished during the posting period.

(12.10)

- a. When an opening within a facility is to be filled using the job posting procedure, the Company will consider all Bids properly submitted during the posting period. Job openings will be filled in the order that the openings appear on the Notice of Job Opening.
- b. Employees filing such properly submitted Bids for jobs in Labor Grade 2 will be considered on the basis of their seniority. Employees filing such properly submitted Bids for jobs in Labor Grade 3 and above will be considered on the basis of their job qualifications and seniority. When job qualifications are approximately equal, then the employee with the greatest seniority shall be selected and assigned to fill such opening. Once selected, the name and identification number of the successful bidder will be posted on bulletin boards identified in Section 12.3. As used in this Section 12.10(b), an employee's job qualifications will not include experience gained from temporary assignments to higher rated jobs. However, an employee who has completed a Joint Training program or a Company-approved educational program for a specified area will be considered at least equally qualified with employees who have worked in that area.
- c. An employee who has an active Bid for any opening and who is selected to fill such opening in accordance with the foregoing provisions of this Section 12.10 will be required to accept such assignment.
- d. If, for real and practical considerations, the Company decides not to fill a posted opening using the job posting procedure, a Notice of Job Opening may be canceled by indicating the reason for the cancellation on the Notice of Job Opening.
- e. Successful bidders must meet any basic qualifications, which have been established for that particular job. Such basic qualifications will be consistent with past practice and could include the ability to read, write, speak English, type, perform basic mathematics or other criteria, which are reasonably related to certain classifications and/or certain jobs. It is

understood that for job openings in Labor Grade 4 and above, basic qualifications will be of a significantly higher standard and may include specific formal training, graduation from an appropriate apprenticeship program and/or extensive experience in the same or related field.

- f. Successful bidders must meet any reasonable physical requirements, which are necessary in order to perform that particular job.
- g. If, for any posted job opening, no Bids are received from successful bidders within the posting period, the senior employee on layoff from that business unit (if such there be) who possesses the basic qualifications and physical requirements will be recalled to the opening. If there are no employees on layoff who possess the basic qualifications and physical requirements, the Company may elect to fill the opening by other means such as reassignment, new hire, reduction in force, apprentice placement, etc.
- h. An employee on layoff who refuses recall in accordance with paragraph (g) above shall be separated as a quit.
- i. Successful bidders will be moved as soon as business conditions allow not to exceed 90 days.
- (12.11) In the event the Company elects to fill a job opening by reassignment, the reassignment shall be based on real and practical considerations, without discrimination; if the reassignment is not based on such reasons, then it shall be made on the basis of seniority.

As used in this Section 12.11, "reassignment" means the assignment of an employee (1) to a job opening in the same classification on the same shift within the same business unit but under a different supervisor, or (2) to a job opening on the same shift but to a different classification within the same Labor Grade.

(12.12) In the event the Company elects to fill a job opening in Labor Grade 3 and above by conducting a canvass, employees who meet all the application requirements will be considered on the basis of their job qualifications and seniority.

When all factors other than seniority are approximately equal, the employee with the greatest seniority shall be selected.

In the event the Company elects to fill a job opening in Labor Grades 1 and 2 by conducting a canvass, employees who complete all the application requirements will be considered on the basis of seniority.

This selection process does not apply to Team Lead or Non-Traditional positions.

(12.13) An employee will not be hired into a job in Labor Grade 1 or 2 in any of the business units identified in Section 2.1 if there are employees on layoff from that business unit who possess the basic qualifications and physical requirements, which exist for that job. It is understood that employees recalled under the above sentence may receive training, not to exceed five or ten days, as appropriate. An employee will not be hired into a job in Labor Grade 3 or higher in any of the business units identified in Section 2.1 if there are employees on layoff from that business unit who have previously held the job in question.

An opening will not be filled by recall or new hire if there is an employee with greater seniority

currently working on the same job but on a different shift and whose Prefiled Shift Preference form indicates the shift of first preference is the same as the shift of the opening. Instead, the opening shall be filled by placing the senior employee already on the same job on a different shift, with such Prefiled Shift Preference form on record for at least forty calendar days, into the opening. No more than two shift changes shall be made under this paragraph for each original opening.

(12.14) Each week the Company will follow established local procedure in providing the Union with a list, in writing, of employees moved to fill openings through job posting, reassignment, temporary assignment and/or new hire during the preceding week. Such list shall include the seniority and/or group seniority date and identification number of each such employee as well as the job classification, labor grade, shift, department number, and seniority unit of the job to which such employee is moved.

Upon request of the Grievance Committee Chairman, and not more than one time per calendar month, the Company will provide a list of employees who have terminated (retired, discharged, separated, or deceased) during the month the request was made.

#### (12.15) Preferential Hire

#### A. An employee

- (i) who has been on a scheduled layoff (as defined below) from one of the bargaining units covered by this Agreement for a continuous period of at least six months, and
- (ii) who possessed two or more years of seniority and/or group seniority (if applicable) in such bargaining unit at the beginning of such layoff under (i) above may file an application for employment in any other bargaining unit covered by this Agreement and will (subject to the succeeding provisions of this Section 12.15) be granted a preferential hiring right to a job opening in such other bargaining unit provided such employee possesses the necessary qualifications for employment in such other bargaining unit and further provided that another applicant for employment in such other bargaining unit does not possess measurably better qualifications for a specific job opening than such laid-off employee.
- B. An application for employment submitted by any such laid-off employee must be filed, electronically, with the Company through its standard hiring procedures and with Corporate Labor Relations specifying the position for which the laid-off employee is seeking employment. Employees may exercise the preferential hiring rights under this Section 12.15 for up to two years from the date of their layoff. Such employees will be subject to the standard pre-employment drug screen.
- C. Any such employee who accepts an offer of employment under the provisions of this Section 12.15 will not be required to serve a probationary period under Section 11.1 of this Agreement and shall be deemed to have voluntarily quit from employment in the previous bargaining unit on the day prior to the date on which such employee begins employment in the new bargaining unit. Such employee shall also be deemed to have met the employment and/or seniority eligibility requirements for Sections 15.1, 15.2, 15.3, and 15.4.

- D. For purposes of this Section 12.15, a scheduled layoff is a layoff, which results from a reduction of schedules to reduce or avoid an increase in inventory of finished products because of sales prospects.
- (12.16) For the purposes of this Article 12, "business unit" or "facility" shall be as defined in Section 4.7.

# ARTICLE 13 Reductions in Force and Layoff

- (13.1) In the application of the seniority provisions of this Article, a layoff shall be considered an interruption of employment caused by a reduction in force due to a reduction in regular operating schedules, but shall not include temporary layoffs which are due to material shortage, equipment failure, power failure, plant rearrangement or retooling, labor dispute, or other circumstances which cause a temporary cessation or reduction in operations.
- (13.2) In the event a temporary layoff due to material shortage, equipment failure, power failure, plant rearrangement, retooling, labor dispute, or other circumstances, exceeds ten working days, the Company will meet with the Union to consider the feasibility of applying the regular seniority provisions.
- (13.3) Employees who are scheduled to be laid off because of a reduction in the force will be given advance notice of not less than five days counting the scheduled day of layoff. (Employees who are scheduled to be laid off on Friday will be given notice of such layoff no later than Monday of the same week.) The Union shall be given a list of employees scheduled to be laid off during the above referred to five-day period after employees have been notified. The Union will also be given a list of employees who are recalled to work.
- (13.4) A laid-off employee is one who possesses seniority in a business unit, but who, in connection with a reduction in force, has not been entitled to be placed or retained on any job in that business unit and who as a result is not working on a job in that business unit.

"Seniority unit" is either an Interchangeable Occupational Group (IO), a Non-Interchangeable Occupational Group (NIO), or a skilled trades group.

- (13.5) In the event of reduction in force or layoff, employees will be placed or laid off in accordance with the following principles:
  - a. Within each of the business units identified in Section 4.7, there shall be Non-Interchangeable Occupational Groups (NIO's) and an Interchangeable Occupational Group (IO). The number and definition of NIO's shall be as set forth in each respective local agreement. The IO shall contain, except by mutual agreement, all job classifications within Labor Grades 1 and 2. Within each business unit, there may also be one or more mutually agreed Specialized Classifications (SC) and/or Specialized Jobs (SJ).
  - b. Within an NIO, if it becomes necessary to reduce the number of employees within a classification, the least senior employee(s) within the classification will, seniority permitting, displace the least senior employee(s) within the NIO on the same or lower labor grades. If several employees are to be simultaneously placed, the Company may displace the least senior employees in a manner, which takes maximum advantage of employees' previous

experience and ability.

- c. Within an NIO, if it becomes necessary to reduce the number of employees within a classification and the least senior employee(s) cannot be placed under paragraph (b) above, the least senior employee(s) within the classification shall, seniority permitting, displace the least senior employee(s) within the NIO on any classification which the employee has successfully held. If several employees are to be simultaneously placed, the Company may place the least senior employees in a manner which takes maximum advantage of employees' previous experience and ability.
- d. Employees who are displaced from an NIO because of lack of seniority shall, seniority permitting, displace the least senior employee in any NIO in the same business unit on any classification which the employee has successfully held or for which the employee has completed a Joint Training program or Company-approved educational program. Employees displaced in accordance with this paragraph (d) will be placed in accordance with the provisions of this Section 13.5. If several employees are to be simultaneously placed, the Company may displace the least senior employees in a manner which takes maximum advantage of employees' previous experience and ability.
- e. Employees who are displaced from an NIO because of lack of seniority shall, seniority permitting, displace the least senior employee(s) within the IO. If several employees are to be simultaneously placed, the Company may displace the least senior employees in a manner, which takes maximum advantage of employees' previous experience and ability.
- f. Within the IO, if it becomes necessary to reduce the number of employees within a classification, the least senior employees within the classification will, seniority permitting, displace the least senior employee(s) within the IO. If several employees are to be simultaneously placed, the Company may displace the least senior employees in a manner which takes maximum advantage of employees' previous experience and ability. The least senior employee(s) within the IO will be scheduled for layoff.
  - Movement under this Section 13.5 may be metered in order to minimize disruption and provide for the continued efficient operation of the facility; however, employees will be placed within 75 calendar days of their scheduled date of layoff. This period may be extended by mutual agreement of the Local Union and business unit management.
- g. Notwithstanding paragraphs (b), (c), (d), (e), and (f) above, employees within an SC may only be displaced by other employees within the same SC. Notwithstanding paragraphs (b), (c), (e), and (f) above, employees within an SJ may only be displaced by other employees within the same SJ or by other employees within the same NIO who have previously performed the job in question.
- h. An employee scheduled for reduction in force or layoff may, at the Company's discretion, be retained for the purpose of and for the period of replacement training.
- i. In the case of all placements under paragraphs (b), (e), or (f) above, employees must possess all basic qualifications and abilities necessary to perform the job within a reasonable period. For the purposes of this Article 13, the reasonable period during which an employee will be expected to become a productive and independent performer will be as follows:

For placement into a job in Labor Grade: period will be: 4 and above 20 work days 2 and 3 15 work days 5 work days

Employees must also possess the necessary physical ability to perform the job. If an employee does not possess all the basic qualifications and abilities and/or does not possess the physical ability, the Company shall proceed to the next step in the reductions in force and layoff procedure. For certain jobs, the above time periods may be extended by mutual agreement between the Local Union and Business Unit Management.

As used in this Section 13.5, "basic qualifications" are as discussed in Section 12.10(e).

- j. For the purposes of this Article 13, the business units shall be defined as in Section 4.7.
- k. Employees may make written application indicating their preferred shifts (in order of preference) on a form provided by the Company and submitted in a manner to be prescribed within each business unit. In order to be effective, such form must be filed at least forty calendar days prior to the effective date of a reduction in force or layoff. Once filed, a form will remain effective until withdrawn or modified by the employee.

If an employee elects to file such a form, it will be used after the determination is made as to which classification the employee is to be assigned in accordance with the preceding provisions of this Section 13.5. The employee shall be placed, seniority permitting, on their shift of highest preference.

If an employee does not file such a form, the employee will have no shift preference rights.

- 1. Notwithstanding the foregoing provisions of this Article 13, because they work at training stations established for that purpose, employees in the Training Department may not be displaced by the previous provisions of this Section 13.5.
- (13.6) In the event a temporary reduction within a plant or facility (as defined in Section 4.7) is scheduled, employees may be placed on temporary layoff without regard to Section 13.5 provided no employee shall be placed on such temporary layoff in excess of 14 weeks in any year (the first Monday in April of one year to the first Monday of April the following year). Up to 25 days of these 14 weeks for any individual employee may be used in increments of less than a full week.

For a temporary layoff of less than one week under the provisions of this Section 13.6, employees shall be paid, in addition to pay for hours worked during such workweek, an amount equal to \$85 for each work day on temporary layoff during such workweek minus required deductions.

For purposes of this Agreement a week (or calendar week) is defined as a period of seven consecutive days beginning on Monday.

(13.7) For the purposes of administering this Article 13, the seniority provisions of the appropriate Local Agreement shall apply with respect to the seniority status accorded Union Representatives.

(13.8) Each employee who has more than three years of seniority and/or group seniority at the time of being placed on layoff shall be eligible for a layoff payment as hereinafter provided. An eligible employee shall be entitled to a layoff payment of \$170 minus required deductions for each full week such employee is placed on layoff or temporary layoff. Such payment(s) shall be made in the pay period immediately following the week(s) that the employee was on layoff or temporary layoff. An eligible employee shall be entitled to layoff payments under the provisions of this Section 13.8 for a maximum of 52 weeks during the life of this Agreement.

## **ARTICLE 14 Leaves of Absence**

(14.1) Leaves of absence shall be granted automatically to employees who, because of physical or mental disability, are unable to work and who provide the Company with proper notice and evidence of such disability.

If the leave of absence is not approved, or, if approved, is later canceled by the Company, the employee shall be so notified in writing with a copy of such notice given to the Union.

Any dispute arising from this action shall be presented, in writing, directly to the Final Step of the appropriate local grievance procedure and, if applicable, the provisions of Section 5.7 shall be utilized. It is understood that an employee will not be discharged while any such above-described dispute is being processed provided the employee and the Union, within 5 regularly scheduled workdays of receipt of such notice, process such dispute and cooperate in utilizing such procedure.

Notwithstanding the foregoing provisions of this Section 14.1, a disability leave of absence shall be in effect only for the period of the physical or mental disability and in any event the leave shall automatically expire upon completion of a leave period of two years, provided that any successive period of physical or mental disability not separated by a period of active full-time work of not less than 45 calendar days shall be considered a continuation of the previous period of disability.

Not later than 10 days prior to such automatic expiration, the Company will send a registered letter to the employee's last known address as shown on the Company records reminding him of the fact that his seniority is subject to being broken if he fails to return to work upon expiration of his leave of absence.

- (14.2) In accordance with the applicable provisions of Local Agreements, the Company will promptly return to work an employee who is able to return from a leave of absence.
- (14.3) Employees who fail to return to work upon expiration of a leave of absence shall be separated from the employment of the Company, unless satisfactory reason is given.
- (14.4) Except as otherwise herein provided, the granting of other leaves of absence by the Company shall depend (1) upon the reason for the requested leave of absence and (2) upon the need for the employee's uninterrupted services.
- (14.5) When a leave of absence is granted by the Company, it shall be verified, in writing, upon request of the employee.

All leaves of absence shall be without pay, except as expressly provided elsewhere in this Agreement, and seniority shall accumulate during such leaves.

An employee who accepts employment elsewhere during a leave of absence, without the consent of the Company, shall be deemed to have voluntarily quit.

An employee on disability leave will be advised by the Company of the Company's position regarding employment elsewhere while on such leave. An employee on such leave must advise the Company of any employment elsewhere and he will receive approval for the employment if it is not inconsistent with needed medical treatment.

Upon written request of the local Union President, the Company shall provide, not more than one time per calendar month, a list of all employees who either began leaves of absence or returned from leaves of absence. This list will include the dates such leaves began or ended.

- (14.6) Any employee who, in time of war or national emergency, is drafted or volunteers into the armed forces of the United States shall be granted a leave of absence, and will be accorded reinstatement rights, as provided by law then in force. Such employee shall also, when date of entry into the armed forces has been determined and established, be entitled to a leave of absence not to exceed the three weeks immediately prior to such entry.
- (14.7) Any employee who volunteers and is accepted for service in the Peace Corps shall, upon proper notice to the Company, be granted a leave of absence for a tour of such Peace Corps service. Upon his return from Peace Corps service, he shall be granted the same reinstatement rights as are provided by law then in force in respect to service in the armed forces of the United States.
- (14.8) An employee who becomes pregnant will, upon request, be granted a personal leave of absence of reasonable duration under Section 14.4 immediately preceding or following the period of disability associated with the pregnancy or birth if the employee's personal physician, after consultation with the Company's physician, recommends such additional leave as being beneficial for the health or well-being of such employee.
- (14.9) Any employee who is elected or appointed to a position with the Union or who is elected or appointed to a temporary national governmental position, or who is elected to a State office, or who is appointed to a temporary position in the Department of Labor or Industrial Commission of his State or the Department of Labor of the United States, shall, upon written request, be granted a leave of absence for the period of such service, provided that no more than the number specified in the appropriate Local Agreement shall be on such leave of absence at any one time.

#### An employee

- (i) who is elected to a public office in a town, municipality, township, county or comparable governmental body;
- (ii) who must occasionally take time off from work in order to perform the necessary functions of such office; and
- (iii)who obtains prior approval from the Company to be absent from work; shall be deemed to be on a leave of absence for any such full day of absence for such purpose.
- (14.10) The Company will grant time off, without pay, to employees for the purpose of attending Union meetings, provided (1) the Company is given advance notice who the employees will be and

the date and hour the time off will be taken, such advance notice to be given by the Union not later than noon on the workday preceding such time off, and further provided (2) that not more than the number of employees specified in the appropriate Local Agreement will be granted such time off during any one day. However, in cases of emergency and for the purpose of conducting Union elections, time off shall be granted employees upon request of the Union. Time off will also be granted for the purpose of meeting with Company representatives.

Temporary leaves of absence will be granted upon request of the Union for employees who are selected by the Union to attend state, national, or international Union conventions, regional or district Union Conferences.

Each day upon which an employee is granted time off under this Section 14.10 will be accumulated, by business unit, and totaled quarterly. Twenty such days will equal one month. For every such month (or prorated portion of a month) the Local Union will reimburse the Company an amount equal to the applicable full group rate then in effect under the applicable benefit plan referenced in the Benefits Agreement, plus the premiums for Short Term Disability coverage and Basic Life Insurance then in effect for each employee who was granted time off under this Section 14.10.

(14.11) Employees will be granted family and medical leave pursuant to the Family and Medical Leave Act of 1993 (as amended) (FMLA). FMLA leave will be used concurrently with leaves of absences covered under Short Term Disability, Workers' Compensation, or any other leave that would also qualify for FMLA leave.

## ARTICLE 15 Salary Plan

#### (15.1) Paid Absence Allowance

- A. Each employee while actively employed in a bargaining unit covered by this Agreement (but not while on layoff) on or after the first Monday in the first full pay period of April but prior to the first Monday in the first full pay period of January the year following is an "eligible employee" for purposes of this Section during the twelve-month period starting on that first Monday in the first full pay period of April and ending on the last day of the last pay period prior to the first Monday in the first full pay period of April the year following, but ceases to be an "eligible employee" in the event and at the time of quit, death, or discharge.
- B. With respect to each such twelve-month period, the number of hours of paid absence allowance credit to which an eligible employee at the beginning of that period shall be entitled is as follows:
  - a. For such eligible regular employee whose first day of work for the Company during such twelve-month period occurs on or after the first Monday in the first full pay period of April but prior to the first Monday in the first full pay period of July, the Paid Absence Allowance credit shall be 50 hours (April 10, 2023; April 8, 2024; April 7, 2025; April 6, 2026; April 5, 2027; April 3, 2028).

For such eligible full-time supplemental employee whose first day of work for the Company during such twelve-month period occurs on or after the first Monday in the first full pay period of April but prior to the first Monday in the first full pay period of July, the Paid Absence Allowance credit shall be 40 hours (April 10, 2023; April 8, 2024; April 7, 2025; April 6, 2026; April 5, 2027; April 3, 2028).

For such eligible part-time supplemental employee whose first day of work for the Company during such twelve-month period occurs on or after the first Monday in the first full pay period of April but prior to the first Monday in the first full pay period of July, the Paid Absence Allowance credit shall be 20 hours (April 10, 2023; April 8, 2024; April 7, 2025; April 6, 2026; April 5, 2027; April 3, 2028).

b. For such eligible regular employee whose first day of work for the Company during such twelve-month period occurs on or after the first Monday in the first full pay period of July but prior to the first Monday in the first full pay period of October, the Paid Absence Allowance credit shall be 37 1/2 hours (July 3, 2023; July 1, 2024; July 14, 2025; July 13, 2026; July 12, 2027; July 10, 2028).

For such eligible full-time supplemental employee whose first day of work for the Company during such twelve-month period occurs on or after the first Monday in the first full pay period of July but prior to the first Monday in the first full pay period of October, the Paid Absence Allowance credit shall be 30 hours (July 3, 2023; July 1, 2024; July 14, 2025; July 13, 2026; July 12, 2027; July 10, 2028).

For such eligible part-time supplemental employee whose first day of work for the Company during such twelve-month period occurs on or after the first Monday in the first full pay period of July but prior to the first Monday in the first full pay period of October, the Paid Absence Allowance credit shall be 15 hours (July 3, 2023; July 1, 2024; July 14, 2025; July 13, 2026; July 12, 2027; July 10, 2028).

c. For such eligible regular employee whose first day of work for the Company during such twelve-month period occurs on or after the first Monday in the first full pay period of October but prior to the first Monday in the first full pay period of January, the Paid Absence Allowance credit shall be 25 hours (October 9, 2023; October 7, 2024; October 6, 2025; October 5, 2026; October 4, 2027; October 2, 2028).

For such eligible full-time supplemental employee whose first day of work for the Company during such twelve-month period occurs on or after the first Monday in the first full pay period of October but prior to the first Monday in the first full pay period of January, the Paid Absence Allowance credit shall be 20 hours (October 9, 2023; October 7, 2024; October 6, 2025; October 5, 2026; October 4, 2027; October 2, 2028).

For such eligible part-time supplemental employee whose first day of work for the Company during such twelve-month period occurs on or after the first Monday in the first full pay period of October but prior to the first Monday in the first full pay period of January, the Paid Absence Allowance credit shall be 10 hours (October 9, 2023; October 7, 2024; October 6, 2025; October 5, 2026; October 4, 2027; October 2, 2028).

d. For such eligible regular employee whose first day of work for the Company during such twelve-month period occurs on or after the first Monday in the first full pay period of January, the Paid Absence Allowance credit shall be 12 1/2 hours (January 1, 2024; January 13, 2025; January 12, 2026; January 11, 2027; January 10, 2028; January 8, 2029).

For such eligible full-time supplemental employee whose first day of work for the Company during such twelve-month period occurs on or after the first Monday in the first full pay period of January, the Paid Absence Allowance credit shall be 10 hours (January 1, 2024; January 13, 2025; January 12, 2026; January 11, 2027; January 10, 2028; January 8, 2029).

For such eligible part-time employee whose first day of work for the Company during such twelve-month period occurs on or after the first Monday in the first full pay period of January, the Paid Absence Allowance credit shall be 5 hours (January 1, 2024; January 13, 2025; January 12, 2026; January 11, 2027; January 10, 2028; January 8, 2029).

C. During each such twelve-month period an eligible employee will be paid for all hours of absence, as defined below, up to his Paid Absence Allowance credit at his then current straight-time hourly rate. As used in this Section 15.1, "hours of absence" means hours during which an eligible employee is absent from work at his own volition, or because of accident, illness, extreme weather conditions or emergency, and during which he otherwise would have worked, but does not include any period of absence of less than two consecutive hours; any period of absence with respect to any one shift in excess of the regularly scheduled hours of that shift; any period of absence dealt with in Sections 15.2, 15.3, 15.4, 15.5 and 15.6 of this Article 15; any period of absence on Union business; any period of absence caused, either directly or indirectly, by any strike, slowdown, work stoppage, picketing (whether or not by employees covered by this Agreement), or concerted action, at a Company facility covered by this Agreement, or any dispute of any kind involving employees covered by this Agreement, or members of other locals of the International Union, United Automobile, Aerospace and Agricultural Implement Workers of America, who are employed by the Company; and any period of absence for which he is entitled to a payment or benefit under any other provision of this Agreement or under any provision of any other agreement (or benefit plan) between the parties hereto. Advance notice shall be given by an employee (whether or not an eligible employee) of any absence due to any of the reasons specified in the definition of "hours of absence" (whether or not such absence counts as "hours of absence") unless an emergency compels absence without reasonable opportunity to give such notice, in which event notice shall be given as soon as reasonably possible.

Any eligible employee who desires to schedule a full or partial day off where the need for time off is foreseeable and receive pay and approval for such absence under the foregoing provisions of this Section 15.1(C) may do so by filing a written request, on a form supplied by the Company, with his Supervisor at least one day in advance of the requested day off. For unforeseeable absences where the employee cannot give notice as provided above but notifies the Company 30 minutes prior to the start of the employee's shift, such absences will be approved. For unforeseeable absences during an employee's shift where the employee cannot give notice as provided above such absences will be approved provided the employee notifies their supervisor. Employees may be required to submit requests for paid absence days electronically. Employees will be provided confirmation of their electronic paid absence request.

D. Any employee who is an eligible employee throughout the last full pay period of any such twelve-month period shall be paid as an additional vacation bonus (in addition to any other pay to which he may be entitled) for the number of hours, if any, by which his Paid Absence Allowance credit for that twelve-month period exceeds his number of hours of absence

during that twelve-month period, such payment to be equal to the amount to which he would have been entitled under this Section 15.1 had he had hours of absence during that pay period equal in number to such excess. Any payments under this Section 15.1(D) shall be made to employees not later than June 15 of each year.

- E. Any eligible employee who, prior to the end of the last full pay period of any such twelve-month period, separates, quits, dies, retires, enters the armed forces (other than for temporary military duty), or is laid off from work because of a reduction in force, shall be paid (in addition to any other pay to which he may be entitled) for the number of hours, if any, by which his Paid Absence Allowance credit for that twelve-month period exceeds his number of hours of absence during that twelve-month period.
- F. An employee who receives a payment under either (D) or (E) above with respect to a twelvemonth period shall be entitled to no further rights under this Section 15.1 with respect to that period.
- G. No payment shall be made under paragraphs (D) and (E) of this Section 15.1 unless at the time thereof there are in effect satisfactory governmental rulings that such payment need not be included in the "regular rate" for purposes of the Fair Labor Standards Act or any similar law.
- H. If during the life of this Agreement, state, federal, or local law requires the provision of paid sick leave to employees covered by this Agreement, it is understood by the Parties that any entitlement to Paid Absence Allowance under this Section 15.1 will be used to meet any such requirement in whole or in part.

#### (15.2) Holidays

Every full-time employee (except employees specified in Letter of Agreement No. 27) covered by this Agreement who has been in the employment of the Company more than 30 days shall receive holiday pay for each of the holidays as referred to in Section 7.8 subject to the provisions below:

## (A) The rate of pay:

- (1) When the holiday occurs on an employee's regularly scheduled workday, his pay for such holiday shall be equal to one full, regularly scheduled day's pay (i.e., 8, 10, 12, etc. hours, whichever is applicable) at the employee's then current regular straight-time rate. This amount of holiday pay shall be applicable even if an employee works such holiday.
- (2) When the holiday occurs on a day other than an employee's regularly scheduled workday, then his pay for such holiday shall be equal to eight (8) hours pay at the employee's then current regular straight-time rate. This amount of holiday pay shall be applicable even if an employee works such holiday.

#### (B) The qualifying day:

To qualify for holiday pay, under this provision, an employee must work either the regularly scheduled workday immediately before or immediately after the holiday in the week in which the holiday falls.

- (C) To the above there shall be the following sole exceptions:
  - (1) If on the day before or after the holiday the employee works only part of his scheduled shift, the employee will still be entitled to the holiday pay if late less than one hour, or if, in a written claim to the Company, the employee is able to show that his reporting to work late or leaving work early resulted from urgent circumstances beyond the control of the employee. For consideration, such written claim is to be in the possession of the Company within ten days after the occurrence of the holiday for which holiday pay is claimed.
  - (2) If the employee is absent on a scheduled workday which is the sole qualifying day for holiday pay, he will still be entitled to the holiday pay if otherwise qualified, provided the employee, in a written claim to the Company, is able to show that the full day's absence resulted from urgent circumstances beyond his control (such as personal illness or injury, or death in the immediate family). For consideration, such written claim is to be in the possession of the Company within ten days after the occurrence of the holiday for which holiday pay is claimed.

Employees who work on a specified holiday will receive the usual overtime premium in addition to the holiday pay. However, an employee who is scheduled to work a specified holiday, but who fails to work as scheduled, shall not receive holiday pay unless, in a written claim to the Company, the employee is able to show that his failure to work as scheduled resulted from urgent circumstances beyond his control.

The provisions of this Section 15.2 are intended to apply only to those employees actively working for the Company at the time of holiday observance, returning to work from layoff on the qualifying day immediately following a Monday holiday, or absent for the sole reason of being on vacation, and shall not apply, for example, to any employee who, having been on layoff, (excepting those described above) or on leave of absence, has not returned to work prior to the holiday observance date; nor shall these provisions apply to any employee who, under a "Group Insurance Plan," enters or could have entered a claim for disability benefits which includes the holiday observance date as a day of disability, provided, however, if such holiday observance date, or either date in the case of consecutive days observed as holidays, occurs on an employee's first or last day of disability, such employee shall nonetheless receive holiday pay if such employee's first or last day of disability is recognized as a day of disability under said Group Insurance Plan (including days for which benefits are paid or would have been paid except for this provision and days counted as a waiting period before commencement of Short Term Disability Benefits).

#### (15.3) Jury Duty and Witness Service

Any full-time employee (excluding employees covered by Letter of Agreement No. 27) who has more than 30 days of seniority and/or group seniority (if applicable) and who either

- (i) is summoned and reports for jury duty in a Court of Record or Grand Jury, or
- (ii) is required by applicable law to appear for examination by a jury commission prior to such jury service, or
- (iii) is subpoenaed and reports for witness service in a Court of Record or Grand Jury

will be reimbursed by the Company, for each day on which he would otherwise have been scheduled to work, in accordance with the succeeding provisions of this Section 15.3.

- a. If he is absent for his entire shift because of such jury duty or witness service, he will be paid his regular scheduled day's pay at his straight-time hourly rate.
- b. If he performs such jury duty, witness service, or examination by a jury commission and works on the same day, he will be paid the difference, if any, between his actual earnings for that day and his regular scheduled day's pay at his straight-time hourly rate.
- c. Reimbursement under (iii) above will not be payable if the witness service is related to a matter in behalf of or as a result of his association with another employer or association.

Reimbursement to any employee under this Section 15.3 shall be payable only if the employee gives the Company prior notice of his summons or subpoena for jury duty, jury commission examination or witness service, and presents satisfactory evidence that jury duty, examination by a jury commission or witness service was performed on the day or days for which such reimbursement is claimed, and returns to work promptly on any day on which his jury duty, examination by a jury commission or witness service totals less than four hours and does not prevent him from completing on that day at least two hours of his regular shift, provided that any employee who serves on a jury and works on the same day will not be required to work more than four hours of his regular shift and further provided, with respect to an eligible employee who is assigned to the third shift, that only in the case of jury duty or witness service performed, the necessity for returning to or reporting for work as set forth above shall not be applicable either to the regular shift preceding, if any, or the regular shift following; if any, the day on which such summons or subpoena ordered him to report and he did report, but such exemption shall not include both days.

Within the Commonwealth of Pennsylvania only, the Magistrate Court shall be deemed to be a "Court of Record".

#### (15.4) Temporary Military Service

Each employee (excluding employees covered in Letter of Agreement No. 27) while actively employed in a bargaining unit covered by this Agreement (but not while on layoff) who is absent because of required performance by him of

- a. temporary active duty for training as a Reservist or National Guardsman (not to exceed in any fiscal year October 1 of one year to October 1 of the year following either 14 consecutive calendar days or two (2) regularly scheduled workweeks, if such training is not performed on consecutive calendar days), or
- b. temporary emergency duty as a National Guardsman

will be reimbursed for each day of such absence on which he (i) possesses one or more years of seniority in such bargaining unit and (ii) would otherwise have been scheduled to work on a regularly scheduled day of work up to a maximum of thirty days during any one fiscal year, in accordance with the succeeding provisions of this Section 15.4.

If he is absent for his entire shift because of such duty, he will be paid the difference between his gross military pay (including longevity pay and extra risk bonuses but excluding quarters,

subsistence, travel or similar allowances) and his daily straight-time pay for his regular shift. If he performs such duty and works on the same day, he will be paid the difference, if any, between his actual earnings for that day plus the military pay received and his daily straight-time pay.

Reimbursement to an employee under this Section 15.4 shall be payable only if the employee gives the Company prior notice of his call to such duty, and submits to his Supervisor a "Military Pay Statement" form furnished by the Company, fully completed by the employee and his Commanding Officer (or other commissioned officer authorized to approve military pay vouchers) and when released or excused from such duty returns to work promptly.

#### (15.5) Bereavement

When death of an employee's brother, brother of a current spouse, sister, sister of a current spouse, spouse, parent, parent of a current spouse (including stepparent and adoptive parent), child, adopted child, stepchild, grandchild, stepfather, stepmother, adoptive father, adoptive mother, occurs, the employee (except employees specified in Letter of Agreement No. 27), on request, will be excused for up to three (3) consecutive normal scheduled days of work (or for such fewer hours as the employee may be absent) during the five days (excluding (a) Saturdays and Sundays or in the case of seven-day operations, the sixth and seventh days of the employee's scheduled work week, and (b) holidays specified in Section 7.8) beginning with the date of death. The employee shall receive pay for each scheduled hour of work for which he is so excused (excluding (a) Saturdays and Sundays or in the case of seven-day operations, the sixth and seventh days of the employee's scheduled work week and (b) holidays specified in Section 7.8). Payment shall be made at the employee's straight-time hourly rate on the last day worked.

When death of an employee's grandparent, half brother, half sister, stepbrother, stepsister, or current spouse of a child (including adopted child and stepchild) occurs, an employee (except employees specified in Letter of Agreement No. 27), on request, will be excused for any normal scheduled day of work (or for such fewer hours as the employee may be absent) during the five days (excluding (a) Saturdays and Sundays or in the case of seven-day operations, the sixth and seventh days of the employee's scheduled work week, and (b) holidays specified in Section 7.8) beginning with the dates of death. Payment shall be made at the employee's straight-time hourly rate on the last day worked.

In the event a member of the employee's immediate family as above defined dies, the employee may, should the funeral or memorial service be delayed, have his excused absence from work as above provided delayed to include the date of the funeral or memorial service.

#### (15.6) Parental Leave

Paid Parental Leave will be provided to employees upon the birth of an employee's child, the legal adoption of a child under the age of eighteen (18) or upon the placement of a child for foster care.

Eligible employees will receive eighty (80) hours of Paid Parental Leave for the regularly scheduled hours at the employee's then current regular straight-time rate. Paid Parental Leave is to be initiated and concluded within twelve (12) months of the birth, adoption or placement of a child for foster care. Employees are eligible for one (1) Paid Parental Leave per twelve (12) month period. This leave must be used in full day increments. Employees are required to complete the Application for Paid Parental Leave and submit it in to the local FML coordinator within thirty (30) days of the

birth, adoption or placement of a child for foster care. Employees should schedule their Paid Parental Leave in advance at a time that is mutually agreeable between the employee and their supervisor, unless the leave is used during the forty (40) hour period immediately following the birth, adoption or placement of a child for foster care.

Paid Parental Leave will be administered concurrently with an employee's Family and Medical Leave (FML) entitlement, if eligible.

#### (15.7) General

- A. For purposes of this Article 15:
  - a. "Basic Hourly Rate of Pay" shall have the same meaning as in Section 18.1 of this Agreement.
  - b. In determining an employee's seniority and/or group seniority, the same period of service shall not be counted twice.
  - c. Except as expressly provided elsewhere in this Agreement, the effective date of this Agreement is March 13, 2023.
- B. Amounts payable to, or for the account of, an employee pursuant to the provisions of Sections 15.1, 15.2, 15.3, 15.4, 15.5 and 15.6 shall be included and taken into account in determining the amount of an employee's regular wages for the purpose of computing vacation or separation payments under Article 9 of this Agreement.
- C. All amounts payable under this Agreement by the Company to an employee for a pay period shall be combined into a single paycheck, subject to any required withholdings and employee authorized deductions.
- D. In applying the provisions of this Article 15 payments or contributions made by the Company pursuant to any Pension Plan or Group Insurance Plan be disregarded and shall not be taken into account for any purpose except to the extent otherwise expressly provided in this Article 15.

## ARTICLE 16 General

(16.1) It is the intention of the Company that nonbargaining unit employees of the Company devote themselves primarily to their own work and not perform the work normally assigned to bargaining unit employees except as such work may be performed incidental to their responsibilities and to such extent that the work being performed by a nonbargaining unit employee does not result in the displacement or elimination of the ongoing need for an employee covered by this Agreement.

Notwithstanding the above, the Company and Union have agreed that there are mutual advantages to having employees covered by this Agreement be permitted to be assigned to certain functions which they traditionally have not performed or have not exclusively performed.

Examples of these functions are:

- a. programming and maintenance of computers and robots;
- b. identifying training needs, developing training materials and conducting training;
- c. leading or participating in project teams where subjects studied may cover a wide range of functional areas;
- d. special assignments given to employees to accommodate temporary medical restrictions;
- e. evaluating, testing and/or analyzing experimental, prove design and/or production parts, assemblies, prime products, etc. in order to develop or prove engineering theory and design; and
- f. special assignments and nontraditional work given to employees assigned to the CJSP Pool in order to accommodate provisions of Letter of Agreement No. 23.

However, these functions are not exclusively and will not exclusively be the work of employees covered by this Agreement.

(16.2) The Company and the Union shall not, in discharging their respective responsibilities under this contract, be discriminatory of any employees because of sex, color, race, ancestry, religion, national origin, age, disability or handicap, protected veteran status, marital status, citizenship status, sexual orientation, gender identity, genetic information, any other characteristic protected by law, or membership in any labor or other lawful organization. Nor will the Company or Union discriminate against qualified individuals with disabilities as described in the Americans with Disabilities Act of 1990 as amended.

Further, the Company and the Union will comply with the applicable provisions of the Age Discrimination in Employment Act of 1967, the Rehabilitation Act of 1973 as amended, Americans with Disabilities Act of 1990 as amended, and the Vietnam Era Veteran's Readjustment Assistance Act of 1974 as amended.

- (16.3) Should any portion of this contract become invalid or unenforceable by reason of any applicable Federal or State Law or Presidential Executive Order, the remainder of the contract shall be unaffected thereby. When any such law or Presidential Executive Order has been rescinded, the provisions of the original agreement shall prevail.
- (16.4) The Company shall provide bulletin boards at locations mutually agreed to by plant management and its Local Union, to be used exclusively for Union notices. The notices will be furnished by the Union and posted by the Company in accordance with the provisions of applicable Local Agreements. The Union notices shall be restricted to the following:
  - 1. Notices of Union recreational and social affairs
  - 2. Notices of Union elections, appointments and results of Union elections
  - 3. Notices of Union meetings
  - 4. Reports of committees, approved by the Company

(16.5) The Union, and the employees it represents, agrees that it will condemn and will not authorize, encourage, or promote any curtailment or restriction of production, sit-down, slowdown, or other form of strike or work stoppage on the part of any employee, or group of employees, in the units covered hereby; that it will in good faith attempt to prohibit same.

There shall be no lockout by the Company during the term of this Agreement.

In the event a dispute arises under Section 16.2 of the Decatur, Section 6.7 of the Peoria, Section 6.6 of the Pontiac, Local Agreements, or Section 18.3 hereof, this Section 16.5 shall be inoperative only at the business unit where the dispute occurs, insofar as it refers to "strikes."

(16.6) If an eligible employee is transferred on or after the date of this Agreement, at the request of the Company from a job in the bargaining unit to an hourly rated job in another plant of the Company he shall receive a moving allowance in accordance with the following schedule:

Miles from	Moving
Employee's Plant	Allowance
to Other Plant	Amount
0-49	\$0
50-299	\$3,000
300 or over	\$4,000

To be eligible for such a moving allowance under this Section, an employee so transferred must:

- 1. establish that he has, in fact, changed his permanent residence as a result of the transfer, and
- 2. make written application to the Company, in a manner to be prescribed within each business unit, for such moving allowance within six months after the date of such transfer, and
- 3. provide documentation for bona fide expenses related to the movement of household goods, and family, if applicable, from one primary location to another as a result of such transfer, and
- 4. in the event the employee is terminated within three years of the date of transfer, the employee will be required to repay the moving allowance.
- (16.7) Employees shall notify the Company, in a manner to be prescribed within each business unit, of any change of address. Every quarter during the term of this Agreement, the Company shall give to the Financial Secretary of each Local Union the names of all employees (active and on layoff with recall rights) covered by the applicable agreement together with their addresses as they then appear on the records of the Human Resources Department. The Financial Secretary of each Local Union shall receive and retain such information in confidence and shall disclose it only to those officials of the Local Union or the International Union whose duties require them to have such information. Such officials shall likewise hold such information in confidence and shall disclose it to no one else.
- (16.8) Whenever the Company requires an employee (1) to undergo a periodic physical or mental examination because of the nature of his present job duties or (2) to undergo a physical or mental examination before being assigned to a different job, such employee shall be paid for hours spent in taking such physical or mental examination.

(16.9) An employee who becomes disabled as a result of an injury arising out of and in the course of his employment shall not suffer a loss of earnings for the remainder of his regular shift on the day on which he was injured.

(16.10)

#### A. Subcontracting, Outsourcing, Resourcing, Discontinuance of Product Line:

The Company shall continue to make the decisions as to whether work shall be performed by Company forces in any Company plant, or by others, consistent with an intention to maintain, so far as practicable, a stable work force. The Company shall make decisions of such nature with such intention taking into consideration such factors as the scope of the project or production requirement, relative costs, possession and availability of Company equipment and of employees qualified to accomplish the production without undue overtime or delay either of the specific production or of any other scheduled activity, desirability of continuity of relations with historic sources of supply and believed best utilization of all the Company's plants with a view to long-term stability and health of the enterprise as a whole.

#### B. Notice

In the event the Company contemplates a decision to subcontract, outsource or resource work or to discontinue a complete product line, as hereinafter defined, where such decision would, during the term of this Agreement, affect adversely the stability of the work force, resulting in a Complete or Partial Plant Closing as hereinafter defined, the Company will give reasonable notice of such proposed decision to the Union which, in the normal case, shall be no less than 180 days in the case of a Complete Plant Closing or 60 days in the case of a Partial Plant Closing prior to the Company making its final decision, provided, however, in the event such 180 or 60 days notice would impair the Company's need for speed, flexibility and confidentiality, the Company will give such notice as soon as practicable.

Thereafter, the Company, upon request, will meet with Union representatives, explain the reason(s) for such contemplated decision and provide relevant information requested by the Union. It will consider and respond to alternative proposals, if any, which may be suggested by the Union.

Liability arising from this Section16.10 (B) shall be limited to providing the benefits set forth in subsections (C) and (D) of this Section 16.10.

#### C. Implementation of Complete Plant Closing Decisions

In the event that the Company, following such prior notice, meetings and conferences, pursuant to this Article makes a decision in the exercise of its business judgment to subcontract, outsource, resource or discontinue productive operations where such decision will result in a Complete Plant Closing, each affected employee to be permanently laid off as a result of such decision will be given or mailed an "Option" form, on which he will be given a one-time-only opportunity to select one or both of the options, if any, for which he is eligible.

The options are:

- (i) Being placed on a Master Recall List as specifically described in Section 16.10 (D) of this Article. If an employee has not been placed on permanent layoff at the time the option form is distributed, he may delay deciding on the Master Recall option until he is placed on permanent layoff.
- (ii) Receiving Outplacement Services as specifically described in Section 16.10(E) of this Article.

Additionally, an eligible employee, or an employee who becomes eligible at a later date, will be able to elect Special Early Retirement as specifically described in the Non-Contributory Pension Plan.

Each such employee shall return the Option form to the Company within 60 calendar days following the date he received such form.

During such 60-day period, each such employee will be given an opportunity for counseling on the benefits and/or effects of each such option for which he is eligible. An employee who fails to return such form to the Company as above provided will be deemed to have waived all rights and/or benefits provided for in this Section 16.10 under either option unless satisfactory reason is given for such failure.

The Company will establish and maintain a "Plant Closing Services" office for a period of time of not less than six months following the date the plant(s) and/or facility(ies) is closed as defined in Section 16.10(F) of this Agreement. Such office will coordinate and/or administer the provisions of Subsections (C), (D) and (E) of this Section 16.10.

#### D. Master Recall List

- (i) A Master Recall List shall be established for each of the business units set forth in Section 4.7. It shall consist of all "eligible employees," hereinafter defined, who have designated that business unit as one in which they will accept employment. Eligible employees will be placed on such List in order of seniority and/or group seniority (if applicable), whichever is greater, and the eligible employee with the greatest such seniority and/or group seniority (if applicable) shall be given the first opportunity for employment in that business unit in accordance with the following provisions of this Section 16.10 (D); provided, however, that in the event of a job opening at Labor Grade 3 or higher, the opportunity for employment will be offered to the eligible employee, if such there be, with the greatest seniority and/or group seniority (if applicable) who can perform such job without being trained.
- (ii) An employee who has exercised the Master Recall option, and
  - (a) who possesses (or acquires prior to being placed on permanent layoff) five or more years of seniority and/or group seniority (if applicable) (including employees on layoff or leave of absence under the first paragraph of Section 14.1 or Sections 14.4, 14.6, 14.7, 14.8, 14.9, or 14.11 of this Agreement) on the date of notice referred to in Section 16.10 (B), and
  - (b) who has designated, as hereinafter provided, one or more of the business units set forth in Section 4.7 in which he will accept employment,

shall be an "eligible employee" for employment opportunities in any such designated business unit(s) in accordance with the provisions of this Section 16.10 (D) but ceases to be an "eligible employee" in the event and at the time of: separation; death; retirement; refusal to accept an offer of employment under this Section 16.10(D); acceptance of an offer of employment in any business unit covered by this Agreement, whether or not the offer was made under this Section 16.10(D); failure to promptly notify the "Plant Closing Services" office established by the Company or, after the date it is closed, Corporate Labor Relations, 100 N.E. Adams Street, Peoria, Illinois 61629 of any change of address or that the-reason for a leave of absence no longer exists; the expiration of a period of five years from the date of permanent layoff; or a break in seniority and/or group seniority (if applicable) in accordance with any other provision of this Agreement, the appropriate Local Agreement or any other Plan or Agreement between the parties.

- (iii) Within ten calendar days of the date the Option form is returned to and received by the Company, each eligible employee who has selected the "Master Recall List" option will be mailed a form on which such employee will be given a one-time-only opportunity to designate the business unit(s) covered by this Agreement in which he will accept an offer of employment. Each such employee shall return such form to the Company within 15 calendar days following the date he received such form. Employees who defer their Master Recall option until being permanently laid off shall return such form within 15 calendar days following their permanent layoff date. If the employee has designated one or more business units as ones in which he will accept employment, the provisions of this Section 16.10 (D) will become operative for such employee on the first or third Monday of the month following receipt of such form by the Company provided such form is received 10 calendar days prior to such Monday.
- (iv) A new employee will not be hired under the provisions of Sections 12.13 or 12.15 of this Agreement into a job in a business unit covered by this Agreement if there is an eligible employee (excluding those who are still on a leave of absence under the first paragraph of Section 14.1 or Sections 14.6, 14.7, 14.8 or 14.11 of this Agreement only, but no others) on the Master Recall List for that business unit who possesses the necessary qualifications for employment on such job and provided that an applicant for employment on such job does not possess measurably better qualifications for that job than such eligible employee.
- (v) An eligible employee will not be offered employment on a job in a business unit if there is an employee who is downgraded or laid off from such job in that business unit and who still possesses recall rights to such job.
- (vi) An eligible employee who is offered an opportunity for employment in a business unit as above provided must accept or reject such offer by 4:00 p.m. of the third regularly scheduled workday following the day the employee receives such offer. The employee will be advised as to the plant or facility involved and the job classification, rate of pay and shift of the job being offered. An eligible employee who accepts an offer of employment under this Section 16.10 (D) shall report for work within ten calendar days (unless his release from his current business unit is delayed as provided below) after being offered such job, unless satisfactory reason is given, or the provisions of Section 11.2 (3) will be applied to him as though the notice requirements of such Section had been fully met.

At its option the Company may require an employee who has accepted a Master Recall offer to remain at his current business unit until his services are no longer needed. In such event, the employee's seniority date at the business unit where he has accepted the Master Recall offer shall be on the day following his acceptance of such offer. Subject to the provisions of Article 13 of this Agreement, the Company will hold the job offered until the employee is released by his current business unit provided that the Company may temporarily assign other employees to fill such job.

- (vii) An eligible employee who accepts an offer of employment in a business unit under the provisions of this Section 16.10 (D)
  - (a) shall be deemed to be eligible for the appropriate moving allowance, if any, as set forth in Section 16.6 of this Agreement;
  - (b) shall not be required to serve a probationary period under Section 11.1;
  - (c) shall have "his last date of hire" under the provisions of Section 9.8 computed as though he were still working in the business unit from which he was or would be laid off as a result of the plant or facility being closed;
  - (d) shall be deemed to have met the employment and/or seniority eligibility requirements for Sections 15.1,15.2, 15.3, and 15.4;
  - (e) shall be required, at the Company's option, to take a medical examination prior to the date he begins work in such business unit to determine his fitness to be employed in such business unit and perform the work being offered;
  - (f) shall, on the first day he begins work in the business unit in which he accepted employment under this Section 16.10 (D), be deemed to have broken his seniority and/or group seniority (if applicable) in the business unit in which he last worked prior to becoming an "eligible employee"; and
  - (g) except as provided in the last paragraph of (vi) above, and subject to paragraph (e) above, shall have seniority and/or group seniority (if applicable) in the business unit in which he accepts employment measured from the day following his acceptance of such offer.

Such employee, who accepts employment in a business unit under this provision and is subsequently laid off from such bargaining unit under the provisions of Section 13.5 before he accumulates more than two (2) years seniority in such business unit would, concurrent with the time he would break seniority within such business unit under the provisions of Section 11.2, be returned to the Master Recall list for those business units he originally designated and be considered an "eligible employee" subject to the provisions of Section 16.10 (D) (ii).

(viii) In the event an eligible employee with less seniority and/or group seniority (if applicable) than another eligible employee on a Master Recall List accepts an offer of employment that was made in error or an employee was hired in error, the Company will not incur any financial liability to such more senior eligible employee provided the next job

- opening for which he is qualified (in any business unit in which he has indicated he will accept employment) occurring after the Company becomes aware of such error is offered to such more senior employee.
- (ix) The Company will provide to the UAW Agricultural Implement Department, on a quarterly basis, a list of "eligible employees" who are on the Master Recall List for each business unit set forth in Section 4.7. Each Local Union will receive a copy of the list applicable to their business unit. Such list shall include each employee's name, badge number, business unit from which the employee is laid off, date of layoff and seniority and/or group seniority (if applicable) date.

#### E. Outplacement Services

- (i) An employee placed on permanent layoff as a result of such Company decision who possesses one or more years of seniority and/or group seniority (if applicable) (including employees on layoff or leave of absence under the first paragraph of Section 14.1 or Sections 14.4, 14.6, 14.7, 14.8, 14.9 or 14.11 of this Agreement) on the date of notice referred to in Section 16.10 (B), shall be an "eligible employee" for purposes of this Section 16.10 (E) but ceases to be an "eligible employee" in the event and at the time of: separation; death; retirement; acceptance of an offer of employment with any employer, unless otherwise mutually agreed; the expiration of the lesser of (i) a period of two years from the date of permanent layoff, or (ii) the period of time during which he accumulates seniority and/or group seniority (if applicable) in accordance with Sections 11.2 (4) and 11.4 or the comparable provisions covering group seniority; or a break in seniority and/or group seniority (if applicable) in accordance with any other provision of this Agreement, the appropriate Local Agreement or any other Plan or Agreement between the parties.
- (ii) Each "eligible employee" shall be given an opportunity to avail himself of any or all of the following benefits and services:
  - (a) Counseling by the Company about retirement, insurance and related benefits plan entitlements.
  - (b) Vocational counseling by the Company or an organization selected by the Company which provides such services.
  - (c) Training and assistance in the preparation of a resume suitable for distribution to prospective employers; such assistance shall include a listing of the jobs and/or work experience that the employee has had with the Company.
  - (d) Outplacement assistance, in which the Company will contact the appropriate public employment service, private employment agencies, and area employers in search of employment opportunities for eligible employees; such assistance may include:
    - (i) the distribution of a list of employees showing their job and/or work experiences and the date their services will be available to interested agencies, area employers, or organizations with a request that the employees be given employment consideration:
    - (ii) arrangements for recruiting and interviews by area employers, including formal

job fairs (as warranted by external placement opportunities and interest among area employers), consistent with plant operation requirements and efficiency considerations; or

- (iii) solicitation of lists of job opportunities from area employers with posting or distribution of copies of these lists to interested employees.
- (e) Severance payment of \$2,000 to each eligible bargaining unit member as follows. A bargaining unit member is eligible to receive the above lump sum payment if all the following conditions are met:
  - (i) The employee was actively at work or on an approved leave of absence on the date of notice referred to in Section 16.10(B);
  - (ii) The employee has signed, returned, and not revoked a Group Separation and General Release Agreement;
  - (iii) The employee remains actively working for the Company or on an approved leave of absence until the Company terminates his employment.

## F. Implementation of Partial Plant Closing Decisions

In the event that the Company following such notice, meetings, and conferences pursuant to this Article makes such a decision in the exercise of its business judgment to subcontract, outsource, resource or discontinue productive operations where such decision will result in a Partial Plant Closing, employees to be laid off for indefinite periods as a result thereof shall be granted or denied benefits, if any, in accordance with and subject to the limitations of the Letter of Agreement No. 23.

#### G. Definitions

- (i) A "decision to subcontract, outsource, resource work or discontinue a complete product line that will affect adversely the stability of the work force" means such a decision by the Company that will result in a "Complete Plant Closing" or a "Partial Plant Closing" during the term of this Agreement as hereinafter defined.
- (ii) A "Complete Plant Closing" means a Company decision to subcontract, outsource, resource or discontinue all productive operations within the plant(s) and/or facility(ies) where employees comprise one of the bargaining units set forth in Section 2.1 with the result that no employees in such bargaining unit (except those who may be retained for a period of time to maintain the buildings and equipment) are working in that bargaining unit and the Company has no expectations of productive operations being resumed in such bargaining unit.
- (iii) A "Partial Plant Closing" means a Company decision to subcontract, outsource, resource or discontinue some productive operation(s) within a plant(s) and/or facility(ies), where employees comprise one of the bargaining units set forth in Section 2.1, that will result in the elimination of 50 jobs or 5% of the jobs in such bargaining unit, whichever is less.
- (iv) "Subcontracting" means a Company decision to have other than Company forces perform

production-related operations that have been normally and customarily performed in the Tool Room, Plant Engineering or Technical Center by employees in one of the bargaining units set forth in Section 2.1 during the term of this Agreement.

- (v) "Outsourcing" means a Company decision to have other than Company forces perform productive operations other than production-related operations defined in "Subcontracting" that have been normally and customarily performed by employees in one of the bargaining units set forth in Section 2.1 during the term of this Agreement. Outsourcing does not include decisions by the Company concerning the placement of productive operations related to the manufacture of new products, components or parts, not heretofore produced by the Company and which are not replacements for existing Company products, components or parts.
- (vi) "Resourcing" means a Company decision to have productive operations that have been normally and customarily performed by employees in one of the bargaining units set forth in Section 2.1 performed by Company forces other than employees in such bargaining unit during the term of this Agreement. Resourcing does not include decisions by the Company concerning the placement of productive operations related to the manufacture of new products, components or parts not heretofore produced by the Company and which are not replacements for existing Company products, components or parts.
- (vii) "Discontinuing a complete product line" means a Company decision to discontinue the manufacture, distribution and sale of a type of product (crawler tractors, wheel tractors, motor graders, hydraulic hose, etc.) or to discontinue the manufacture of a model of any such type of product, other than hydraulic hose, that is not replaced by another model.

#### (H) Logistics Services Work

Notwithstanding the foregoing provisions of this Section 16.10, the Company shall have the sole right, in the exercise of its managerial discretion, to remove logistics services work performed for third party customers from any facility in which it is placed. Nothing herein shall be construed as restricting the Company's right to outsource, resource, relocate or discontinue such work.

## (I) Returning Work To Bargaining Unit

In order to allow the Company to have the flexibility it needs to operate efficiently and effectively and to minimize the impact of employment changes on employees, the Company, at its sole discretion, may elect to temporarily return work to a bargaining unit that has been previously outsourced, subcontracted or resourced in accordance with this Section 16.10.

If the Company elects to temporarily return such work to a bargaining unit, the Union acknowledges and agrees that such work may be removed again without becoming subject to the provisions of Section 16.10 (A) or Section 16.10 (B) of this Agreement. Nothing in this Section 16.10 (I) shall, however, be construed as a waiver by the Union of otherwise applicable rights provided to employees in the remaining provisions of this Section 16.10. The Union also agrees that, if the Company temporarily returns such work to a bargaining unit and later removes that work again, it will not file or process grievances to protest such decisions.

The appropriate business unit representative will provide the appropriate local union representative with specific information concerning the job classification(s) and number of jobs involved when work is to be temporarily returned to a bargaining unit or when such work is to be removed again. Business unit representatives and local union representatives will consult and agree on a case-by-case basis that the provisions of this Section 16.10 (I) are applicable prior to such work being temporarily returned to a business unit.

(16.11) The parties jointly support the objective of voluntary improvement of the skills and capabilities of employees. The Company will continue to provide (within the capabilities of particular plants) off-shift classes on work-related subjects. These classes shall be available to all employees and a notation of satisfactory completion made a part of the employee's record. At the request of any Local Union a committee shall be created for its appropriate bargaining unit to consist of not more than three Union and three Company representatives to consider the operation of such classes and make suggestions, if such be deemed advisable, as to their further operations.

## ARTICLE 17 Training

(17.1) The provisions of this Article 17 shall apply only to the business units listed in paragraph (c) below.

When used in this Article 17:

a. "Apprentice" means an employee in a bargaining unit employed in any of the following classifications (provided such classification is included in the Exhibit A of the applicable Local Agreement):

Electrician Apprentice
Toolmaker Apprentice
Machine Repair Apprentice
Maintenance Mechanic Apprentice
(East Peoria (Transmissions, Undercarriage, and Prime Product); Decatur; Pontiac; Morton only)

who is engaged in learning and working in the trade in which he has enrolled.

- b. "Designated Craft" means a job classification as defined in Section 17.12.
- c. "Business Unit" means the following facilities:

Local 751 Decatur

Local 974 East Peoria – Transmissions (Bldgs KK, HH (West) & Bldg H Electric Drive)

East Peoria – Undercarriage (Bldgs BB, EE, GG HH (East)) & Maintenance Components

East Peoria – Prime Product (Bldgs LL, NN & SS)

Mapleton – CMO

Mapleton – Seals

Morton

#### Technical Center and Peoria Proving Ground

Local 2096 Pontiac

#### (17.2) Business Unit Joint Apprenticeship Committee

a. At each business unit where this Article 17 is applicable, there shall be established a Business Unit Joint Apprenticeship Committee (herein referred to as Business Unit Committee) composed of four members, two appointed by the Company, one of whom shall be a member of that business unit's Training organization, and two appointed by the Union. Either the Company or the Union may at any time remove a member appointed by it and may appoint a member to fill any vacancy among its members. The Company and the Union shall each notify the other in writing of any appointment to a Business Unit Committee and no such appointment shall become effective prior to the giving of such notice.

The Union may also identify, in writing, an alternate member, who, in the event a regular member was unavailable, would attend either regular or special meetings.

Each member of a Business Unit Committee appointed by the Union, including the alternate member, shall be:

- (i) an employee of the Company; and
- (ii) employed in and holding UAW Journeyman status in a "Designated Craft" classification for one of the Apprentice Courses at that business unit.

In addition, at least one member of each Business Unit Committee appointed by the Union shall be a graduate of one of the Apprentice Courses at that business unit, if there are such graduates.

The Business Unit Committee shall meet at an agreed upon time once each quarter unless the time between meetings is extended by mutual agreement. Members of the Business Unit Committee appointed by the Union will not lose pay for regularly scheduled hours because of attendance at a regularly scheduled meeting of that Business Unit Committee. Through mutual agreement, special meetings of a Business Unit Committee will be held at the request of either the Union or the Company. The requesting party shall provide the other party with a written agenda at least three working days prior to the date of the special meeting. Special meetings requested by the Union will not be held on Company time. Meetings called by the Company will not result in loss of pay for regularly scheduled hours for members of the Business Unit Committee appointed by the Union.

The Company will prepare minutes of each meeting. Such minutes shall include: the date of the meeting; names of the individuals present; a brief statement of items discussed; and the consensus or disposition, if any, reached on those items. A copy of those minutes will be provided to each member of the Business Unit Committee within ten working days of the meeting.

b. Each Business Unit Committee shall:

- 1. Have the right to designate a Business Unit Committee member to counsel individuals about to enter an Apprentice Course to ensure that they are informed and impressed with the responsibilities they are about to accept as well as the benefits they may receive.
- 2. Review applicants for an Apprentice Course who have met the requirement set forth in paragraphs 17.5 (a) and 17.5 (b). Following such review, an applicant may be rejected if a majority of the Business Unit Committee agree that rejection is appropriate.
- 3. Discuss and review cases in which any Apprentice at that business unit is failing to satisfactorily progress. One Union member of the Business Unit Committee, together with one Company member of the Business Unit Committee, at a mutually agreeable time, shall, without loss of pay for regularly scheduled hours, be permitted to counsel with such Apprentice. After such consultation, the responsibility for making the decision regarding the continuation in the Apprentice program for those demonstrating an inability to learn or a lack of interest in work or education rests with the Company. In the event a majority of the Business Unit Committee does not agree, the situation may be referred to the Central Joint Apprenticeship Committee. (Nothing in this Article 17 shall limit the Company's right to discipline/discharge Apprentices for matters not related to their training as Apprentices. Such action by the Company shall not be referred to the Business Unit Committee, but may be subject to the grievance procedure.)
- 4. Review the Apprentice Course(s) established for that business unit and make recommendations for the improvement of such Course(s).
- 5. Receive a list of employees who are Apprentices covered by this Article 17 at that business unit.
- 6. Report and/or refer matters of discussion to the Central Joint Apprenticeship Committee as well as accept findings from the Central Joint Apprenticeship Committee.

#### (17.3) Central Joint Apprenticeship Committee

There shall be established a Central Joint Apprenticeship Committee composed of six members, three appointed by Caterpillar Inc., one of whom shall be a member of the Company's corporate Training organization, and three representatives appointed by the Union. Two of the Union representatives shall be from the International Union and shall hold UAW Journeyman status in one of the crafts referenced in Section 17.1(a). The third Union representative shall be a member of one of the Business Unit Joint Apprenticeship Committees.

Either the Company or Union may at anytime remove a member appointed by it and may appoint a member to fill any vacancy among members appointed by it. The Company and Union shall notify each other in writing of any appointment to the Committee and no such appointment shall become effective prior to giving such notice.

#### The Central Joint Apprenticeship Committee shall:

1. Meet at an agreed upon time each six months unless the time between meetings is extended by mutual agreement. Special meetings of the Committee will be held at the request of either the Union or the Company at a mutually convenient time. Special meetings may be conducted by conference phone calls. The requesting party shall provide the other party with a written

agenda at least 15 working days prior to the date of such special meeting.

The Company will prepare minutes of each meeting. Such minutes shall include: the date of the meeting; names of the individuals present; a brief statement of items discussed; and the consensus or disposition, if any, reached on those items. A copy of these minutes will be provided to each member of the Central Joint Training Committee within ten working days of the meeting.

- 2. Review reports and matters of discussion referred by Business Unit Committees.
- 3. Advise appropriate Business Unit Committees of the Central Joint Apprenticeship Committee's findings on matters referred from respective Business Unit Committees.

Items, which remain in dispute following discussion by the Central Joint Apprenticeship Committee, may, if the Union so elects, be referred directly to the Final Step of the grievance procedure in the appropriate Local Union and, if so referred, will thereafter be subject to the normal grievance and arbitration procedure.

(17.4) Application to become an Apprentice in a business unit will be received and dated by the Training organization of the Company at that business unit. Such Training organization shall determine which of the applicants meet the requirements of paragraphs 17.5(a) and 17.5(b) and will furnish the Business Unit Committee established for that business unit a list of those so approved. Opportunities for training afforded under this Article 17 shall be available to all otherwise eligible and qualified candidates without regard to sex, color, race, ancestry, religion, national origin, age, disability or handicap, protected veteran status, marital status, citizenship status, sexual orientation, gender identity, genetic information, any other characteristic protected by law.

#### (17.5) Eligibility Requirements

A person selected to become an Apprentice shall possess the following eligibility requirements:

- a. Must have attained the age of 18.
- b. Indication of potential for successfully completing an Apprentice Course by reason of an evaluation of biographical data, school records, previous work experience, Company work history (if any), and successful completion of an evaluation system to be administered and interpreted by the Company. This evaluation system will also determine the successful individual's initial level of competency for purposes of determining (1) placement on the waiting list for an Apprenticeship and (2) placement on a rate step upon entry into an Apprenticeship.
- c. Review by the appropriate Business Unit Committee.

## (17.6) Entry Into An Apprenticeship

Each business unit shall establish a waiting list for entry into each Apprentice Course at that business unit. This list shall be composed of all applicants for such Apprentice Course and relative position on this list shall be determined by the level of competency demonstrated by each individual when completing the evaluation system, which is administered during the application process and subsequent demonstrations of competency which are initiated by individual applicants. The

individual who has demonstrated the highest level of competency shall be at the head of the list; provided, however, that no individual shall be placed on the list until 30 calendar days following the date such individual was determined to have met the eligibility requirements listed in Section 17.5 and that, after being placed on the list, an individual's position on the list will not be changed until 30 days following the date such individual demonstrates additional competency. If individuals have demonstrated the same level of competency, a current employee will be considered to be higher on the list than an equally competent non-employee, and the employee with the greatest seniority and/or group seniority (if applicable) will be considered to be higher on the list among equally competent current employees. An individual cannot be on more than one waiting list at each business unit at the same time. By mutual agreement, the provisions of this Section 17.6 may be waived for Affirmative Action purposes.

## (17.7) Terms of Apprenticeship

The terms of Apprenticeship for each individual shall be in accordance with the training work schedule and classroom schedule as determined on the basis of such individual's skill development map by the Company at each business unit. In each business unit, the Business Unit Committee shall be provided an outline of these schedules. These schedules provide a guide and may from time to time be revised, as conditions require. A copy of all such revisions shall be furnished to the Business Unit Committee involved.

#### (17.8) Apprentice Reductions and Returns to Apprentice Courses

- a. In the event it is necessary to reduce the number of Apprentices in a Course within a Business Unit, Apprentices in that Course shall be removed from that Course on the basis of their demonstrated level of competency sixty (60) days prior to the effective date of the reduction in force, beginning with the Apprentice with the lowest level of competency. If Apprentices have the same level of competency, the Apprentice with the least seniority shall be removed first. An Apprentice removed from an Apprentice classification as designated in Section 17.1 (a), in accordance with this Section 17.8 (a) shall be placed in accordance with the provisions of Article 13 as specified in Section 17.12 in the same manner as though his departure from that Apprentice classification had resulted because of those provisions.
- b. The provisions of Article 13 as specified in Section 17.12 shall not apply to Apprentices unless a reduction in force of Apprentices occurs or the application of the foregoing provisions of this Section 17.8 results in a reduction in force of Apprentices.
- c. Apprentices removed from an Apprentice classification solely as the result of a reduction in force, shall be returned to that Apprentice classification, when conditions deem appropriate, on the basis of their demonstrated level of competency sixty (60) days prior to the effective date of the return to that Apprentice classification, beginning with the Apprentice with the highest level of competency. If Apprentices have the same level of competency, the Apprentice with the greatest seniority shall be returned first.

#### (17.9) Wages

Apprentices shall be paid for required classroom attendance. Hours spent in required classroom attendance shall be considered hours of work in computing overtime. An Apprentice shall be paid the wage rate corresponding to the individual's demonstrated level of competency.

#### (17.10) Supervision of Apprentices

Apprentices shall be under the supervision of the Training organization of the Company and under the immediate direction of the Supervisor of the area to which they are assigned. The Training organization of the Company will determine the movement of Apprentices in accordance with training work schedules.

#### (17.11) Tools

The Company will provide the tools required for each Apprentice Course to the Apprentice, and such tools shall become the personal property of the Apprentice upon graduation. The Company will provide the Business Unit Committee with the list of tools provided for each Apprentice Course in the business unit for which the Business Unit Committee is established.

#### (17.12) Placement Upon Graduation

#### An employee

- (i) who graduates from one of the Apprentice Courses hereinafter set forth, will, provided there is a job opening, be placed in his "Designated Craft" within the business unit where he is assigned at the time he graduates, provided there is no employee within that business unit with greater seniority (a) who is currently working in that "Designated Craft" but on a different shift and whose Prefiled Shift Preference form indicates the shift of first preference is the same as the shift of the opening; or (b) who has previously worked in that "Designated Craft" or previously graduated from such Apprentice Course and whose job bid for return to such "Designated Craft" has not been honored.
- (ii) who graduates and who is not, in accordance with (i) above, placed in his "Designated Craft" shall be placed in accordance with the provisions of Article 13, or Article 10 whichever is applicable. The starting point of such reduction in force procedure shall be that "Designated Craft" to which such graduate Apprentice would have been placed in accordance with (i) above.
- (iii) who graduates or has graduated, and whose Local Agreement provides for group seniority, shall, for purposes of (i) and (ii), above, and for purposes of bidding to a "Designated Craft" under the provisions of Article 10 be deemed to have group seniority equal to the period of his apprenticeship.

#### Apprenticeship Course

From Which the "Designated Craft"

Employee Graduated: Classification:
Electrician Electrician (6)

Machine Repair Machine Repair Mechanic (6)

Toolmaker (6)

Maintenance Mechanic Maintenance Mechanic (6)

(17.13) Apprentice programs not listed in Section 17.1 and other training programs will not be subject to the provisions of this Article 17 unless mutually agreed by the parties.

(17.14) During 2023 negotiations, the parties discussed the importance of maintaining and developing a skilled workforce. As a result of these discussions, the parties agreed that upon the request of the Final Step committee, the Company and the Union would meet to share ideas on how to promote and develop training programs that may create additional job opportunities for employees in their respective locations.

## ARTICLE 18 Wages

(18.1)

- A. The basic hourly rates of pay during the term of this Agreement are set forth in Appendix A to this Agreement.
- B. The Company reserves the right to increase the rates of a Labor Grade or Labor Grades and/or adjust the number of rate steps on this rate schedule within any business unit or all of the business units.
  - The Company reserves the right to place an employee being hired or rehired on any rate step. The Company also reserves the right to place an employee entering an apprenticeship or training program on any rate between the minimum rate and the minimum graduating rate of the employee's program.
- C. Employees on this rate schedule shall be placed on the next rate step in the Labor Grade every 26 weeks until they reach the maximum rate for their job classification.
  - Each rate step change will be effective at the beginning of the first pay period following the completion of the required number of weeks of active work.
  - Notwithstanding the above, the Company reserves the right to move an employee to a higher rate step based on merit.
- D. As used in this Section, "basic hourly rates of pay" means straight time hourly rates excluding night shift premiums and all other premiums or bonuses of any kind.
- E. Effective the first full pay period following ratification, the basic hourly rates of pay in effect shall be increased by seven percent (7%).
- F. Effective the first full pay period in March of 2025, the basic hourly rates of pay in effect shall be increased by four percent (4%).
- G. Effective the first full pay period in March of 2027, the basic hourly rates of pay in effect shall be increased by four percent (4%).
- H. Effective the first full pay period in March of 2028, the basic hourly rates of pay in effect shall be increased by four percent (4%).
- (18.2) If an employee is temporarily assigned to a lower rated job, he shall continue to receive his regular rate of pay. This provision shall not apply if the assignment to the lower rated job is in lieu of layoff, is upon request of the employee, or because the employee is unable to satisfactorily perform the higher rated job, in which event the employee will receive the rate of pay for the lower

rated job.

- (18.3) During the life of this Agreement either party may request, in writing, a meeting of a Business Unit Grievance Committee (In Local 974, the Bargaining Unit Chairman may also participate in these meetings.) to enter negotiations for the following purposes:
  - 1. To establish a classification and schedule of rates for a new type of work not within the scope of existing classifications. (In such event the Company may establish a temporary classification and schedule of rates pending completion of negotiations.) When activities within such "new type of work" are actually assigned to the business unit, the Company will so notify the appropriate Local Union.
  - 2. To make adjustments, if warranted, when the work in an existing job classification is significantly altered subsequent to the effective date of this Agreement.

Negotiations, as provided above, shall commence within ten days following date of written request, except that in the event the new machine, tooling, or equipment involved in a "new type of work" referred to in (1) above is not performing as expected by the Company, such negotiations shall be deferred until such time as the new machine, tooling, or equipment is performing satisfactorily not to exceed six months unless otherwise mutually agreed. The temporary assignment of an employee to operate such machine, tooling, or equipment during such period shall not establish a precedent for the establishment of any new classification and schedule of rates. In the event such deferred period exceeds 90 calendar days and the negotiations commencing thereafter as provided above result in a new classification and schedule of rates to which an employee will be assigned, the opening will be filled in accordance with the provisions of Article 12. In the event an employee is moved, such employee will receive a retroactive payment for the rate increase such employee receives, if any, as a result of being so moved. The retroactive payment shall be an amount equal to the difference between the employee's hourly rate of pay immediately prior to such move and the rate of pay which the employee receives upon such move multiplied by the number of hours worked by that employee from the end of such ninety-day deferred period to the effective date of the employee's move.

However, if the "new type of work" referred to in (1) above is within the scope of a job classification and schedule of rates already existing in the Exhibit A of one or more of the other bargaining units covered by this Agreement, such negotiations shall only consist of the parties incorporating such existing classification, with needed name and/or product identification changes, and schedule of rates into their Exhibit A and the following provisions of this Section 18.3 shall not be applicable to such new type of work. The schedule of rates referred to in this paragraph shall be the rates in effect for employees in accordance with the Appendix A of this Agreement.

After negotiations, as above provided, have continued for at least five days without agreement on the classification in question, the provisions of Section 16.5 shall be inoperative insofar as, but only insofar as:

- (a) such provisions refer to "strikes"; and
- (b) such provisions refer to the bargaining unit where the dispute exists; and
- (c) such negotiations are concerned;

for sixty days from the date on which either party received from the other party a written request to enter negotiations as above provided; provided, however, in the event such negotiations are deferred as above provided, such sixty-day period shall date from the date the new machine, tooling or equipment is performing satisfactorily. Upon expiration of such sixty days, all provisions of Section 16.5 shall be automatically effective if no strike action has as yet been taken.

(18.4) The schedules of wage rates applicable to each business unit are incorporated as a part of this Agreement by reference to them as Appendix A.

The schedule of job classifications and basic hourly rates of pay and the related descriptions of the work performed in each job classification within each bargaining unit are incorporated as part of each Local Agreement by reference therein as Exhibit A. It is understood that the description of the work performed in each job classification briefly states the major significant characteristics of that classification which are the determining factors in establishing its wage rate level, and of necessity cannot describe all activities (whether regularly performed or not) involved in individual job assignments.

## ARTICLE 19 Duration

(19.1) Subject to the ratification of this Agreement by the membership of the Union on or before March 12, 2023:

For all Business Units except the Caterpillar Logistics Services Business Units and except as otherwise specifically provided elsewhere, this Agreement shall be effective on the Monday following the ratification of this Central Agreement and shall remain in effect until March 1, 2029, and thereafter from year to year, unless sixty days prior to the date of expiration either party gives notice to the other, in writing, that it desires to modify or terminate.

For Caterpillar Logistics Services Business Units and except as specifically provided elsewhere, this Agreement shall be effective on the Monday following the ratification of this Central Agreement and shall remain in effect until the sixtieth day following the effective date of the Central Agreement and all Local Agreements covered by the Central Agreement to succeed those expiring on March 1, 2029.

Agreed and subscribed to the date and year above written.

#### CATERPILLAR INC.

BY: Ed Thompson John Hubert Wes Knepp Brad Butler

Melissa Hathcock

INTERNATIONAL UNION, UNITED AUTOMOBILE, AEROSPACE AND AGRICULTURAL IMPLEMENT WORKERS OF AMERICA

BY: Chuck Browning Brandon Campbell

Brandon Keatts
Thomas Weber
Bobby Garland
JC Zimmerman
Craig Karnes
Lucas Despain
Bobby Garland
Joe Morel
Scott DeVrieze

## Josh Hogan

# LOCAL UNIONS OF INTERNATIONAL UNION, UNITED AUTOMOBILE, AEROSPACE AND AGRICULTURAL IMPLEMENT WORKERS OF AMERICA

LOCAL NO. 751

BY: Steve Collins Jason Hess

Sam Hart

LOCAL NO. 1872

BY: Carroll Lewis

LOCAL NO. 974

BY: Randy Diehl Bobby Koller

Matt Butler Matt Diebel

LOCAL NO. 2096

BY: Rob Drysdale John Bogucki

### Letter of Agreement No. 1 ARBITRATOR SELECTION

In order to implement the provisions of Section 6.2 of the Central Agreement, the parties agree to select Permanent and Panel Arbitrators in accordance with the following procedures:

- 1. That the same individual serve as Permanent Arbitrator or Panel Arbitrator at all plants covered by the Central Agreement.
- 2. That we forthwith mutually endeavor to select such an individual to serve under terms substantially conforming to the annexed draft of employment contract.
- 3. That if we are unable to make a selection within a reasonable period of time, we will jointly furnish the American Arbitration Association with the names of those who have been proposed but not agreed to by the parties or who may have declined appointment and request it, as rapidly as it can, to furnish a panel of five other individuals it feels qualified and able and who have indicated a willingness to serve upon substantially the terms of the annexed proposed employment contract, from which one will be selected by agreement if possible, otherwise by the alternate striking of names, with the party to do the first striking determined by lot.

Should it become necessary during the term of the labor contracts to select a successor Permanent Arbitrator or Panel Arbitrator, the same procedure will be utilized.

## Appendix to Letter of Agreement No. 1 AGREEMENT

Agreement made between Caterpillar Inc. (Company), and the International Union, United Automobile, Aerospace and Agricultural Implement Workers of America (Union) and (Permanent Arbitrator):

- 1. The Company and Union hereby appoint and employ as Permanent Arbitrator, to function under and in accordance with Labor Agreements entered into, between the Company and Local Unions 751, 1872, 974, and 2096 for a term, subject to the conditions of each Agreement, ending and hereby accepts such appointment and agrees to act as Permanent Arbitrator subject to the terms of this and those contracts.
- 2. The Union and the Company each agree to pay the Permanent Arbitrator one-half (1/2) of (a) a retainer of dollars (\$ ) per month for the term hereof, or until such time as the agreement is terminated in accordance with paragraphs 4 and 5 hereof, and (b) in addition thereto a per diem of dollars (\$ ) for each day required by him to be used as Permanent Arbitrator together with (c) such business expenses as are reasonably incurred by the Permanent Arbitrator in acting as such.
- 3. On the first of each month the Permanent Arbitrator is to bill the Company and the Union severally for one-half of his retainer for the current month, one-half of the amount due for per diem services rendered and expenses incurred during the preceding month.
- 4. In the event either the Company or the Union desires to terminate the services of the

Permanent Arbitrator prior to the expiration date of this Agreement, it may do so by giving notice in writing to the Permanent Arbitrator and the other party. Such notice shall include a termination date and shall be effective as of such date. Thereafter the Permanent Arbitrator shall accept no new cases nor hold further hearings (except to complete hearings, if any, in which the evidence has not been completed); he shall, however, complete any hearings in which partial evidence has been heard and render decision thereon and on any pending cases in which hearings have been completed. Each of the parties shall pay the Permanent Arbitrator one-half (1/2) of all unpaid retainer due on said termination date, all unpaid fees, and all unpaid expenses, and in addition thereto the party issuing such termination notice shall make a further payment of Dollars (\$ ) as a termination fee to the Permanent Arbitrator for loss and expenses resulting to him from such termination.

- 5. The Permanent Arbitrator, at any time, by written notice to the Company and the Union, may terminate his services under this Agreement, provided, however, that if not incapacitated he shall render decision in all cases in which hearings have been completed. If the Permanent Arbitrator terminates his services as aforesaid, he agrees to waive all rights to further retainer but shall be entitled to such further per diem and expense payments as may be requested in concluding his services.
- 6. Although the Permanent Arbitrator shall be entitled to no termination fee if his authority terminates by expiration of the stated term, the provisions hereof as to completion and decision of uncompleted cases shall apply.
- 7. It is the intent, purpose and desire of all the parties hereto that every effort of the Union and the Company be directed toward the disposition of grievance cases before participation therein by the Permanent Arbitrator is requested.

### Letter of Agreement No. 2 RESLOTTING OF JOBS

In the event the Company assigns a particular job from one existing job classification to another existing but lower rated job classification, on the basis of significant changes occurring in the content of such job, subsequent to the date of the Local Agreement between us, supplementary to the Central Agreement between us dated March 1, 2023, the Union shall have the right to file a grievance regarding such action on the Company's part (including the right to arbitrate in accordance with the provisions of Article 6 of the Central Agreement).

The Union shall also have the right to file a grievance (including the right to arbitrate as above provided) requesting that a particular job be reclassified from one existing job classification to another existing but higher rated job classification provided such grievance is based on significant changes occurring in the content of such job subsequent to the date of the Local Agreement between us, supplementary to the Central Agreement between us dated March 1, 2023.

In the event either such grievance is submitted to arbitration, the arbitrator shall first determine whether the job in question has been significantly changed subsequent to such effective date. If the arbitrator determines that the content of the job in question has significantly changed as above described, then he shall determine into which of the existing classifications such job shall be assigned.

## Letter of Agreement No. 3 WAIVER OF TIME LIMITS

When vacation shutdowns occur (at those business units which observe a shutdown), the Company and the Union agree to waive the periods of such vacation shutdown at those business units with respect to the time limits imposed in the below-specified provisions of the Central and Local Agreements.

These periods of waiver apply only to the following time limits:

#### Central Agreement

The 30th day referred to in Section 3.1
The 5-day period of Section 5.5
The 60-day period of Section 6.1
The 80-day, 65-day, 10-day, 30-day, 10-day and the 30-day periods of Section 6.5
The 60-day period of Section 6.9

Local 974 Agreement
The 10-day period of Section 3.1

Local 751 Agreement
The 5-day period of Section 5.4

Local 2096 Agreement
The 10-day period of Section 3.2

The parties also agree to waive any periods of vacation shutdown for the International Union with respect to the time limits imposed in Sections 6.1 and 6.5 of the Central Agreement.

### Letter of Agreement No. 4 UNION EDUCATION SCHOLARSHIPS

It is understood and agreed between the parties that notwithstanding any other provisions limiting the number of concurrent leaves of absence for Union business, employees who are selected by the Union to receive scholarships to attend educational programs at the Union's Black Lake Center, will be given leaves of absence for not more than two weeks (to be taken consecutively) for this purpose. In any given calendar year, no more than two such leaves will be requested or granted for any Business Unit covered by the Central Agreement between the parties except only one such leave will be requested or granted at the Caterpillar Logistics Services Business Units.

It is further agreed that, notwithstanding any existing local practices or agreements relative to working during the vacation shutdown, such employees will be offered a preferential opportunity to work during the summer vacation shutdown next following their designation as scholarship recipients if there is work available in the employees' business unit which they are qualified to perform. Notice of designation of scholarship recipients must be provided to the Company not later than April 1 of each year; otherwise the opportunity to work will be provided during the summer vacation shutdown next following such leave.

### Letter of Agreement No. 5 EMPLOYEE INVOLVEMENT

The parties agree that employee involvement in workplace issues is crucial to the success of any business and the responsibility of every employee. More specifically, they agree that in the interest of creating a safe, highly productive work environment with widespread employee involvement and satisfaction, free and open communication between management and bargaining unit employees is necessary. To this end, the parties agree that the Company may establish teams consisting of selected bargaining unit employees (and as appropriate, non-bargaining unit employees), to cooperate with management in the identification of best practice for the improvement of efficiency, effectiveness, safety, productivity, quality, and profitability regarding operations about which they have knowledge. Examples of such teams include certification teams, productivity teams, problemsolving teams, new product development teams, safety teams, cost reduction teams, and work teams. It is a part of each employee's job duties to participate and cooperate in these endeavors. Any overnight travel associated with any inter-business unit teams formed under this Letter of Agreement will be on a voluntary basis.

The parties recognize that any team formed under this Letter of Agreement is not a replacement of or substitute for the long standing practice of individuals and/or teams formed for purposes such as but not limited to new machine runoffs, machine field follow-up or customer and OEM input/quality trips.

The parties also recognize that no team may take any action which conflicts with any provision of the Central Agreement or the respective Local Agreement. Moreover, the parties recognize that certain teams composed of bargaining unit employees (in whole or in part) may submit ideas or specific recommendations to management, which relate to wages, hours, and working conditions. While management may discuss and consider such ideas or proposals, no change or addition to the written terms and conditions of employment set forth in the Central Agreement or the respective Local Agreement may be made unless such change or addition is bargained over and agreed to by the Company and the Union.

## Letter of Agreement No. 6 EQUAL EMPLOYMENT OPPORTUNITY COMMITTEE

Caterpillar Inc. and the International Union UAW have long recognized both the moral and legal obligation and the practical desirability of developing and implementing policies which will insure equal employment opportunity, and, to the extent possible, consistent with responsible manufacturing operations, the elimination of the effects of discrimination if it may have existed in past years.

Accordingly, the parties agree to the establishment within thirty days of the ratification of this Agreement of a joint committee at the Corporate-International Union level and, additionally, joint committees at each Company business unit as set forth in Section 4.7 of the Central Agreement where employees are represented by a Local Union as set forth in Section 2.1. The consist, function and objectives of these committees will be as follows:

#### **Central Committee**

1. The Joint Central Equal Employment Opportunity Committee will consist of two representatives of the Company and two representatives of the International Union to be

- appointed by the parties.
- 2. The Joint Committee shall appoint a Chairman and a Secretary from among the members serving. When a Company representative is Chairman, a Union representative will be Secretary and vice versa.
- 3. The Joint Committee will meet as frequently and at times as may be mutually agreed upon.
- 4. The Secretary will keep minutes of meetings, copies of which will be provided to the Company and Union representatives on the committee within a reasonable time following each meeting.
- 5. The function and objectives of the committee will be:
  - (a) To develop means by which minority and/or disadvantaged persons will be encouraged to participate in apprentice or training programs in order to enhance their promotability.
  - (b) To keep informed on affirmative concepts and programs developed by other companies or by governmental agencies and consider ways and means of adapting same to the advantage of Company employees.
  - (c) To develop and adopt guidelines for employees and Union representatives designed to encourage the use of the grievance procedure as the primary method of resolving equal employment opportunity problems.
  - (d) To review the progress of and provide counsel to the local business unit committees.

#### **Business Unit Committees**

- 1. The Business Unit Equal Employment Opportunity Committee at each business unit as set forth in Section 4.7 of the Central Agreement where employees are represented by a Local Union as set forth in Section 2.1 will consist of three representatives of the Company and three representatives of the Union to consist of the Local Union President, Chairman of the Bargaining Committee, and Chairman of the Fair Employment Practices Committee except in Local 974. In Local 974 each business unit committee will consist of equal numbers of business unit management and Union representatives.
- 2. The Business Unit Committees will each appoint a Chairman and a Secretary from among the members serving. When a Company representative is Chairman, a Union representative will be Secretary and vice versa.
- 3. The Business Unit Committees will meet at least quarterly at mutually agreed times. Union representatives who are members of these committees will not lose pay from regularly scheduled hours for time spent in attendance at scheduled meetings up to a maximum of three hours for any quarter. However, if the aggregate time spent in such meetings in any one-quarter is less than three hours, the remainder shall not be carried over into a succeeding quarter.
- 4. The Secretary will keep minutes of meetings, copies of which will be provided to the Company and Union representatives on the Committee and to the Joint Central Committee

within a reasonable time following each meeting.

5. The function and objectives of the Business Unit Committees will be similar to the first three functions and objectives of the Joint Central Committee. Local developments or proposals in these three areas will be channeled specifically to the Joint Central Committee by means of the minutes of Business Unit Committee meetings.

As stated in the opening paragraph of this letter, Caterpillar Inc. and the International Union UAW recognize the several aspects of responsibility inherent in their respective obligations to avoid discrimination. In particular, the legal responsibilities, which of course cannot be delegated, require that the activities of the Central Committee and the several Business Unit Committees shall be advisory, consultative and cooperative only. The Company and the Union will give due consideration to the recommendations of the Committees but the Committees may not commit either the Company or the Union to a specific course of action. However, the Union agrees that it will discourage its members from bypassing the grievance procedure with respect to any allegations against the Company, which may be made the subject of a grievance under the Labor Agreement.

### Letter of Agreement No. 7 GOVERNMENTAL AGENCY RULINGS

If a governmental agency having appropriate authority holds that any increase in rates of pay or benefits provided for by this Agreement or any supplement thereto is disallowed or postponed, the Company will periodically, as the proscribed payments become due, place in escrow an amount of money equal to that necessary to provide the rates of pay and benefits so disallowed or postponed, if so doing is permissible under government regulations. The parties will negotiate, without strike, lockout or other interference with production, and without arbitration, means of making available to employees benefits equal in value to any monies so deposited in escrow in a manner permissible under government regulations.

## Letter of Agreement No. 8 PROTECTIVE CLOTHING

The Company agrees to furnish coveralls or other suitable clothing (which may include paper clothes) to employees in Locals 751, 1872, 974, and 2096 who are assigned to one of the job classifications indicated below and whose job duties regularly involve the use of spray painting equipment at least fifty percent of the time.

Materials Specialist - 2 Product Painter - 3 Maintenance Painter - 4 Painter, Maintenance - 4 Painter (Maintenance) - 4

After initially issuing such clothing to such employees, the Company will provide clean or new changes once each week on an exchange basis only.

The Company agrees to furnish and replace on an exchange basis (provided such exchange shall not exceed once per calendar quarter) welders jackets for employees whose job duties regularly require use of open arc welding processes.

The Company agrees to supply a new or refurbished welding helmet and associated equipment as required to perform welding work. Welding helmets and associated equipment will be refurbished and cleaned in accordance with the manufacturer's recommendations. When the nature of an employee's work results in damage to welding helmets and/or associated equipment to the extent that the Company's Safety Supervisor advises replacement, the replacement cost will be borne by the Company.

### Letter of Agreement No. 9 NOTICE OF CONTRACTED WORK

During 2017 negotiations the parties earnestly discussed the continued need for the Notice of Contracted Work at all locations. The parties agreed the Company would continue to provide the Notice of Contracted Work only at Decatur and Mapleton.

Upon request of the Grievance Committee Chairman, and not more than one time per calendar month, the Company will provide the Union with a list of all decisions that have been made the previous month as to Maintenance, Construction, and Tool Room work that has been performed by other than Company forces. Such list will include:

- 1. Identification of the work.
- 2. Name of subcontractor.
- 3. Reason(s) for subcontracting such work.

If the Union has questions on the above information, the Company will discuss the matter with the appropriate Final Step Committee.

### Letter of Agreement No. 10 REINSTATEMENT OF GRIEVANCES

The parties acknowledge the desirability of ensuring prompt, fair and final resolution of employee grievances. The parties also recognized that the maintenance of a stable, effective and dependable grievance procedure is necessary to implement the foregoing principle to which they both subscribe. Accordingly, the parties view any attempt to reinstate a grievance properly disposed of as contrary to the purpose for which the grievance procedure was established and violative of the fundamental principles of collective bargaining.

However, in those instances where the International Union, UAW, by either its Executive Board, Public Review Board, or Constitutional Convention Appeals Committee, has reviewed the disposition of a grievance and found that such disposition was improperly effected by the Union or a Union representative involved, the Union's Vice President and Director of the Agricultural Implement Department may inform the Company's Director of Corporate Labor Relations in writing that such grievance is reinstated in the grievance procedure at the step at which the original disposition of the grievance occurred. For the purposes of this Letter of Agreement No. 10, an improper disposition shall include only those instances wherein a grievance:

a. is specifically settled or specifically withdrawn in one of the steps of the Local grievance procedure or in an arbitration docketing session; or

b. is not referred by the Regional Director or his designated representative to the UAW Agricultural-Implement Department or appealed to arbitration by the UAW Agricultural-Implement Department as provided in Section 6.1 of the Central Agreement and is therefore deemed closed.

It is agreed, however, that the Company will not be liable for any claims for damages, including back pay claims, arising out of the grievance that either are already barred under the provisions of the Central Agreement or appropriate Local Agreement at the time of the reinstatement of the grievance or that relate to the period between the time of the original disposition and the time of the reinstatement as provided herein. It is further agreed that the reinstatement of any such grievance shall be conditioned upon the prior agreement of the Union and the employee or employees involved that none of them will thereafter pursue such claims for damages against the Company in the grievance procedure, or in any court or before any federal, state or municipal agency.

Notwithstanding the foregoing, a decision of the Permanent Arbitrator or any other arbitrator on any grievance shall continue to be final and binding on the Union and its members, the employee or employees involved and the Company and such grievance shall not be subject to reinstatement.

This letter is not to be construed as modifying in any way either the rights or obligations of the parties under the terms of the aforementioned Central Agreement or appropriate Local Agreements, except as specifically limited herein, and does not affect Sections thereof that cancel financial liability or limit the payment or retroactivity of any claim, including claims for back wages, or that provide for the final and binding nature of any decisions by the Permanent Arbitrator or other grievance resolutions.

It is understood this letter and the parties' obligations to reinstate grievances as provided herein can be terminated by either party upon thirty (30) days notice in writing to the other.

It is agreed that none of the above provisions will be applicable to any case settled, withdrawn, or deemed closed prior to the effective date of this letter.

## Letter of Agreement No. 11 RETURNS TO THE UNIT

An employee being returned to the bargaining unit under the provisions of Section 11.3 of the Central Agreement will be placed on a job in accordance with the following procedure:

1. Such employee will be placed on an open job in the classification or successor classification, if applicable, within the Business Unit in which such employee worked at the time he/she last transferred out of the bargaining unit provided there is no employee within that Business Unit with greater seniority who has previously worked in that classification (or its successor classification) and whose job bid for return to such classification has not been honored.

An employee being returned to a bargaining unit in accordance with this letter shall not be placed on an opening in Labor Grade 1, 2 or 3 if there is an employee with greater seniority who is laid off from that business unit, who possesses the basic qualifications which have been established for that particular job, and who meets the physical requirements which are necessary in order to perform that particular job.

An employee being returned to a bargaining unit in accordance with this letter shall not be placed on an opening in Labor Grade 4, 5 or 6 if there is an employee with greater seniority who is laid off from that business unit, who has previously worked and successfully performed in that classification and who meets the physical requirements which are necessary in order to perform that particular job.

2. In the event such employee cannot be placed on an open job as provided in (1) above, such employee will be placed in the reduction in force procedure in accordance with the applicable provisions of Article 13 of the Central Agreement using as the starting point of such reduction in force procedure the classification (or its successor classification) and business unit to which the employee was last assigned within the bargaining unit.

### Letter of Agreement No. 12 LOCKOUT, HAZARDOUS MATERIAL

During our negotiations, the parties discussed various aspects of Caterpillar safety programs. It was recognized that, to be effective, these programs require training of employees in their proper use and further requires appropriate enforcement measures. In addition, to maintain effectiveness, it is necessary to periodically review and reemphasize these programs.

#### Lockout Procedure

The Company will continue to provide training on the safety lockout procedure to all employees and apprentices who are required to observe such procedure. Locks and other necessary equipment will also continue to be provided to such employees.

The Company will instruct its management at each business unit to review the procedure with its Local Health and Safety Committee during one of its scheduled meetings. The Health and Safety Committee will also be informed of any subsequent significant changes made to the procedure.

#### Hazardous Material Program

The Hazardous Material Program will be a joint program.

The Company will provide each employee who comes in contact with hazardous materials with training on the safe use of such materials. When a hazardous material is introduced into an employee's work area, the employee will receive additional training, if necessary, prior to the use of such material. The Company agrees to provide employees exposed to hazardous materials with the Material Safety Data Sheets for those materials during the shift that the request was made, provided sufficient time is available.

The Company is committed to purchase only those hazardous materials that have adequate Material Safety Data Sheets and labels. Also the Company will request that suppliers of hazardous materials provide it with all identifiable chemical names and composition of such products on their Material Safety Data Sheets. The Company will confirm through toxicology references that supplier-provided health warnings are appropriate. The Company is committed to the proper labeling of all transfer containers used to carry hazardous materials with appropriate labels and will make an ongoing effort to accomplish this goal.

Further, the Company will make an ongoing, good faith effort to maintain consistent enforcement

of these programs. These programs will be monitored by business unit personnel (and Corporate Offices) in order to assure that each facility has an effective safety lockout procedure and hazardous material program.

#### Volunteer Emergency Response Program

During the course of 2004 bargaining, the parties have discussed emergency first responders, volunteer emergency medical technicians (EMT's), automatic external defibrillators (AED's), the role of Company security forces in emergency response, and the role of all such resources in emergency situations. As a result of these discussions, the Company renewed it's commitment to maintain the current volunteer emergency response programs, including continuing to seek volunteers (with the assistance of the plant safety committee) willing to achieve certification in cardio pulmonary resuscitation (CPR) and the use of AED's.

While most Company facilities have 24 hour, 7 day a week coverage or at least coverage at all times when employees are working, the Company committed that where such coverage was not practical, it would seek (with the assistance of the plant safety committee for such locations) volunteers willing to be trained in CPR and the use of AED's. Where sufficient volunteers could be found willing to complete the training on an off shift basis, and where a minimum of two (2) volunteers per shift could be certified after completing training, Caterpillar would make available the necessary equipment for response to emergency situations. The parties agreed that where sufficient volunteers could not be identified and certified, those facilities would continue to rely on the appropriate municipal emergency response obtained through the community 911 emergency call system.

The parties recognize the value of these programs to the health and safety of all employees. Therefore, the parties pledge to encourage volunteers to not take any action using their participation as leverage for matters that properly belong in one of the established dispute resolution processes.

## Letter of Agreement No. 13 TRAINING STATIONS

During 1979 negotiations, the parties discussed the Company's creation of training stations for apprentices and trainees. The Union agrees that a variety of training stations must be established in order to provide the different types of work assignments that are necessary to train apprentices and trainees.

As apprentices and trainees progress through the various programs, it becomes necessary to change the number and type of training stations. However, the Company will, on a semiannual basis, give the Union an updated listing of the training stations (including those stations where apprentices and trainees are assigned on an intermittent basis) within each bargaining unit covered by this Agreement.

The Company further agrees that, to the greatest extent practical, no employee will be involuntarily displaced to create a new training station. In addition, the Company will create training stations on more than one shift when warranted by the number of apprentices and/or trainees.

Letter of Agreement No. 14 ARBITRATION SCHEDULING During the 2011 negotiations the parties had extensive discussions concerning the buildup in the number of grievances pending arbitration that had occurred during previous years. Significant modifications were made to this Letter of Agreement to address the Union's concerns.

The parties have agreed upon the following arbitration schedule for 2023 through 2028.

Panel Arbitrator		Permanent Arbitrator				
Discharges, Suspensions		Issues				
Jan.	2 days East Peoria	open 2 days				
	1 day Morton					
Feb.	2 days Open					
Mar.	2 days Open – to be scheduled	open 3 days				
Apr.	1 day Mapleton	open 3 days				
May	2 days Decatur					
	2 days East Peoria					
June	1 day Pontiac	open 2 days				
	1 day Morton					
	1 day York					
Aug.	2 days Open					
	2 days Open - to be scheduled					
Sept.	1 day Morton					
	3 days East Peoria					
Oct.	1 day Pontiac	open 2 days				
	2 days Open - to be scheduled					
Nov.	2 days Decatur	open 3 days				
Dec.	1 day Mapleton	-				
	1 day Technical Center and Peoria Proving	g Ground				

It is the objective of the parties to fully utilize the scheduled days of arbitration.

Notwithstanding the above arbitration scheduling, the parties recognize the undesirable effects caused by having a buildup in the number of grievances pending arbitration. Quarterly during the life of the Agreement, representatives from the UAW-Agricultural Implement Department and Corporate Labor Relations Office will consider whether such an undesirable buildup exists at any facility. If they agree that it does, the representatives from the UAW-Agricultural Implement Department and Corporate Labor Relations Office will meet in an attempt to resolve any grievances pending arbitration by determining whether or not the facts of the case have resulted in a violation of the provisions cited.

The Regional Director of the UAW (or his designated representatives), the Bargaining Chairman, in the case of Local 974 the appropriate Final Step Grievance Chairman, and the Labor Relations Manager (or his designated representatives) may attend such meetings.

#### Letter of Agreement No. 15 CONTRACT WORK

During 1979 negotiations the parties discussed the Company's use of contractors to perform work within our Plant Engineering Departments, which the Union claimed should have been performed by UAW bargaining unit employees.

The Company, while noting that during the life of the 1976 Agreement skilled trades employment grew at a faster rate than total employment, appreciates the concerns expressed by individual members of the Bargaining Committee.

It is neither practical nor feasible for the Company to perform all of its plant engineering work with its own work force. The Company shall continue to take into consideration all the factors it has agreed to consider, under Section 16.10, when making those decisions.

The parties also discussed the matter of factory servicemen working on production, Tool Room or Technical Center machines that have been accepted and are operating within the various Caterpillar plants. The parties recognized that in many cases the appropriate craft employee, including a machine repairman, might gain necessary knowledge and experience by working with the factory representative servicing the production machine.

Therefore, insofar as is practical, the appropriate craft employee will be assigned to work with factory representatives on those production, Tool Room or Technical Center machines which the Company intends to regularly repair and maintain with bargaining unit employees.

### Letter of Agreement No. 16 VACATION USE

If an employee believes that a request for vacation has been unfairly denied, the employee should bring the request to the attention of the second-level supervisor who will investigate the reason(s) for the denial of the request. If the employee is still dissatisfied with the second-level supervisor's explanation for the denial of the request, the employee's Committeeman will be permitted to discuss the situation with the Labor Relations Manager.

## Letter of Agreement No. 17 TRAINING STATIONS - STUDENTS & TEACHERS

In order to continue attracting and retaining the highest caliber employees to careers in manufacturing, it is mutually beneficial to the community, the Union, and the Company to provide opportunities for exposing students and teachers to such careers. During 1991-1998 negotiations, the Company informed the Union it intended to provide a chance for students and teachers to become more familiar with industrial career opportunities and to provide schools with training stations for their students. Toward that end, the Company will from time to time designate bargaining unit work assignments for students and/or teachers at the high school or college level. Since the purpose of this program is to provide as broad an exposure as possible in the time allotted, candidates may be rotated periodically among the designated jobs. The work being performed by students and teachers will not result in the displacement or elimination of the on-going need of an employee covered by the Central Agreement. Caterpillar employees will be expected to provide appropriate orientation and assistance to participants in order to maximize the educational experience.

#### Letter of Agreement No. 18 GRIEVANCE PREVENTION AND RESOLUTION

During negotiations the parties reaffirmed their commitment to seeking new approaches to the ways they conduct their business with each other. Among the subjects discussed in this regard was a mutual concern about the ability to effectively resolve grievances, which were already pending

arbitration at some locations, and problems, which may arise during the life of the current Central Agreement. As a result of these discussions, the Union and the Company have agreed that Union and management officials from a facility may by mutual agreement selectively introduce the following jointly developed alternative concept for grievance prevention and resolution.

The parties have agreed that the following principles shall guide the implementation and functioning of this Grievance Prevention and Resolution program:

- Commitment of the International Union and Corporate Labor Relations to develop a constructive relationship based on trust and respect.
- Commitment of Plant Management and Local Union leadership to develop a constructive relationship based on trust and respect.
- Agreement by both parties that neither party should give up contractual rights and obligations to resolve grievances.
- Recognition by both parties of the rights and roles of the other party.
- Effort by both parties to resolve grievances at the earliest possible step.
- Commitment of Union representatives not to encourage and to actively discourage the filing of frivolous and/or repetitive grievances.
- Commitment of Company representatives to approach grievance settlement with flexibility and willingness to resolve underlying causes.
- Direction from upper management to supervision to attempt to resolve grievances at the earliest step.
- Direction from Union leadership to shop floor union representatives to attempt to resolve grievances at the earliest step.
- Commitment by both parties that their representatives have the right, the obligation, and the responsibility to resolve grievances at their respective levels.
- Recognition by Company and Union representatives at all levels that the parties will not always agree, but that they can disagree without being disagreeable.
- Commitment by both parties to steady improvement in their ability to work with each other to resolve problems and recognition that neither party should expect an instant turnaround as this approach is implemented.
- Recognition by both parties that their primary purpose is to provide quality products or services to customers and not to win arguments.
- Commitment by both parties to approach problem resolution on the basis of seeking the right solution "What's right rather than who's right."

An initial explanation of the basic principles will be made to Union and management officials who participate in the grievance procedure at the facility. Once the participants understand their responsibilities under this approach, Union and management officials from that facility shall begin implementation of the process by initiating the grievance prevention program and attempting to resolve any current grievances at their existing level. In order to permit a full and objective evaluation of these grievances by both parties, it may be necessary to suspend the normal grievance/arbitration procedure.

Top Union and management officials at a facility implementing this procedure shall meet periodically to evaluate the progress of this approach and to reinforce both parties' commitment to its principles.

In conjunction with the implementation of this approach, the arbitration of grievances at that facility may be suspended. Arbitration will continue for the remainder of the other locations, which are covered by the Central Agreement.

The parties will then initiate the following procedure to resolve those grievances, which are pending arbitration as of the agreed implementation date.

- Within sixty days of the agreed implementation date, members of the appropriate final step
  grievance committee, the appropriate regional servicing representative and representatives
  of the International Union shall review these grievances. A list of only those grievances
  believed to have merit will be submitted to the facility Labor Relations Manager. A grievance
  shall be deemed closed if it is not included on this list.
- 2. Within thirty days after receiving the Union list, management shall review the identified grievances and submit to the appropriate final step grievance committee a list of proposed settlements for those grievances which management agrees have merit.
- 3. Within thirty days following delivery of management's list of proposed settlements the parties shall meet and attempt to resolve any remaining grievances.
- 4. By mutual agreement of facility Union and management officials, grievances that are not resolved by this meeting may be submitted to the mediation procedure contained in Letter of Agreement No. 25. Absent mutual agreement to refer a case to mediation, the case would be automatically returned to the same pending arbitration status it held prior to implementing this procedure.
- 5. During the final step of the grievance procedure, concurrent with the company's answer of such grievance, either party may declare such grievance exempt from the Grievance Prevention and Resolution Program. Such exempted grievance would be processed as if the resolution program did not exist.
- 6. Grievances not exempted as provided for above would accumulate and be handled as follows:
  - A. No later than five calendar days following each February 1, May 1, August 1, and November 1 during the life of the Central Agreement, the Final Step Grievance Committee would deliver a list of such grievances to the Labor Relations Manager. Prior to such delivery, the members of the Final Step Grievance Committee and the appropriate Regional Servicing Representative will have reviewed such list consistent with the

principles and subject to the conditions of the first step of the resolution program.

- B. Within fifteen days (or longer if mutually agreeable) following delivery of the list to the Labor Relations Manager, a meeting will be held during which local management, the final step grievance committee, and the Regional Servicing Representative or his/her designated representative will review and attempt to resolve any remaining grievances.
- C. By mutual agreement of the final step grievance committee and management, grievances that are not resolved during this meeting may be submitted to the mediation procedure contained in Letter of Agreement No. 25. Absent mutual agreement to refer a grievance to mediation, the grievance would be automatically returned to the final step of the grievance procedure and would be deemed to be answered at the next regularly scheduled grievance meeting.
- D. Such grievance may then be referred to arbitration in accordance with the provisions of Article 6 of the Central Agreement if the Union is of the opinion that the Company's final step decision is unsatisfactory and that the grievance merits consideration for possible arbitration.

It is agreed that all grievances resolved by utilizing the provisions of this Letter of Agreement No. 18 or Letter of Agreement No. 25 will not set precedent for the resolution of any grievances in the grievance/arbitration procedure.

It is understood this letter and the parties' obligations to implement and participate in this procedure as provided herein can be terminated by either party at any plant or division covered by the Central Agreement upon thirty days notice in writing to the other.

### Letter of Agreement No. 19 PRIOR ARBITRATION – SECTION 12.1

The parties recognize that under prior arbitration authority, the Company has properly filled job openings by numerous means other than those listed in Section 12.1 of the Central Agreement. During 1986 negotiations, Article 12 was extensively revised. Nevertheless, it is understood that this same arbitration dicta as it relates to the interpretation of Section 12.1 remain in force and effect.

## Letter of Agreement No. 20 RETENTION RATE

During negotiations, the parties discussed the difficulties of retaining qualified employees on the Metal Treating Specialist-3, Metal Treating Specialist-4, Iron Pourer-4, Foundry Crane Operator-4, Melting Operations Controller-5, Melting Specialist-4, Ladle Repair Specialist-4, Heat Treat Manufacturing Systems Operator-4, Assembly and Test Specialist (3A64) in Local 751, Fabrication Specialist-3 and Fabrication Specialist-4 classifications. As a result of these discussions, the parties agreed to the following provisions:

1. Any employee assigned to the Metal Treating Specialist-3, Metal Treating Specialist-4, Iron Pourer-4, Foundry Crane Operator-4, Melting Operations Controller-5, Melting Specialist-4, Heat Treat Manufacturing Systems Operator-4, Ladle Repair Specialist-4, Assembly and Test Specialist (3A64) in Local 751, Fabrication Specialist-3 or Fabrication Specialist-4 classification at all locations who has been assigned to that classification shall be eligible to

apply for a Retention Bonus Rate.

2. The Retention Bonus Rate for Local 974 shall be two dollars (\$2.00) per hour for employees in the Metal Treating Specialist-4, Iron Pourer-4, Foundry Crane Operator-4, Melting Operations Controller-5, Melting Specialist-4, Ladle Repair Specialist-4, and Heat Treat Manufacturing Systems Operator-4. The Retention Bonus Rate shall be three dollars (\$3.00) per hour for employees in Fabrication Specialist-4 and Metal Treating Specialist-3 classification. The Retention Bonus Rate shall be four dollars (\$4.00) per hour for employees in the Fabrication Specialist-3 classification.

The Retention Bonus Rate for Local 751 shall be four dollars (\$4.00) per hour for employees in the Fabrication Specialist-4 classification and Fabrication Specialist-3 classification. The Retention Bonus Rate shall be one dollar (\$1.00) for employees in the Assembly and Test Specialist (3A64) classification.

The Retention Bonus Rate for Local 2096 shall be four dollars (\$4.00) per hour for employees in the Fabrication Specialist-4 and the Fabrication Specialist-3 classification. The Retention Bonus Rate shall be three dollars (\$3.00) per hour for employees in the Metal Treating Specialist-3 classification.

- 3. In order to apply for a Retention Bonus Rate, an employee must agree in writing to waive for a period of three years all rights to Bid under Article 12 except for the right to Bid to a job at Labor Grade 4 or higher within the NIO to which the employee is currently assigned and the right to bid for the same job on a different shift. Such three-year period shall commence on the date the employee begins to receive the Retention Bonus Rate.
- 4. The Company will make available to each eligible employee in the Metal Treating Specialist-3, Metal Treating Specialist-4, Iron Pourer-4, Foundry Crane Operator-4, Melting Operations Controller-5, Melting Specialist-4, Ladle Repair Specialist-4, Heat Treat Manufacturing Systems Operator-4, Assembly and Test Specialist (3A64) in Local 751, Fabrication Specialist-3 or Fabrication Specialist-4 classification a form on which to apply for the Retention Bonus Rate. An eligible employee may apply for the Retention Bonus Rate by completing the form and submitting the form to his/her supervisor.
- 5. The Retention Bonus Rate shall begin within 30 days of the date the application is submitted to the supervisor.
- 6. At the conclusion of the three-year period, an employee receiving the Retention Bonus Rate will have the option of either:
  - a. continuing to receive the Retention Bonus Rate and agreeing to waive bidding rights for another three-year period; or
  - b. ending the Retention Bonus Rate and resuming all bidding rights under Article 12 of the Central Agreement.
- 7. The Retention Bonus Rate shall automatically end for any employee who, for any reason, leaves the Metal Treating Specialist-3, Metal Treating Specialist-4, Iron Pourer-4, Foundry Crane Operator-4, Melting Operations Controller-5, Melting Specialist-4, Ladle Repair Specialist-4, Heat Treat Manufacturing Systems Operator-4, Assembly and Test Specialist

(3A64) in Local 751, Fabrication Specialist-3 classification or Fabrication Specialist-4. Concurrent with ending the Retention Bonus Rate, the employee will resume all bidding rights under Article 12 of the Central Agreement.

- 8. An employee who elects to end the Retention Bonus Rate under 6 (b) above or for whom the Retention Bonus Rate is automatically ended under 7 above may reapply for the Retention Bonus Rate at any time provided the employee is assigned to the Metal Treating Specialist-3, Metal Treating Specialist-4, Iron Pourer-4, Foundry Crane Operator-4, Melting Operations Controller-5, Melting Specialist-4, Ladle Repair Specialist-4, Heat Treat Manufacturing Systems Operator-4, Assembly and Test Specialist (3A64) in Local 751, Fabrication Specialist-3 or Fabrication Specialist-4 classification.
- 9. The Company reserves the right to increase the amount of the retention bonus rate for any classification listed in this letter within any business unit or all of the business units.

The parties agree that this Letter of Agreement shall remain in effect only until the expiration of the Central Agreement and shall not be renewed or extended except by express written agreement of the parties.

## Letter of Agreement No. 21 REWARD & RECOGNITION PROGRAMS

The parties agree that rewarding and/or recognizing outstanding performance or exemplary behavior by an employee or group of employees is crucial to the success of any business in creating a safe, highly productive work environment with widespread employee involvement and satisfaction. The Company has developed and implemented a wide variety of reward and recognition programs within the different business units over the years to demonstrate its appreciation for employees' efforts. These programs have included, but have not been limited to, such things as clothing (jackets; t-shirts; hats; etc), gift certificates, event tickets, trips, and money.

### Letter of Agreement No. 22 MEDICAL RESTRICTIONS LAID-OFF EMPLOYEES

During the 1988 negotiations the parties discussed the dilemma faced by laid-off employees who have medical restrictions. In the past there have been instances where an employee was on layoff status and could not be recalled in line with his seniority and qualifications because of medical restrictions. In such cases the employee, upon subsequent recovery (full or partial), then was required to wait for another opening before being recalled on the basis of qualifications and then current medical condition. Employees in such circumstances were confronted with the possibility of having their seniority broken prior to jobs becoming available which fit their medical restrictions.

In order to address this situation the parties agreed to provide such employees with an additional opportunity to be returned to work in the following circumstances. In the event an employee's medical restrictions are removed or modified to the extent such employee could perform a job in an IO or NIO, the laid-off employee shall displace the employee with the least seniority who is assigned to such job provided:

(i) in the case of a job in the IO, such employee has more seniority than an employee who is performing such job and possesses all basic qualifications and abilities necessary to perform the job within a reasonable period as defined in Section 13.5(i) of the Central Agreement, or

(ii) in the case of a job in an NIO, such employee has more seniority than an employee who is performing such job and has previously performed such job.

The employee who is displaced as a result of this Letter of Agreement shall thereafter be placed in accordance with the provisions of Section 13.5 of the Central Agreement.

### Letter of Agreement No. 23 CATERPILLAR JOB SECURITY PROGRAM

The Company and the Union have agreed to provide the following Caterpillar Job Security Program (hereinafter referred to as CJSP or the Program). The cornerstone of the Program is the Company's commitment that eligible employees will not be subject to indefinite layoff during the term of this Letter of Agreement except as provided herein. The Program has been designed to, in part, protect employees against indefinite layoffs resulting from outsourcing and other managerial actions. However, it should not be inferred from the provisions contained herein that the Program either adds to or detracts from the Company's rights to outsource or take other actions that are otherwise not restricted by the Central Agreement.

#### I. Definitions

- A. Active Employee Active employees, for the purposes of this Letter of Agreement, shall include employees actively at work and those employees not at work due to time off for vacation, paid absence allowance, bereavement, jury duty or witness service, unexcused absence, short-term illness including short-term disability and Workers' Compensation, short-term personal leave, LTD, union leaves of absence, or any other leave of absence. Active employees shall exclude supplemental employees, and all employees on indefinite layoff.
- B. Attrition Attrition means CJSP-eligible employees who retire, die, quit, or otherwise cease to be employed by the Company in a business unit on a job in a bargaining unit covered by this Program. For all purposes under this Program, a CJSP-eligible employee's attrition date shall be deemed to be the last day actually worked by the employee.
- C. Business Unit Business unit(s) means those business unit(s) or divisions identified in Section 4.7 of the Central Agreement.
- D. Eligible Employee An eligible employee shall be an employee within the bargaining unit within one of the business units who meets one or more of the following qualifying conditions:
  - i. Was an active employee in the business unit (excluding York) as of June 6, 1992; or
  - ii Was recalled to work after June 6, 1992, and has 1.5 years seniority following such recall (excluding York); or
  - iii. Was hired or rehired after June 6, 1992 but prior to January 10, 2005, and has 5 years seniority following such hire or rehire (excluding York); or

- iv. Was hired or rehired to work on or after January 10, 2005 but prior to March 1, 2013, and 12 years seniority following such hire or rehire (excluding York); or
- v. Was moved into a business unit from another business unit in accordance with the applicable provisions of the Central and Local Agreements, or was Company transferred from one business unit to another business unit, and in either case was an eligible employee in the previous business unit.
- E. Nonqualifying Actions Nonqualifying actions are those against which CJSP-eligible employees are not protected and which may result in a reduction of the benefits provided under this Program including a reduction in the forty hours of available work and/or pay in each workweek. They include:
  - i. Temporary layoffs due to material shortage, equipment failure, power failure, labor dispute, plant re-arrangement, retooling, or other circumstances as provided in Section 13.2 of the Central Agreement and/or the applicable Local Agreement;
  - ii. Participation in strikes, walkouts, or other forms of labor disputes;
  - iii. Reductions in force occurring as a result of strikes, walkouts, and other forms of labor disputes;
  - iv. Reductions arising out of or resulting from Acts of God, terrorism, war, or other events beyond the Company's control;
  - v. A sale of the Company or all facilities within a bargaining unit (as defined in Section 2.1 of the Central Agreement) which shall terminate this Letter of Agreement and all of the benefits hereunder for all affected employees, as of the closing date of the sale;
  - vi. Any Complete Plant Closing as defined in Section 16.10 of the Central Agreement; or
  - vii. Temporary Shutdowns/Layoffs for the purpose of reducing or avoiding an increase in inventory of finished products because of sales prospects as provided in Section 13.6 of the Central Agreement and/or the applicable Local Agreement.

An employee affected by any of the above nonqualifying actions will, if otherwise eligible, be covered by other benefits and obligations that may exist under other negotiated agreements between the parties.

F. Qualifying Actions — A qualifying action is any event that would, absent the protections provided herein, result in the indefinite layoff of a CJSP-eligible employee. Qualifying actions shall include, but not be limited to, sourcing decisions, introduction of technology, productivity improvements, consolidation of operations, and volume reductions; they shall exclude actions or events specified in Section I (E).

#### II. Benefits

The obligation to provide benefits hereunder shall not arise unless one or more qualifying actions would otherwise result in the layoff (other than layoffs described under

"nonqualifying actions" above) of a CJSP-eligible employee. The benefits provided under this Program can be generally defined as the opportunity for active work and/or other constructive activities and assignments provided for in Section V below for at least forty hours in each workweek or for forty hours' straight-time pay if no work or activities are assigned or for some combination thereof.

#### III. Losing Eligibility

An otherwise eligible employee shall lose the rights and benefits of that status under the following circumstances:

- i Discharge for "just cause";
- ii. Disciplinary suspension (for the period of the suspension);
- iii. Voluntary termination or for any reason whereby the employee ceases to be an active employee as defined above;
- iv. Retirement;
- v. Death;
- vi. Refusing placement into an existing opening in another Business Unit (excluding York) as defined by Section 4.7;
- vii. Refusing placement into a CJSP Pool;
- viii. Refusing assignment once in a CJSP Pool; if a discharged or separated employee who previously was an eligible employee is subsequently reinstated, the ineligibility shall be reinstated.

#### IV. Indefinite Layoffs

Employees who are not eligible employees may be placed on indefinite layoff even if their seniority (including preferential seniority) is greater than the seniority of eligible employees who are retained at work.

Employees who refuse placement in the CJSP Pool or who refuse an assignment after being placed in a CJSP Pool may be placed on indefinite layoff even if their seniority (including preferential seniority) is greater than the seniority of eligible employees who are retained at work.

#### V. CJSP Pools

- A. In the event that declining manpower needs and the application of the reduction-in-force procedures would otherwise cause an eligible employee(s) to be placed on indefinite layoff, then such employee(s) will be placed in a CJSP Pool.
- B. The parties recognize that the scope of this program requires flexibility with regard to the assignment by the Company of duties and activities to Pool employees and the selection

of Pool employees for training. An employee in the Pool may be:

- (i) placed in a training program,
- (ii) used as a replacement to facilitate the training of another employee,
- (iii) given a job assignment within or outside the bargaining unit which may be nontraditional,
- (iv) placed in an existing opening, or
- (v) given other assignments consistent with the purposes of this Program (including assignments to work in another business unit within the geographic area).

In the event that the Company assigns no duties to a CJSP Pool employee, he or she shall remain available for assignment upon reasonable notice in accordance with rules and procedures established by the Company for administration of the CJSP Pool.

#### C. CJSP Pool Guidelines

- i. An employee in the Pool will receive the regular straight-time hourly rate of pay the employee last held prior to after being placed in the Pool. In the event an employee in the Pool is assigned to another classification, the employee will receive the rate of pay for that classification as provided by the Central Agreement.
- ii. Pool assignments will be considered temporary and not subject to provisions governing permanent filling of vacancies or the application of shift preference.
- iii. When, for training purposes, a non-Pool employee is replaced by a Pool employee and is given a different work assignment, such non-Pool employee will continue to receive his/her regular straight-time hourly rate of pay and will be returned to the same classification and job assignment upon completion of the training assignment. In the event the employee has insufficient seniority to return to the formerly held classification, the employee will be placed pursuant to the applicable provisions of the Central and Local Agreements.
- iv. A training assignment will be voluntary on the part of a non-Pool employee being replaced by a Pool employee, unless such training is to develop or improve technical skills relevant to the employee's current job assignment or anticipated future job needs.
- v. An employee may decline the opportunity to be assigned to the Pool or, while in the Pool, decline an assignment. In such event, the employee will no longer be an eligible employee under this Program and will be laid off until entitled to recall to a non-Pool position. In no case may the employee then claim a violation of seniority rights because a lesser seniority employee is in the Pool or on a job assignment regardless of the lesser seniority employee's job assignment.

In the event an eligible employee accepts placement in another business unit under this Letter of Agreement, such placement will be considered to be at the request of the Company for purposes of determining eligibility under Section 16.6 of the Central Agreement.

In the event an eligible employee refuses placement under Section (III)(iv) of this Letter of Agreement, the employee will be separated and Caterpillar agrees to pay a lump sum payment. The amount of the lump sum payment will be based on the affected bargaining unit member's years of seniority as follows: a lump sum payment equal to twenty (20) hours multiplied by his/her then current regular rate of pay, and then multiplied by each full year of seniority at the time of separation.

#### VII. Administration

A National CJSP Committee will be established at the Company-International Union level consisting of three representatives selected by the Company and three representatives selected by the UAW Caterpillar Department. The National CJSP Committee will meet periodically as required to:

- A. Review the number of eligible employees and the size, if any, of CJSP Pools;
- B. Monitor the placement of employees who are assigned to a CJSP Pool;
- C. Review the work assignments of Pool employees to nontraditional work assignments;
- D. Review any complaint regarding the administration of the Caterpillar Job Security Program;
- E. Jointly develop and initiate proposals to improve operational effectiveness in order to secure existing jobs, and to attract customers and additional business in order to provide additional job opportunities.

The National CJSP Committee is specifically empowered to periodically review and evaluate the operation of this Letter of Agreement and make mutually satisfactory adjustments to its provisions during the term of this Letter of Agreement.

#### VIII. Duration

This Program shall become effective on the Monday following the ratification of the Central Agreement and shall remain in effect until midnight February 28, 2029, and shall not be renewed or extended except by express written agreement of the parties.

### Letter of Agreement No. 24 PLANT CLOSING RESTRICTION

As a result of deep concern about job security in our negotiations and the many discussions, which took place over it, this will confirm that until March 1, 2029, the Company will not close any facility covered by the Central Agreement, excluding Aurora.

It is understood that conditions may arise that are beyond the control of the Company; e.g., act of God, terrorism, war, etc. Should these conditions occur, the Company will discuss such conditions

with the Union.

## Letter of Agreement No. 25 MEDIATION

The parties agree that experience indicates the mediation process has advantages as an alternative to the regular arbitration procedure. Therefore, each facility covered by the Central Agreement may, by mutual agreement of Union and management officials at that facility, and with the involvement and assistance of the International Union and Corporate Labor Relations, implement this mediation procedure.

The parties recognize that mediation is not appropriate to deal with a large and/or growing backlog of grievances pending arbitration. Therefore, prior to the initial implementation of a mediation procedure at a facility having such a backlog, the jointly developed grievance prevention and resolution concept provided in Letter of Agreement No. 18 will be implemented. At a facility where a large or growing backlog is not a problem, facility Union and management officials may by mutual agreement submit grievances to mediation.

The mediation procedure will consist of the following:

- A. Grievances that have been referred to arbitration may be taken out of the backlog and referred to mediation by mutual agreement of facility management and final step grievance committee(s).
- B. Unless by mutual agreement otherwise, grievances referred to mediation shall be heard in the chronological order of their referral. A grievance referral date for the purposes of mediation will be considered to be the initial date on which a grievance was referred to arbitration.
- C. Mediation conferences will be informal. There will be one spokesperson for each party. Presentation of evidence is not limited by formal rules of evidence. Each party's presentation will be limited to a maximum of twenty minutes. The mediator will be allowed to meet privately with each party. There will be no record made of the mediation hearing.
- D. The Grievant shall have the right to be present at the mediation conference. If the issue involves a group grievance or multiple grievances on the same issue, these grievances will be consolidated and one employee will represent all Grievants.
- E. If no settlement is reached during the mediation conference, the mediator shall provide the parties with an immediate advisory oral opinion and the grounds for this opinion unless both parties agree that no opinion shall be provided. Either party may reject the advisory opinion.
- F. A mediator's rendering of an opinion will neither be considered evidence nor establish precedent for any purpose in the arbitration procedure provided by Article 6 of the Central Agreement. Further, nothing said or done by either party in a mediation conference may be referred to or used in any other grievance or mediation conference or used against that party in any grievance procedure discussion or arbitration, unless it was also said or done in a prior step of the grievance procedure.
- G. Each party will bear their own expenses, and the expenses of the mediator will be divided

equally between the parties.

- H. Grievances not settled during the mediation conference may be returned to the regular arbitration procedure. A grievance that is returned to arbitration will be returned to the same pending arbitration status it held prior to implementing this procedure. The Company must be notified in writing of the Union's intention to return a grievance to the arbitration procedure within thirty days of the mediation conference or the grievance will be deemed closed.
- I. In the event a grievance that was mediated goes to arbitration, no person that served as mediator may serve as arbitrator of that grievance.

### Letter of Agreement No. 26 LUMP SUM PAYMENTS

As specified below, payments shall be made to each eligible employee equal to four percent (4%) of the employee's basic hourly rate on the "eligibility date" times the number of hours worked by the employee during the "base period" less required deductions. The "base periods" and "eligibility dates" for purposes of computing such payments shall be as follows:

Base Period	Eligibility Date	Payment Not Later Than
January 1, 2023 – December 31, 2023	December 31, 2023	March 31, 2024
January 1, 2025 – December 31, 2025	December 31, 2025	March 31, 2026

### Letter of Agreement No. 27 SUPPLEMENTAL EMPLOYEES

#### Supplemental Employees

A. The parties to the Central Agreement recognize that the Company's Business Units encounter fluctuations in employment needs due to the cyclical nature of their business. Therefore, within each Business Unit, in order to provide stable employment and job security to our regular employees, the Company may, as the need arises, in addition to sourcing of work in accordance with the provisions of Section 16.10, employ the services of supplemental employees to perform bargaining unit work.

#### В.

- 1. Supplemental employees are either:
  - a. Employees hired or contracted by the Company to work forty (40) or more hours per week on a temporary but indefinite basis; or
  - b. Employees hired or contracted by the Company to work less than forty (40) hours per week.
- 2. As an aggregate within each business unit, the total number of supplemental employees regularly scheduled to work forty (40) hours or more per week may not exceed the higher of twenty-five employees (five employees when there are fewer than fifty regular bargaining unit employees) or twenty percent (20%) of the number of regular bargaining unit employees within the Business Unit. As an aggregate within each business unit, the

total number of supplemental employees regularly scheduled to work less than forty (40) hours per week may not exceed the higher of twenty-five employees (five employees when there are fewer than fifty regular bargaining unit employees) or five percent (5%) of the number of regular bargaining unit employees within the Business Unit. The foregoing limitations on the number of supplemental employees in this paragraph B (2) may be exceeded for a specific business unit by written mutual agreement between the Company and the Chairman of the Bargaining Committee for such business unit.

Notwithstanding the foregoing provision of this Letter of Agreement No. 27, if the Company elects to return material handling/logistics-work that had been outsourced to a supplier to the bargaining unit and utilizes supplemental employees who are assigned to jobs in Labor Grade 1 and/or Labor Grade 2 to perform such work, the number of supplemental employees performing such work will not be considered part of the total number of supplemental employees within the Business Unit for purposes of this Letter of Agreement No. 27.

- 3. The Company will not be obligated to separate supplemental employees to maintain the limits outlined in paragraph B (2) above, except in conjunction with a reduction in force resulting in a layoff of regular employees.
- C. The Company will not use supplemental employees in lieu of recalling laid-off employees to their business units unless there are either (1) rare and unusual circumstances where qualified, laid-off employees are not available or (2) anticipated short-term needs where the recall of a laid-off employee is not warranted. If either of these situations arises, the Company will review the matter with the Union and will give consideration to the Union's position before using supplemental employees.
- D. In the event of a reduction in force, a regular employee will displace a supplemental employee who is on a job that the regular employee is qualified to perform and to which the regular employee is contractually entitled.
- E. Contractual rights, seniority rights, and benefits, except those required by statute or those which the Company, at its discretion, may provide, shall not accrue to supplemental employees.
- F. Supplemental employees shall receive the same pay rate as would a newly hired regular employee assigned to the same job classification within that business unit.
- G. Notwithstanding the provisions of Section 7.7 of the Central Agreement, supplemental employees shall only be paid time and one-half for work in excess of forty hours in any week.
- H. Such supplemental employees may be terminated at any time for any reason. No claim of wrongful termination or discharge of a supplemental employee shall be taken up as a grievance, unless such claim is based on alleged personal prejudice or Union activity. Such grievances must be in writing and supported by written evidence. The foregoing provision shall not prohibit a supplemental employee from filing a grievance alleging a violation of other contractual provisions applicable to such employee.
- I. A supplemental employee will not be assigned to a job if there is a regular employee working on the same job but on a different shift and whose Prefiled Shift Preference form indicates

the shift of first preference is the same as the shift of such job. Instead, the regular employee already on the same job on a different shift, with such Prefiled Shift Preference form on record for at least forty calendar days, shall be moved to such job. No more than two shift changes shall be made under this paragraph for each original job.

- J. If a supplemental employee is hired as a regular employee, the employee shall receive credit with all workdays as a supplemental employee for purposes of completing the employee's probationary period under Section 11.1 of the Central Agreement.
- K. Once a supplemental employee has been continuously employed by the Company for a period of twenty-four (24) months, the employee shall be offered regular employment. If the supplemental employee accepts this offer, the employee shall be hired to regular employment at the beginning of the first pay period following acceptance of the offer, except that such employee will be placed on the same rate step of the job classification to which the Company hires such employee as the rate step to which such employee was assigned as a supplemental employee, regardless of job classification, on the day preceding such employee's first day of regular employment. If the supplemental employee refuses an offer of regular employment, then such individual shall remain a supplemental employee and the Company shall not be obligated to make additional offers of regular employment to such individual.

Notwithstanding the foregoing provision of this subsection (K), the Company shall not be obligated to offer regular employment to supplemental employees who are performing material handling/logistics work that had been outsourced to a supplier and returned to the bargaining unit.

### Letter of Agreement No. 28 TEMPORARY ASSIGNMENTS

During 2023 negotiations the Union expressed concerns with the duration of temporary assignments. The Company endeavors to manage temporary assignments in a fashion least disruptive to employees, however, certain situations may require temporary assignments of a longer duration. If the need for a temporary assignment arises, the following provisions shall apply:

- 1. An employee will not be on a temporary assignment for longer than 180 days.
- 2. When an employee's temporary assignment ends, they will not be placed on another temporary assignment for 60 days.

The following exceptions shall apply:

- A. A temporary assignment under 1 and 2 above may be modified by mutual agreement of the Business Unit Grievance Committee and the Company Final Step Representative. Such mutual agreement will not be unreasonably withheld.
- B. The temporary assignment limit does not apply to employees assigned to Team Lead or Non-Traditional positions.

Letter of Agreement No. 29 EMPLOYEE COMMITMENT TO BUSINESS SUCCESS In order to ensure a positive impression of our business for our customers and dealers, who from time to time tour our facilities, the Union and the Company agree that employees will not wear competitor apparel and/or display other competitor stickers, buttons, signage or similar items while on Caterpillar property.

### Letter of Agreement No. 30 VOLUNTARY SEPARATION PROGRAM

The Company may develop and initiate a Voluntary Separation Program (VSP) at any time during the term of this Agreement. The applicable "window period(s)" will be determined by the Company. The eligibility criteria shall provide that opportunities to participate in any VSP will be offered in seniority order within the applicable group(s). The applicable group(s) will be determined by the Company and may include but are not limited to any of the following or any combination thereof: job classification(s), plant(s), facility(ies), functional area(s), seniority group(s) and/or retirement eligibility. The Company shall determine the amount of the separation payment by multiplying the severance amount by the employee's full year(s) of seniority at the time of separation. Employees who accept such offer may be required to remain at work for up to four (4) months from the effective date of such acceptance to facilitate the training of employees who will replace them or to meet other operational needs of the Company.

## Letter of Agreement No. 31 HIRING BONUS

During the term of the 2023 Central Agreement, the Company may, at its discretion, offer a hiring bonus to newly hired or rehired bargaining unit employees (supplemental and/or regular) within any business unit or all business units, in any job classification or all job classifications, for purposes of attracting new employees to bargaining unit positions. The amount of such payments shall not exceed \$5,000.00 dollars. Such payment will not be considered deferrable compensation for the Tax Deferred Retirement Program (TDRP). The Company may elect to make such payments in intervals over the course of the first year of employment at its discretion.

The Company maintains the right to discontinue the payment of such hiring bonus at its discretion. The Company will meet with the facility Grievance Committee Chairman to explain any offering made under this letter. The agreement of the Union will be required should the Company wish to provide a bonus in excess of the limit described above or otherwise modify this Letter of Agreement.

Any employee receiving such hiring bonus payment whose employment is terminated within one year of the date of hire, may be required to repay such hiring bonus payment. Any employee whose employment is terminated after the effective date of this Letter of Agreement, and subsequently rehired, will not be eligible for any hiring bonus payment under this Letter of Agreement.

This letter shall become effective on the Monday following the ratification of the 2023 Central Agreement and shall remain in effect until January 1, 2028.

Letter of Agreement No. 32 SUBSTANCE ABUSE PROGRAM The Company and the Union had extensive discussions regarding the need to address the problems of alcoholism and drug addiction as it impacts the troubled worker, workplace, and fellow employees. Recognizing the need to promote healthy and safe lifestyles; the workplace dangers that can be created through the use of illegal drugs, alcohol or misused prescription drugs; the Company's long standing prohibition on possession or use of such substances on Company property; and the Company's status as a Federal contractor that requires it to maintain a drug-free workplace, the parties have agreed to implement the following process design:

- 1. It will include provision for random, reasonable suspicion, return from any leave of absence and recall from indefinite layoff testing.
  - a. "Reasonable suspicion" is defined as those circumstances, that would cause a reasonable person to believe that the employee is demonstrating signs of impairment. "Signs of impairment" include, but are not limited to, such things as the employee having difficulty maintaining balance, slurred speech, erratic or atypical behavior, or otherwise appearing unable to perform his job in a safe manner. The following items will not be used as the exclusive basis for "reasonable suspicion" testing: poor attendance, untidy appearance, failure to meet production or quality standards, or merely being involved in an accident.

In the event that reasonable suspicion testing is requested, the member of management requesting the testing will consult with a member of management who has received training on recognizing the symptoms and signs of drug and alcohol use to provide corroboration regarding the need for testing.

- b. The number of employees selected for random testing, on an annual basis, will not exceed 12% of the employees at any business unit.
- c. Employees will be subject to testing upon return from any leave of absence and recall from indefinite layoff greater than 30 days.
- 2. It will utilize drug thresholds and collection procedures consistent with those of either the Department of Transportation or the Department of Health and Human Services. Notwithstanding the above, marijuana will be excluded from random, return from leave of absence and recall from indefinite layoff testing.
- 3. Employees who test "in compliance" will not lose any pay for regularly scheduled or overtime hours due to testing procedures or related-travel time, there will be no impact on future job opportunities, and no written record of testing will be included in the employee's work history folder.
- 4. Employees who are "not in compliance" as a result of any random, reasonable suspicion (for alcohol this will include any BAC test result of .08 or higher), return from leave of absence or recall from indefinite layoff testing will be discharged upon the first "not in compliance" test.

Notwithstanding the above, employees who are "not in compliance" solely for the use of marijuana or alcohol (this will only apply to any BAC test result of .04 to .079) as a result of reasonable suspicion testing will have one opportunity for treatment and rehabilitation at Company expense under the terms of a "Last Chance Agreement", will be on a disciplinary suspension for 30 days pending successful completion of a treatment program

and an "in compliance" return-to-work test. The opportunity for treatment and rehabilitation under the terms of a "Last Chance Agreement" will be extended only to those employees who have over two years of seniority at the time the employee is determined to be "not in compliance" as a result of reasonable suspicion testing. Employees with less than two years of seniority will be discharged upon the first "not in compliance" test.

In the event alternate testing methods are developed to measure impairment as a result of marijuana use and those methods are accepted by the Department of Transportation or law enforcement in the State of Illinois, the Company may, at its sole discretion, implement such testing method. If implemented, employees who are "not in compliance" as a result of reasonable suspicion testing for marijuana will be discharged upon the first "not in compliance" test.

For purposes of this section, "at Company expense" means the appropriate Company-sponsored group medical plan will pay the costs of treatment and rehabilitation, but only if the employee is enrolled in the Company-sponsored group medical plan and only to the extent owed under the plan.

- 5. Once an employee signs a Last Chance Agreement, the Union will not file or pursue a grievance on behalf of employees if they test "not in compliance."
- 6. Employees who successfully return to work will be subject to targeted, unannounced testing for a reasonable period after their return not to exceed three years.
- 7. The process will include the involvement of the Employee Assistance Program (EAP) and Company medical services, but will not preclude the use of a professional health care provider or laboratory in the execution of the procedures and processes to be used.
- 8. It will comply with all applicable privacy practices for any records related to the testing process.
- 9. This process does not impede in any way the Company's right to take disciplinary action at any time, up to and including discharge.
- 10. An employee who voluntarily enters a treatment program after submitting to a test, but prior to receiving the results of that test, will be subject to the same limited opportunity for treatment and rehabilitation as described above in Section 4.
- 11. Unless selected for random or reasonable suspicion testing, the process will have no impact on an employee who voluntarily presents themselves to the EAP or Company medical services for assistance with a drug or alcohol problem.
- 12. The Company will provide employees with information and training concerning this Letter of Agreement, including the reasons for the program, available EAP support, methods of referring employees, and testing conditions.
- 13. Employees who refuse to participate in testing will be subject to discharge.

14. Employees who are currently on a Last Chance Agreement for violating this Letter of Agreement, solely as it applies to the use of marijuana, will be deemed to have met the obligations of the Last Chance Agreement, upon ratification of this agreement.

## Letter of Agreement No. 33 TEAM LEAD AND NON-TRADITIONAL ROLES

During 2023 Negotiations the Union expressed concerns related to the Company's use of Team Lead and Non-Traditional roles.

Team Leads, under the guidance and supervision of their manager, are responsible for coordinating the activities of the team. While the manager has ultimate responsibility for operational results, the Team Lead plays a key role in guiding team members toward the achievement of established people, quality, velocity and cost (PQVC) metrics. Team Leads are expected to carry out their responsibilities in the spirit of teamwork and co-operation, not through direct supervision.

The Company continues to have the right to select and remove both Team Lead and Non-Traditional employees in its sole discretion.

It is understood that neither a Team Lead nor Non-Traditional assignment are part of a management training program, and as such are not a guarantee of future placement in a job outside of the bargaining unit.

## BASIC HOURLY RATES OF PAY APPENDIX A

(Applicable to employees in Local Unions Nos. 751, 1872 (except for employees hired or recalled after April 6, 1992 who are assigned to jobs in Labor Grades 1 or 2), 974 (except for employees hired or recalled after March 23, 1998 who are assigned to jobs in Labor Grades 1 or 2 in Morton), and 2096.)

1	16.69	16 04	17 10	17 44	17.60	17 04	19 10
2	19.56	19.81	20.06	20.31	20.56	20.81	21.06
3	22.25	22.60	22.95	23.30	23.65	24.00	24.35
4	25.56	25.90	26.20	26.50	26.80	27.10	27.40
5	29.41	29.76	30.11	30.46	30.81	31.16	31.51
6	34.36	34.67	34.98	35.29	35.60	35.91	36.22

# BASIC HOURLY RATES OF PAY APPENDIX A

(Applicable to employees in Local Union No. 1872 hired or recalled after April 6, 1992 who are assigned to jobs in Labor Grades 1 or 2.)

1 16.69 16.86 17.02 17.19 17.36 17.52 17.69 17.86 18.02 18.19

2 20.07 20.32 20.57 20.82 21.07 21.32 21.57 21.82 22.07 22.32

# BASIC HOURLY RATES OF PAY APPENDIX A

(Applicable to employees hired or recalled after March 23, 1998 who are assigned to jobs in Labor Grades 1 or 2 in Morton.)

1 16.69 16.86 17.02 17.19 17.36 17.52 17.69 17.86 18.02 18.19

2 19.06 19.31 19.56 19.81 20.06 20.31 20.56 20.81 21.06 21.31

# SUPPLEMENTARY BASE RATE SCHEDULE APPENDIX A

(Applicable to employees in Local Union 974)

6T27 Machining Technologist 6T22 Metrology Technologist	6-1	6-2	6-3	6-4	6-5	6-6	6-7	6-7 times 1.0217		
6T24 Fabrications/Process Technologist										
6T25 Weld Technologist	C									
6T26 Heat Treat Process Techno	ologist									
(Technical Center and Peoria Pro	oving G	round)								
2X10 Machine Repair Apprentic	e 609	65%	70%	75%	80%	85%	90%	95% 6-1*		
2X36 Maintenance Mechanic Ap	prentic	e 60% 70	)% 80%	90%	6-1*					
2X12 Toolmaking Apprentice	60%	70% 8	0% 90	% 6-1	1*					
2X13 Machine Shop Trainee	60%	70% 8	0% 90	% 3-1	1*					
2X15 Machinist Apprentice	60%	65% 7	0% 75	% 80	% 85	% 90%	95%	3-1*		
2X16 Welding Trainee	60%	70% 8	0% 90	% 3-1	1*					
2X18 College Graduate Trainee	4-4									

2X19 Cooperative Trainee	60%	65%	70%	75%	80%	85%	90%	4-4
2X20 Associate Degree Trainee	60%	65%	70%	75%	80%	90%	4-3	
2X18 College Undergraduate Tr	ainee	60%	65%	70%	75%	80%	90%	4-3
2X45 Electrician Trainee	60%	70%	80%	90%	6-1*			
2X42 Maintenance Mechanic Tr	ainee	60%	70%	80%	90%	6-1*		
2X44 Toolmaking Trainee		60%	70%	80%	90%	6-1*		
2X46 Foundry Machine Repair	Γraine	e	60%	70%	80%	90%	6-1*	
2X31 Electrician Apprentice		60%	70%	80%	90%	6-1*		

<sup>\*</sup>Minimum Graduating Rate

# Supplementary Base Rate Schedule Appendix A (Applicable to employees in Local Union No. 751)

2X20 Associate Degree Trainee	60%	65%	70%	75%	80%	90%	4-3		
2X18 College Graduate Trainee	4-4								
2X19 Cooperative Trainee	60%	65%	70%	75%	80%	85%	90%	4-4	
2X31 Electrician Apprentice	60%	70%	80%	90%	6-1*				
2X13 Machine Shop Trainee	60%	70%	80%	90%	3-1*				
2X15 Machinist Apprentice	60%	65%	70%	75%	80%	85%	90%	95%	3-1*
2X12 Toolmaking Apprentice	60%	70%	80%	90%	6-1*				
2X16 Welding Trainee	60%	70%	80%	90%	3-1 *				
2Xl0 Machine Repair Apprentice	60%	65%	70%	75%	80%	85%	90%	95%	5-1*
2X40 Maintenance Specialist	60%	65%	70%	75%	80%	85%	90%	95%	6-1*
Apprentice									
2X36 Maintenance Mechanic	60%	70%	80%	90%	6-1*				
Apprentice									

<sup>\*</sup>Minimum Graduating Rate

# Supplementary Base Rate Schedule Appendix A (Applicable to employees in Local Union No. 2096)

2X31 Electrician Apprentice	60%	70%	80%	90%	6-1*				
2X18 College Graduate Trainee	4-4								
2Xl3 Machine Shop Trainee	60%	70%	80%	90%	3-1*				
2X15 Machinist Apprentice	60%	65%	70%	75%	80%	85%	90%	95%	3-1*
2X12 Toolmaking Apprentice	60%	65%	70%	75%	80%	85%	90%	95%	6-1*
2X36 Maintenance Mechanic	60%	70%	80%	90%	6-1*				
Apprentice									

<sup>\*</sup>Minimum Graduating Rate

### Memorandum of Agreement Employees Hired Prior to January 10, 2005

The 2023 Central Agreement is amended as hereinafter provided for employees hired prior to January 10, 2005. These modifications shall be applicable to such employees for the duration of the 2023 Central Agreement. This Memorandum of Agreement shall not be applicable to any employee who upon the expiration of the 1998 Central Agreement was subject to the provisions of Letter of Agreement No. 26 of the 1998 Central Agreement.

- (1) Section 11.2 of the 2023 Central Agreement for regular employees shall be deemed to read as follows:
  - (11.2) Seniority in one of the business units specifically set forth in Section 2.1 of this Agreement shall be measured from an employee's last date of hire into the bargaining unit or the unit which currently constitutes such bargaining unit.

If an employee who has never worked in a bargaining unit as described above transfers into any such bargaining unit, either at his request or the Company's request, his "date of hire" will be considered the date of such transfer. Seniority with respect to any one such bargaining unit is broken for the following reasons only:

- 1. If the employee quits.
- 2. If the employee is discharged and not reinstated.
- 3. If the employee refuses, while laid off, to accept a recall to work (issued by the Company), or fails to report for work in accordance with such recall within ten calendar days after the mailing by certified letter mailed to the employee's last known address. The Company will grant an extension to the ten-day period if, prior to the expiration of the ten-day period, satisfactory reason is given for the employee's inability to immediately report for work.
- 4. If an employee has been laid off for a period greater than his accumulated seniority at time of layoff. (However, this shall not apply during the first two years of layoff.)
- 5. If an employee fails, unless satisfactory reason is given, to report for work by his fourth scheduled workday following a temporary layoff under the provisions of Section 13.6 of this Agreement or any other layoff where the return to work date has been announced prior to the beginning of such layoff and such employee has received written notice of such date.
- 6. If an employee (other than a laid off employee whose seniority has not been broken under the foregoing provisions of this Section 11.2) accepts employment at a location that is not within the geographical limits of the bargaining unit in which he possesses seniority.

The seniority of an employee in a bargaining unit shall be applied as provided in

the Local Agreement covering that bargaining unit.

- (2) Except for employees who are assigned to Labor Grades 1 and 2 at the Caterpillar Logistics Services facilities, the Appendix A of the 2023 Central Agreement for regular employees shall be deemed to read as follows:
  - 1 22.31
  - 2 22.73
  - 3 23.55 23.88 24.35
  - 4 25.56 26.20 26.80 27.40
  - 5 29.41 30.11 30.81 31.51
  - 6 34.36 34.98 35.60 36.22

The Appendix A of the 2023 Central Agreement for regular employees who are assigned to Labor Grades 1 at the Caterpillar Logistics Services Facilities shall be deemed to read as follows:

- 1 18.67
- (3) The Appendix A Supplementary Base Rate Schedule, (applicable to employees in Local Union 974) for employees at the Technical Center and Peoria Proving Ground in the identified job classifications shall be deemed to read as follows:

Machining Technologist 5-1 6-1 6-2 6-3 6-4 6.4 times 1.0217 Metrology Technologist Fabrications/Process Technologist Weld Technologist Heat Treat Process Technologist

(4) The provisions of Section 18.1 (E), Section 18.1 (F), Section 18.1(G), and Section 18.1(H) of the 2023 Central Agreement shall not be applicable to employees who are covered by this Memorandum of Agreement and whose rate of pay is above the corresponding rate of pay for employees hired on or after January 10, 2005.

# CENTRAL AGREEMENT

between

#### CATERPILLAR INC.

and the

**UAW** 

March 1, 2023

This booklet also contains the

LOCAL SUPPLEMENT

for the

### **DECATUR FACILITY**

and

# **LOCAL UNION 751**

March 1, 2023

# CONTENTS

A(! -1 - 1		
Article 1		
Purpose*		
Article 2		
Recognition	1	
Article 3		
Union Security	2	
Article 4		
Representation	3	
Article 5		
Grievance Procedure	5	
Article 6		
Arbitration*		
Article 7		
Hours of Work and Overtime		8
Article 8		
Safety	13	
Article 9		
Vacation and Separation Pay*		
Article 10		
Skilled Trades Groups*		
Article 11		
Seniority	14	
Article 12		
Job Openings	16	
Article 13		
Downgrade, Layoff, Recall		18

Article 14	
Leave of Absence	18
Article 15	
Salary Plan*	
Article 16	
General	19
Article 17	
Training*	
Article 18	
Wages*	
Article 19	
Group Seniority	20
Article 20	
Duration	26
Letter of Agreement #1 – Service Claim Work	27
Letter of Agreement #2 – Top Rate for Employe	ees Placed During Reductions in Force 27
Letter of Agreement #3 – Alteration of Wage R	ates or Job Descriptions 27
Letter of Agreement #4 – Safe Hooking Proced	ure 28
Letter of Agreement #5 – Line Spacing Procedu	re 28
Letter of Agreement #6 – Scheduling Work Dur	ring Plant Vacation Shutdown 29
Letter of Agreement #7 – Apprentice Overtime	30
Letter of Agreement #8 – Grievance Committee	e AWS Provision 30
Letter of Agreement #9 – Non-Traditional Lists	31
Letter of Agreement #10 – Exhibit Books	_
Exhibit B - Skilled Trades Groups	31
Exhibit C - NIO/IO Groups	33

<sup>\*</sup>No Local Sections

#### **AGREEMENT**

This Agreement entered into and concluded at Decatur, Illinois, on this 1<sup>st</sup> day of March, 2023 by and between Caterpillar Inc., Decatur, Illinois, hereinafter designated and referred to as the "Company," and International Union, United Automobile, Aerospace and Agricultural Implement Workers of America and its affiliated Local Union 751, thereinafter designated and referred to as the "Union" through their duly authorized representatives, do hereby agree as follows.

# ARTICLE 2 RECOGNITION

(2.1) The Unit appropriate for collective bargaining purposes is the unit covered by the certification of the National Labor Relations Board dated September 26, 1955, in Case No. 13-RC-4494 and Case No. 13-RC-4508, namely:

all production and maintenance employees of the Company's Decatur, Illinois, Plant but excluding all employees in the job classification of: Clerk, File; Clerk, Routine; Messenger, A; Assistant, Laboratory; Clerk, General; Comptometer, B; Key Punch Operator, Numerical; Sorter Operator, IBM; Teletype Operator; Typist; Auxiliary Mach. Operator, IBM, B; Key Punch Opr., alphabetical; Switchboard Operator; Xerox Operator; Auxiliary Mach. Opr., IBM, A; Blueprint Machine Operator; Clerk, Advanced; Clerk, Blueprint Control; Clerk, Car Dispatch; Clerk, Production Accounting; Comptometer, A; Multilith Operator; Receptionist, A; Safety Cart Operator; Stenographer; Tabulating Mach. Opr., IBM, B; Chauffeur; Clerk, Production Acctg., General; Detailer; Locksmith; Secretary- Stenographer, B; Tabulating Mach. Opr., IBM, A; Technician, Metallurgical; Clerk, Accounting, Jr.; Clerk, Layout; Clerk, Safety Store; Coordinator, Duplicating; Investigator, Accident; Nurse, Industrial; Secretary-Stenographer, A; Bookkeeper, General Ledger; Coordinator, Insurance; Secretary; Chemist, Metallurgical; Clerk, Accounting; Clerk, Cutting Tool Records; Clerk, Inbound Shipments; Clerk, Outbound Shipments; Clerk, Stock Control; Coordinator, Equip. & Communications; Detailer, Sr.; Draftsman, Layout; Technologist, Physical; Analyst, Stores Requirements; Clerk, Engr. Release & Change; Clerk, Follow-up; Draftsman; Inspector, Safety; Releaser, Production; Accountant,

Accounts Payable; Clerk, Accounting, Sr.; Technician, Plant Engineering; Technician, Punched Card Acctg. Sys.; Analyst, Job; Coordinator, Scheduling; Designer, Tool, B; Engineer, Timestudy; Interviewer; Planner, Layout; Releaser, Sr.; Scheduler, Factory; Accountant, Projection; Analyst, Rate; Designer, Tool, A; Engineer, Processing; Clerk, Prime Products Follow-Up; Clerk, Weight; and plant protection employees and professional and supervisory employees as defined in the Act, as amended.

#### **ARTICLE 3**

#### **UNION SECURITY**

- (3.1) "Authorization for Dues Check-Off" will be placed in effect by the Company, in accordance with Article 3 of the Central Agreement, during the next month following delivery to the Company, if such assignment is delivered not later than the twenty-fifth day of the month.
- (3.2) The Company shall submit a check for the amount of deductions made, in accordance with the provisions of Section 3.2 of the Central Agreement, accompanied by a statement listing the names of the employees and the amount of deductions made, payable to the Financial Secretary of the Union.
- (3.3) The Company recognizes and will not interfere with the right of employees to become members of the Union, and will not discriminate against employees because of their Union affiliation or legitimate Union activity. The Union agrees that there shall be no Union activity on Company time unless such activity is permitted elsewhere in this Agreement, or the Company grants permission therefor in writing.

#### **ARTICLE 4**

#### REPRESENTATION

(4.1) In order to provide a system of Union representation for the settlement of grievances, the representation structure will be as follows:

One Steward shall represent the employees under each Supervisor. The Steward shall be an employee within the group he represents.

There shall be a Plant Grievance Committee composed of six (6) members, one of whom shall be designated as the Chairman of the Grievance Committee. One Committeeman shall represent Assembly and Test Specialists and Fabrication Specialists on first shift. One Committeeman shall represent all other employees on first shift. One Committeeman shall represent Assembly and Test Specialists and Fabrication Specialists on second shift. One Committeeman shall represent all other employees on second shift. One Committeeman shall represent employees on third shift. Each Committeeman shall be elected from among the employees whom he/she represents.

The Union may designate a number of Alternate Committeemen equal to the number of Committeemen on such grievance committee. Each Alternate Committeeman shall be selected from among the employees whom he/she represents. The Union will provide the business unit Labor Relations Manager with a list of all Alternate Committeemen for that business unit and shall notify the business unit Labor Relations Manager of any changes to this list as promptly as possible.

When it is necessary for a Committeeman to be absent for a full day or more, such Committeeman may be replaced by an Alternate Committeeman, provided the business unit Labor Relations Representative is notified no later than Thursday of the week preceding the week in which the time off is to be taken. In the event an emergency or unforeseen circumstances prevent such advance notification, the Labor Relations Manager may elect to waive such requirement.

(4.2) In taking Step I, Stewards may, without loss of pay for regularly scheduled hours discuss a grievance (1) with the aggrieved employee, provided the aggrieved employee first informs his immediate supervisor of the specific issue the aggrieved employee believes the Company has violated (2) with the aggrieved employee's immediate Supervisor, and (3) if the grievance is not satisfactorily settled in Step I, with the appropriate Committeeman who would handle the grievance in Step II. However, in the absence of the Steward, or if for a temporary period there

is no Steward, a Steward from an adjacent area may represent the employee under the jurisdiction of the Steward involved. Where the individual aggrieved employee is the Steward, the aggrieved Steward will be represented by the appropriate Committeeman.

In taking Step II, Committeemen may, without loss of pay for regularly scheduled hours in accordance with Article 4 of the Central Agreement, discuss the grievance individually with each of the following: (1) the Steward, (2) the aggrieved employee, (3) the Supervisor/Superintendent or other Company designated representative. Where the individual aggrieved employee is a Committeeman, he may be represented in the second step by the Alternate Committeeman.

The Chairman of the Grievance Committee may elect to perform his duties as stated below while on Union Leave or on Union call-out for the duration of their term of office.

Those privileges granted to the Chairman of the Grievance Committee are as follows: (1) discuss grievances in the Final Step with the Company representatives as provided in Article 5.5 of this Local Agreement; (2) participate in joint investigations agreed to in a Final Step Grievance meeting.

An Alternate Committeeman shall have no contractual privileges, duties or responsibilities unless called upon to replace a Committeeman. While replacing a Committeeman, the Alternate Committeeman shall assume all of the privileges, duties and responsibilities of the Committeeman whom he/she is replacing. The Alternate Committeeman may exercise such privileges without loss in pay for regularly scheduled hours when replacing a Committeeman.

(4.3) If an employee working overtime desires to process a grievance related to an incident occurring during such overtime period, Union representatives handling such grievance shall, without loss of pay during such overtime period in accordance with Article 4 of the Central Agreement be permitted to process such grievance as provided in Article 5 of this Agreement.

(4.4) If a Steward or alternate committeeman is absent from work, the President of the Union (or his designated representative) may appoint an alternate who will be temporarily recognized following notification of such appointment, in writing, to the Labor Relations Department.

(4.5) It is understood that the President of the Union may serve as alternate to the fulltime Chairman, as provided in Section 4.2 of this Local Agreement. The President may elect to perform the duties of his office while on union leave or while on union call-out. He shall notify the Company in writing of his intended status within thirty (30) days of the effective date of this Agreement or the beginning of his elected term. His status shall remain in effect for the length of his term in that office.

#### **ARTICLE 5**

#### **GRIEVANCE PROCEDURE**

- (5.1) It is mutually desired that grievances be satisfactorily settled as quickly as possible. Whenever a grievance arises, an earnest effort will be made to settle such grievance in the following manner.
- (5.2) STEP I. The aggrieved employee or Steward shall present the grievance directly to the employee's immediate Supervisor at the time the grievance arises. If the employee has identified the issue for his/her Supervisor, upon request of the employee the Supervisor shall without undue delay send for the Steward for the purpose of (1) discussing his grievance with him prior to presentation to his Supervisor, and (2) presenting the grievance. The aggrieved employee shall be present at the time the grievance is presented to the immediate Supervisor, except when the aggrieved employee has requested his Supervisor to send for the Steward, in which event presence of the aggrieved employee is optional. The Supervisor shall render his decision within three working days from the date of referral to first step. The time limits for answering grievances as prescribed in the above procedure may be extended by mutual agreement between the parties.
- (5.3) STEP II. Grievances not satisfactorily adjusted in the first step shall, at a mutually agreed time within two working days after the date of appeal to this step be discussed between the appropriate area Committeeman as defined in Section 4.1 of the Local Agreement and the Superintendent or other Company designated representative who shall render a decision within two working days following such second step meeting. The time limits for answering grievances as prescribed in this section may be extended by mutual agreement between the parties.

- (5.4) Any grievance not appealed from a decision in Step I within five working days or Step II within ten working days of such decision shall be considered settled on the basis of the last decision and not subject to further appeal.
- (5.5) FINAL STEP. If the grievance is not satisfactorily adjusted in the second step, the Chairman of the Grievance Committee of the Union or his designated representative shall present the grievance to the Final Step of the grievance procedure in writing prior to the Final Step meeting. Each Thursday from 12:00 p.m. until 4:30 p.m., a meeting will be held between the Grievance Committee and Labor Relations. On the days of these meetings, the time from 12:00 p.m. until 1:00 p.m. shall be allotted to discuss issues of mutual concern. The balance of the meeting shall be used to discuss grievances that have been referred to the Final Step, provided there are grievances to discuss. The Grievance Committee shall have an opportunity to assemble in the Conference Room forty minutes prior to the Final Step Meeting.
- (5.6) Except as provided in Section 5.5 of the Central Agreement, to be processed under the above grievance procedure, a grievance must be presented in Step I within a reasonable period after the occurrence of the action asserted to be the cause of a grievance or within a reasonable period after the time the employee or Union could have been aware of such action asserted to be the cause of the grievance. In all cases, presentation of grievances within fifteen calendar days shall be considered within the above referred to "reasonable period." The time limits for answering grievances as prescribed in the above procedure may be extended by mutual agreement between the parties.
- (5.7) At the Final Step grievance meeting safety grievances, grievances concerning layoffs, current separations and indefinite disciplinary suspensions will be discussed first in chronological order of referral to Final Step. The balance of the grievances will be discussed in order of referral to Final Step.
- (5.8) The Company will prepare minutes of each Final Step Grievance Meeting. Such minutes will include (a) date of meeting, (b) names of individuals attending, (c) brief statement of grievances discussed, (d) summary of Union's contention in the event of failure to adjust, (e)

management's answer to each grievance. If there is no objection raised at the succeeding meeting

between the parties, the minutes shall be initialed by the parties and shall stand of record. Union

Statements of Unresolved Grievance may be delivered to the Plant Labor Relations Manager

within 30 calendar days of the date of exchange of such written minutes. However, this time will

not delay the time limits established in Article 6 of the Central Agreement.

(5.9) Meetings requested by the Union will not be held on Company time except as provided in

the handling of grievances. Final Step Grievance meetings and meetings called by Management,

however, will not result in loss of pay for regularly scheduled hours of those Union

representatives while in attendance at such meetings, provided that such Union representatives

are not on Union call out during that time.

**ARTICLE 7** 

HOURS OF WORK AND OVERTIME

(7.1) When overtime is scheduled, the opportunity to work such overtime shall be offered to

qualified employees, in the classification, in the area under the jurisdiction of each Supervisor,

who normally perform such work, beginning with the employee or employees having the least

amount of charged overtime hours. In the event two or more employees have an equal number

of charged overtime hours, the opportunity shall be offered first to the most senior employee.

In the event employees who are eligible as provided above are not available, the overtime will

be assigned in the following order:

a. to other qualified employees on the shift under the jurisdiction of the supervisor which the

overtime work is delegated.

b. to qualified employees, in the classification, on the shift, under the jurisdiction of the second

level supervisor.

c. to any qualified employee.

Caterpillar: Confidential Green

(7.2) Whenever an employee is temporarily or permanently assigned to another overtime

allocation unit, he shall immediately be charged with the average of the overtime hours between

the employee with the least and the employee with the most overtime hours.

(7.3) When the opportunity arises for an employee to be offered an overtime assignment, the

hours available at such time shall immediately be charged against him as hours worked, provided

such overtime hours were available in accordance with the provisions of Section 7.12 of the

Central Agreement. However, this shall not apply to any employee on a regular scheduled

vacation. For purposes of this Section, "hours available" shall be interpreted to mean pay hours.

(7.4) When an employee is brought in to work in a job classification on a job which the Company

requires to be performed on overtime, that employee may be asked to perform work in other job

classifications/allocations provided that such work in such other job classifications/allocations

does not exceed 25% of the time he is required to work on that particular day, and provided such

employee does not perform work in a classification in a higher labor grade than the labor grade

to which such employee was assigned for the overtime opportunity.

(7.5) A list of those receiving overtime charges (including totals) in the allocation unit shall be

given to the Steward and displayed at the Supervisor's desk on the scheduling day. If such list is

presented to the Steward on the regular scheduled workday immediately prior to overtime being

worked on a regular scheduled workday or on the second regularly scheduled workday prior to

overtime being worked on a Saturday, Sunday or holiday, and the employee or Union fail to call

an error in overtime assignment to the Company's attention at least two hours in advance of the

end of the employee's last shift before the date upon which the overtime is to be worked, the

Company shall not be liable for such error.

Stewards are granted the privilege of taking a reasonable amount of time to check the accuracy

by recording of overtime lists provided by the Company. This privilege shall be subject to the

provisions of Section 4.2 of the Central Agreement.

(7.6) OVERTIME GROUND RULES

Charge Overtime Pay Hours:

Caterpillar: Confidential Green

- 1. When an employee is asked to work overtime in his own overtime allocation unit at least two regularly scheduled workdays prior to a Saturday, Sunday, and/or holiday overtime assignment.
- 2. When an employee is asked to work in his own overtime allocation unit on a regularly scheduled workday at least on the regularly scheduled workday prior to the day upon which the overtime is to be worked.
- 3. Whenever an employee accepts any overtime assignment.
- 4. When an employee has left work or is absent, for any reason, at the time overtime is properly scheduled.
- 5. Although overtime pay hours made available are charged against an employee when properly scheduled, when an employee works less overtime hours than initially scheduled due to machine failure, lack of work, etc., adjust overtime pay hours to hours actually worked. Do not adjust if reason for starting late and/or leaving early results from employee's own action including leaving when an alternate assignment is available, or disciplinary action.
- 6. Whenever an employee works any overtime.

#### Do Not Charge Overtime Pay Hours:

- 7. When an employee refuses an overtime assignment within his own overtime allocation unit and such offer was not made two regularly scheduled workdays prior ... in the case of Saturday, Sunday, and/or holiday, and ... one regularly scheduled workday prior, in the case of regularly scheduled workdays.
- 8. When an employee refuses an overtime assignment outside his own overtime allocation unit.
- 9. When an employee is unable to accept an overtime opportunity for the sole reason of fulfilling a jury duty obligation or training obligation with a reserve component of the Armed Forces or the National Guard, provided the employee notifies his supervisor of such obligation.
- 10. When management determines the employee is not qualified to perform the work assignment for the overtime opportunity without being trained. Should an employee be deemed not qualified under this ground rule he shall be notified prior to the overtime in question being worked.

- 11. When employees are on temporary layoff scheduled under the provisions of Sections 13.2 or 13.6 of the Central Agreement.
- 12. Whenever a supervisor is assigning overtime and determines that such overtime, if worked by the employee, would violate the state Six- Day Week Law, he should skip the employee in rotation. The overtime hours should not be charged against the employee.
- 13. When an employee is not available to accept an overtime opportunity because he is assigned to another shift during the scheduled overtime hours.

#### Other:

- 14. Whenever an employee has been scheduled to work overtime and there is one or more days between the scheduling of such overtime and the time such overtime is to be worked, it is the employee's responsibility, in the case of absence on the day prior to such overtime, to notify the Company as to whether or not he will be available for such overtime. Such notification is to be given within four hours following the beginning of that employee's shift preceding the scheduled shift of overtime. If the employee fails to notify the Company accordingly, the overtime will, if necessary, be rescheduled.
- 15. Whenever an employee has been scheduled to work overtime on a regularly scheduled workday, it is the employee's responsibility in the case of tardiness and/or absence on such day to notify the Company of his intent to work such overtime no later than forty-five minutes following the commencement of such overtime. If no word is received during this period, the supervisor will, if necessary, call in a replacement.
- 16. An employee must accept or reject an overtime opportunity at the time such overtime is offered. If an employee rejects such overtime opportunity, no further offer for the same overtime opportunity shall be made to that employee. "Same overtime opportunity" shall be defined as an offer for the same span of hours.
- 17. The term "regular scheduled vacation" shall, for the purpose of applying the provisions of Section 7.4 (and only Section 7.4), be interpreted to mean one or more seven consecutive calendar days, including holidays, that commence on Monday. In the case of employees assigned to third shift the week shall begin concurrent with their normal start time for their Monday. Under this ground rule, employees on vacation are not to be charged for any pay hours related to an overtime opportunity that involves working

- overtime during their vacation or for any overtime opportunity involving scheduling overtime during such vacation.
- 18. If an employee working on overtime is unable because of machine breakdown, material shortage or other unforeseen circumstances to complete the assignment for which he/she was originally scheduled, the Company may, at its discretion, assign such employee to perform work in any classification, in which event the Company assumes no liability to other employees.
- 19. In the event an employee has been scheduled to work an overtime assignment in accordance with Article 7 of this Agreement and that employee fails to report for work on such overtime assignment, another employee in any classification may perform the scheduled work assignment of the absent employee in which event the Company assumes no liability to other employees.
- 20. If an employee is absent on scheduling day (proper notice) he/she may notify his/her supervisor that he/she desires to be scheduled for overtime. Such request will be honored provided he/she works or notifies the supervisor within the first 4 hours on the regularly scheduled work day immediately preceding the overtime day.
- 21. Employees who are absent on the scheduling day (proper notice) shall be reconsidered in rotation order for any additional overtime work within their allocation unit once the allocation unit is exhausted, but before such work is offered to employees in another allocation unit. If an employee refuses such opportunity, his/her original charge and count shall stand.
- 22. In the event overtime is to be scheduled on a day outside of an employee's regular workweek schedule, it is understood and agreed between the parties that when such an overtime opportunity arises, the eligible employees within the allocation unit in which the overtime is required who are working on the day the overtime is scheduled will be offered the opportunity first. If additional employees are required to perform the available work, the appropriate call-in provisions will be implemented, beginning with employees in the overtime allocation unit in which the overtime is needed. Under the circumstances described above, refusals by either employees working or employees called will not be charged.
- 23. It is understood that second shift will have the opportunity to work up to four (4) hours over on Friday and third shift will have the opportunity to work up to four (4) hours early

- on their Monday without violating any other shift's overtime allocation. It is understood that under this ground rule second shift will have the opportunity to voluntarily work over, up to four (4) hours, on Friday and that third shift will have the opportunity to voluntarily work, up to four (4) hours, early on Monday.
- 24. Employees with approved vacation on Friday or Monday will not be offered or required to work overtime opportunities on the Saturday or Sunday adjacent to such vacation; nor will employees be charged for such opportunities. Notwithstanding the above, it will be each employee's individual responsibility to notify the supervisor on or before the scheduling day of their desire to work overtime in writing on a form supplied by the Company.

# ARTICLE 8 SAFETY

(8.1) It is agreed that the Union shall have a Union Safety Committee composed of one employee working on each shift and in each of the buildings of the Decatur Plant from each shift (provided there is a minimum of 50 employees regularly working in that building on that shift) to encourage the observation of safety rules and the furtherance of the safety program. The President of the local, or his/her designated representative, will be the Director of the Union Safety Committee. Each Union Safety Representative shall be appointed by the President of the local (or his designated representative) from the jurisdiction he represents. Members of this committee may not serve in the capacity of Steward or Grievance Committeeman while serving as a Union Safety Representative.

The Union Safety Committee may appoint one Union Safety Sub-Representative for each Superintendent's jurisdiction who will also encourage the observation of safe practices and the furtherance of a safety and sanitation program. A Union Safety Sub-Representative's jurisdiction shall be only the area for which his Superintendent is responsible. Union Safety Representatives and Union Safety Sub-Representatives may not serve in the capacity of Steward, or Grievance Committeeman.

A safety tour will be conducted monthly on each shift by a major product area manager, or his

designated representative, and the President of the Local Union, or his designated representative.

(8.2) As a condition of employment, employees shall wear approved safety glasses or goggles

in all work areas outside of the designated office and cafeteria areas. Employees who are required

to wear prescription glasses shall not be required to purchase prescription safety glasses.

However, such employees shall wear approved safety goggles or shields over such prescription

glasses in all work areas outside of the designated office and cafeteria areas.

(8.3) Replacement costs of safety glasses lost by an employee shall be solely the employee's

responsibility.

(8.4) The Company will keep minutes of each monthly Safety Committee meeting, a copy of

which will be provided to the Local Union President or his authorized representative within 15

calendar days.

(8.5) The Company will provide to the facility Union Safety Committee Chairman a copy of that

facility's report on OHSA Form #301 and the manhours worked at that facility during the period

covered by such report.

ARTICLE 11

**SENIORITY** 

(11.1) For the purpose of applying Article 13 of the Central Agreement and Article 12 of the

Local Agreement, seniority units shall apply by "Non-Interchangeable Occupational Groups"

(NIO) and an "Interchangeable Occupational Group" (IO) except as expressly provided

elsewhere in this Agreement. The Non-Interchangeable Occupational Groups are as follows:

1. Assembly

2. Inspection

3. General Machining

- 4. Materials
- 5. Lathes
- 7. Grinding
- 8. Welding
- 9. Facilities Engineering
- 14. Tool Grinding
- 15. Press Shop
- 16. Training
- 21. Gearing
- (11.2) In the event of a reduction in force or layoff, members of the Grievance Committee (Section 4.1 of the Local Agreement) shall be considered to have sufficient seniority in their respective jurisdiction to retain a job in their respective jurisdiction as long as there is a job remaining in their respective jurisdiction which they are able to perform without being trained.
- (11.3) Stewards who have served as Stewards for a continuous period of at least six months, shall, in the event of a reduction in force, be retained in their classification within their jurisdiction, provided they have sufficient seniority to retain that same classification in their seniority group, and further provided there is a job in that classification within their jurisdiction which they are able to perform without being trained. Stewards who have served as Stewards for a continuous period of at least six months, shall, in the event of a layoff, be retained at work as long as there is a job in their respective jurisdictions which they are able to perform without being trained, and shall be recalled to work in their respective jurisdictions which they are able to perform without being trained.

Stewards, who have served as Stewards for a continuous period of less than six months, shall, in the event of a reduction in force, be permitted to remain in their respective jurisdictions as long as there is a job which they are able to perform without being trained and on which their seniority holds them.

(11.4) A seniority list shall be submitted to the Union every three months. This list shall give each employee's name, identification number, seniority date, rate, shift, and shall be in seniority

date order within each Group, NIO, and IO. Additionally, once each year, a seniority list will be submitted to the Union in plant seniority order. This list shall provide each employee's name, identification number and plant seniority date.

(11.5) In applying the provisions of this Agreement, the seniority of employees in job classifications not included in Exhibit B who have the same seniority shall be determined by such employees' identification numbers. Where employees have identical seniority, the employee with the lowest identification number shall be deemed to have the greatest seniority.

(11.6) For purposes of this Article 11, "without being trained" means that the employee already possesses the knowledge and skill to perform the work, needing only a brief period to familiarize himself with the work to be performed and the equipment, if any, to be operated.

# ARTICLE 12 JOB OPENINGS

(12.1) If the employee is unable, for reasons of lack of skill or knowledge, to perform satisfactorily on a job to which he was assigned under the provisions of Article 12 of the Central Agreement, such employee will be placed in accordance with the following:

- a. He shall be placed on an open job which he can perform (within the seniority unit in which he is working).
- b. In the event placement under (a) above cannot be made, then the employee will be placed under the provisions of Section 13.5 of the Central Agreement.

A grievance arising under the provisions of this Section, may be presented in writing, directly to the Final Step of the grievance procedure.

(12.2) In the event an employee is unable to perform satisfactorily the job to which he is assigned solely as a result of a health defect, he shall be placed on an open job which he is able to perform in the seniority unit in which he is then working.

Employees who cannot be placed in accordance with the above paragraph may be placed on an

open job which he is able to perform in any seniority unit.

Any such employee who is so assigned to an open job because of a health defect for a period

over 30 calendar days will return to his former job to which he was regularly assigned

immediately prior to such medical placement when his medical restrictions are removed or

modified to the extent that would allow him to perform such job. An employee refusing to return

to his former job shall be separated as a quit. If upon return to his former job to which another

employee is assigned, the replacement employee shall be placed in accordance with Article 13

of the Central Agreement, subject to the circumstances prevailing at that time.

Any such employee assigned to an open job because of a health defect for a 30 calendar day

period or less will return to his former job when his medical restrictions are removed or modified

to the extent that would allow him to perform such job. If upon return to his former job to which

another employee was assigned, the replacement employee will return to his previous job as long

as his seniority permits.

Employees who are medically placed in accordance with Article 12 of this Agreement will no

longer continue to have rights or be required to return to their former job after successfully

bidding to another job.

**ARTICLE 13** 

DOWNGRADE, LAYOFF, RECALL

(13.1) The provisions of Sections 11.2 and 11.3 of this Agreement shall apply in all matters

relating to Section 13.6 of the Central Agreement with respect to the seniority status accorded

members of the Grievance Committee and Stewards.

**ARTICLE 14** 

LEAVE OF ABSENCE

Caterpillar: Confidential Green

(14.1) The Company will promptly return to work an employee who returns from a leave of absence of more than one month's duration, provided the employee notifies the Human Resources Department, in writing, of the date of return not more than ten days, and not less than three days, in advance of such return. Otherwise, the Company shall return the employee to work as soon as necessary arrangements can be made. Employees returning from a leave of absence will be placed on their former job subject to circumstances prevailing at the time of their return.

(14.2) In granting leaves of absence under the provisions of Section 14.10 of the Central Agreement, not more than nine such leaves shall be in effect at any time.

Also, in granting time off for the purposes of attending Union meetings under the provisions of Section 14.9 of the Central Agreement, not more than twenty employees will be granted time off during any one day. No more than 2 of these employees, excluding committeemen, will be from the same Supervisor. Exceptions to the number of employees granted time off can be made by mutual agreement.

# ARTICLE 16 GENERAL

(16.1) The Company shall provide eight (8) bulletin boards placed in convenient places to be used exclusively for Union notices. The notices will be furnished by the Union and posted by the Company. Notices delivered to the Company not later than noon on one workday will be posted within two regularly scheduled workdays.

(16.2) Any dispute concerning an increase in a production standard which has been increased after the date of this Agreement, or any dispute concerning the addition of machines to an operation, which addition of machines does not conform to past practice and occurs after the date of this Agreement, may be referred to the grievance procedure beginning with Step I, but shall not be subject to arbitration.

With respect to disputes arising under this Section 16.2 and arising out of changes to production standards, or additions of machines to operations, which addition of machines does not conform to past practice and occurs after the date of this Agreement, the provisions of Section 16.5 of the Central Agreement insofar as, but only insofar as, it refers to strikes, shall be inoperative with respect to disputes arising under this Section 16.2, and then shall be inoperative only with respect to a strike which (a) is officially called and sanctioned by the Union for the sole purpose of enforcing its demands with respect to a disputed matter covered by this Section 16.2, and (b) is commenced within a period of sixty days following the Company's final answer concerning such dispute in the Final Step of the grievance procedure.

(16.3) Exhibits A and B are incorporated as a part of this Agreement pursuant to Sections 18.4 and 10.1, respectively, of the Central Agreement.

(16.4) A supervisor will introduce an employee entering his area of jurisdiction on a permanent basis to the steward representing the employees under such supervisor.

(16.5) The Company will inform employees when "work sampling" his/her work area prior to the time such work sampling is to begin.

# ARTICLE 19 GROUP SENIORITY

(19.1) Within the product areas as defined in Section 7.1 of this Agreement there shall be established skilled trades groups (hereinafter referred to individually as "groups") as set forth in Exhibit B.

(19.2) Group seniority will be measured from an employee's last date of hire in a group, in any capacity, at the Company's present Decatur Plant, provided that for a graduate apprentice who enters a group, his group seniority within the group shall be the period of his apprenticeship plus his length of service in that group. ("Date of hire" for an employee moved to a group at the Company's present Decatur Plant, in any capacity, either at his request or the Company's request, shall be considered the date of such move.)

(19.3) Group seniority is broken for the following reasons only:

a. If the employee quits.

b. If an employee is discharged and not reinstated.

c. If the employee refuses, while laid off, to accept a recall to work (issued by the Company)

or fails to report for work in accordance with such recall within a period of ten calendar

days after the mailing of recall notification by the Company. The Company will grant an

extension to the ten-day period if prior to the expiration of the ten-day period a

satisfactory reason is given for the employee's inability to immediately report for work.

d. If an employee has been laid off from a group for a period greater than his accumulated

group seniority at time of layoff. (However, this shall not apply during the first two years

of layoff.)

e. If an employee fails, unless satisfactory reason is given, to report for work by his fourth

scheduled workday following a temporary reduction in force or any other layoff where

the return to work date has been announced prior to the beginning of such layoff and such

employee has received written notice of such date.

(19.4) An employee who moves from one group to another group, or from a job in a group to

another job outside of any group but within the Bargaining Unit, shall retain his group seniority

in the group from which he was moved, but shall not accumulate group seniority in the group

from which he was moved after the date of such move.

(19.5) If, under this Agreement, an employee is laid off from a group, the employee's group

seniority shall accumulate during the layoff in the group in which he last worked.

Upon an offer by the Company to the employee of the opportunity to return to a job within a

group, he may exercise the seniority he accumulated prior to such date plus any additional

seniority he acquired while working on a job within a group subsequent to such dates in accordance with the provisions of Section 11.3 of the Central Agreement.

(19.6) The provisions of Article 13 of the Central Agreement and applicable to employees covered by this Article 19 and, in such application, the term "seniority" where used in Article 13 of the Central Agreement, shall be deemed to mean "group seniority" and the term "NIO" where used in Article 13 of the Central Agreement, shall be deemed to mean "group." In the event of a reduction in force within a classification within a group, group seniority with that group will be used.

(19.7) In the event of a reduction in force or layoff members of the Grievance Committee who are within a group shall be considered to have sufficient group seniority in their respective group to retain a job in their respective jurisdiction as long as there is a job remaining in their respective jurisdiction which they are able to perform without being trained.

For purposes of application of these Sections 19.7 and 19.8, the definition of "without being trained" in Section 11.6 of this Agreement shall be applicable.

(19.8) If employed within a group, any Steward who has served as Steward for a continuous period of at least six months, shall, in the event of a reduction in force, be retained in his classification within his jurisdiction and group, provided he has sufficient group seniority to retain that same classification in his group, and further provided there is a job in that classification within his jurisdiction and group which he is able to perform without being trained.

If not so retained in his classification, any such Steward shall be continued at work on a job in his jurisdiction and group, but outside his classification, as long as there is any such job which he is able to perform without being trained.

If not so retained within his group, any such Steward shall be retained on a job within his jurisdiction, but outside any group, as long as there is any such job which he is able to perform without being trained, and, if not so retained in his jurisdiction, shall be returned to work in his

jurisdiction as soon as there is work in such jurisdiction which he is able to perform without being trained.

If employed in a group, any Steward who has served as Steward for a continuous period of less than six months shall, in the event of reduction in force, be permitted to remain in his jurisdiction and group as long as there is a job therein which he is able to perform without being trained and on which his group seniority holds him.

(19.9) The provisions of Section 13.1 of this Agreement shall be applicable to employees on layoff from a group, provided that for the purpose of this Section only, the term "seniority" shall be deemed to read "group seniority" and/or "seniority."

(19.10) In the application of the group seniority provisions of this Article, a layoff shall be considered as an interruption of employment caused by a reduction in force due to a reduction in regular operating schedules, as distinct from ones which are due to material shortage, equipment failure, power failure, labor dispute, plant rearrangement or retooling, or other circumstances which cause a temporary cessation or reduction in operations.

(19.11) No temporary layoff in a group shall exceed ten working days without consultation with the Union concerning the feasibility of applying the regular group seniority provisions set forth in Article 13 of the Central Agreement.

(19.12) A temporary reduction in force for the purpose of reducing, or avoiding an increase in, inventory of finished products because of sales prospects shall hereafter be subject to Article 13 of the Central Agreement.

(19.13) In applying the provisions of this Agreement, the group seniority of employees within a group who have the same group seniority date in that group shall be determined by their seniority as measured in Article 11 of the Local and Central Agreements, provided that as to any such employees who have the same seniority, as measured in Article 11 of the Local and Central Agreements, the employee with the lowest identification number shall be deemed to have the greatest group seniority.

(19.14) The provisions of Article 12 of the Central Agreement shall be applicable for filling job

openings within a group. For the purposes of such filling of job openings in a group, eligible

employees having group seniority within that group, shall be considered as having more seniority

than otherwise eligible employees who have bargaining unit seniority but no group seniority in

that group.

Job openings within a group shall not be filled by employees not having group seniority so long

as employees possessing group seniority and the basic qualifications are laid off from the

bargaining unit.

"Reassignment" applicable to employees with group seniority means the assignment of an

employee within a group (1) to a job opening in the same classification in the same group on the

same shift within that group but under a different Supervisor, or (2) to a job opening in a different

classification in that group on the same shift within that group, which has a maximum rate equal

to the maximum rate of the classification from which he is moved.

(19.15) If an employee in a group is unable, for reasons of lack of skill or knowledge, to perform

satisfactorily on a job to which he was assigned under the provisions of Article 12 of the Central

Agreement and Section 19.14 of this Agreement, such employee will be placed in accordance

with the following:

a. He shall be placed on an open job which he can perform in that group (within the Non-

Interchangeable Occupational group in which he is working) which is in a classification

with a maximum rate not more than one classification rate level lower than the maximum

rate of the job on which he was unable to perform satisfactorily.

b. In the event placement is not made under (a) then the employee will be placed under the

provisions of Section 19.6.

(19.16) In the event an employee is unable to perform satisfactorily the job to which he is

assigned solely as the result of a health defect, he shall be placed on an open job which he is able

to perform in the seniority unit in which he is then working.

Employees who cannot be placed in accordance with the above paragraph may be placed on an

open job which he is able to perform in any seniority unit.

Any such employee who is so assigned to an open job because of a health defect for a period

over thirty calendar days will return to his former job to which he was regularly assigned

immediately prior to such medical placement when his medical restrictions are removed or

modified to the extent that would allow him to perform such job. An employee refusing to return

to his former job shall be separated as a quit. If upon return to his former job to which another

employee is assigned the replacement employee shall be placed in accordance with Article 13 of

the Central Agreement, subject to the circumstances prevailing at that time.

Any such employee assigned to an open job because of a health defect for a thirty calendar day

period or less will return to his former job when his medical restrictions are removed or modified

to the extent that would allow him to perform such job. If upon return to his former job to which

another employee was assigned, the replacement employee will return to his previous job as long

as his seniority permits.

Employees who are medically placed in accordance with Article 12 of this Agreement will no

longer continue to have rights or be required to return to their former job after successfully

bidding to another job.

ARTICLE 20

**DURATION** 

(20.1) Subject to the ratification of this Agreement by the membership of the Union on or before

March 12, 2023 this Agreement shall be effective as of March 13, 2023 and shall remain in effect

until March 1, 2029 and thereafter from year to year unless sixty days prior to the date of

expiration either party gives notice to the other, in writing, that it desires to modify or terminate.

Caterpillar: Confidential Green

Agreed and subscribed to the date and year above written.

CATERPILLAR INC.

By: John Hubert Abby Damery

Derek Green Shataea Irons

Scott Smith

INTERNATIONAL UNION, UNITED AUTO-MOBILE, AEROSPACE AND AGRICULTURAL IMPLEMENT WORKERS OF AMERICA

By: JC Zimmerman Joe Morel

LOCAL NO. 751 OF INTERNATIONAL UNION, UNITED AUTOMOBILE, AEROSPACE AND AGRICULTURAL IMPLEMENT WORKERS OF AMERICA

By: Steve Collins Jason Hess

Blake Force Sam Hart

Brian Marts Rob Walters

#### Letter of Agreement No. 1

#### Service Claim Work

This will confirm our understanding in regard to parts and/or assemblies returned to the Decatur Plant under the Company's service claim procedure, that: (a) measurement requirements, with the exception of checks made with tape and/or scale, to determine compliance with blueprint specifications, (b) movement of parts and/or assemblies to and from a location where such parts and/or assemblies are to be measured, (c) disassembly work related to service claims, and (d) tagging parts for normal inplant routing will be performed by Bargaining Unit employees except as provided in Section 16.1 of the Central Agreement.

#### Letter of Agreement No. 2

Top Rate for Employees Placed During Reductions in Force

This will confirm our understanding that for the duration of this Agreement, the Company will give the top rate of the classification involved to a downgraded employee if either (i) he has previously, while working in that classification, earned at least the next to top rate of that classification, and since leaving that classification he has continued in the same line of work (i.e., product assembly, machine operation, welding, clerical, etc.) and has earned a rate at least equal to such top rate, or (ii) he previously worked at a rate higher than such top rate in a higher rated classification in which he acquired experience which would qualify him for such top rate in the classification to which downgraded.

#### Letter of Agreement No. 3

Alteration of Wage Rates or Job Descriptions

During the life of this Agreement the Company will not alter either the wage rates of, or the description of the work performed in, a job classification, except as the work in such job classification is significantly altered subsequent to the effective date of the new basic Local Agreement. If a dispute arises hereunder it shall be referred to the grievance procedure provided in the new basic Local Agreement, beginning in the Final Step thereof, but such dispute shall not be subject to arbitration.

Caterpillar: Confidential Green

If such dispute is not settled in such Final Step, the provisions of Section 16.5 of the Central Agreement shall be inoperative insofar as it refers to "strikes," and insofar as, but only insofar as, such dispute is concerned, for sixty days from the date on which the grievance was filed in such Final Step, and upon the expiration of such sixty days all provisions of said Section 16.5 shall be automatically effective if no strike action has as yet been taken.

Letter of Agreement No. 4

Safe Hooking Procedure

The Company and Union recognize the need for safe hooking when lifts are made by employees. To assure that hooking related to any crane operations is performed safely and properly, management will provide an employee experienced in hooking whenever a two-man lift is necessary.

Letter of Agreement No. 5

Line Spacing Procedure

The Company and Union recognize the desirability of having reasonable spacing between units on prime product paint and assembly lines to assure adequate working room and safe working conditions.

As a part of this concern for adequate working room and safe working conditions, if management determines the need to "unpin" prime product units from such a moving line, the Union Safety Sub-Representative for the area will be informed such "unpinning" is taking place and will be given an explanation of the action.

It is the intention of management to provide such spacing; therefore, whenever an employee feels the spacing on the above assembly lines constitutes a serious hazard to imperil his safety, he may make the condition known to his Supervisor. The Supervisor shall investigate such spacing promptly and report the finding and/or reason for the spacing to the employee. If the employee does not agree with the Foreman's position, the problem shall be turned over to the Union Safety

Caterpillar: Confidential Green

Representative for the area who is authorized to bring the spacing problem to the attention of the Major Product Area Manager or his designated representative, during that shift.

#### Letter of Agreement No. 6

Scheduling Work during Plant Vacation Shutdowns

In the event the Company decides to schedule work during a vacation shutdown, such work will be scheduled as follows:

- 1. Notify employees in the area where the work is to be performed that vacation work is available. Only employees who indicate their desire to work during the vacation period will be considered as volunteers for vacation work.
- 2. When the Company is able to determine its vacation work requirement, it will make offers to work to the volunteers as defined above. Offers to work will be made to the volunteers who can perform the work without being trained, who are in the classification and in the area under the jurisdiction of each day shift Superintendent. Such offers will be made beginning with the most senior employee in the above work group.
- 3. If there are not sufficient volunteers, the Company may require employees to work after considering those employees who can perform the work without being trained, regardless of shift, in the classification, in the area under the jurisdiction of each day shift Superintendent. If there are not sufficient volunteers under a day shift Superintendent, the Company will go to other qualified employees in the classification in the major product area as defined in Section 7.1 of this Agreement who have volunteered to work. When it is necessary to require employees to work, assignments will be made beginning with the least senior employee.
- 4. After the vacation work is scheduled, any employee entering the work group will be offered the opportunity to work during the vacation period only if additional employees are needed. Should an employee already scheduled to work during a vacation period leave the work group for any reason, the employee will be removed from the schedule.

5. Employees scheduled to work vacation will be put on a separate overtime list to be used for scheduling vacation overtime. All charged hours from the vacation overtime list will be added to an employee's regular overtime list.

When the vacation schedule is complete, employees who are scheduled to work during the vacation shutdown will be notified and a copy of the proposed schedule will be given to the Steward no later than ten (10) days prior to the vacation period. The Company will endeavor to schedule vacation work in accordance with this letter. If an employee calls to the Company's attention within five workdays after the scheduling is announced his belief that he should have been scheduled to work, the Company will investigate and, if warranted, place the employee on the vacation work schedule. However, the Company will accept no financial liability to any employee for any hours not worked by that employee.

#### Letter of Agreement No. 7

#### Apprentice Overtime

The following procedure shall be used to schedule overtime for apprentices assigned to Skilled Trades Groups: Apprentices in an overtime allocation unit may be scheduled to work overtime if such overtime involves work which they normally perform during their regularly scheduled work hours or if the work is determined to be of an educational benefit to the apprentice. At no time will more apprentices be scheduled than the number of working journeymen.

#### Letter of Agreement No. 8

#### **Grievance Committee AWS Provision**

During the 2004 negotiations, the parties discussed the importance of ensuring the Bargaining Chairman and Committeemen were available to represent employees and to perform their privileges and duties under the terms of Articles 4 and 5 of the Local Agreement. The parties agreed, therefore, that the Bargaining Chairman and Committeemen would not be required to accept an assignment on an alternate work schedule during the life of the Local Agreement unless the entirety of the Committeeman's jurisdiction has been assigned to such a work schedule. In the event that a Committeeman would otherwise be involuntarily assigned to an alternate work schedule, such Committeeman shall be placed in accordance with the provisions of Section 11.2

of the Local Agreement on a job within the Committeeman's respective jurisdiction on a regular work schedule.

#### Letter of Agreement No. 9

#### Non-Traditional Lists

The Company will provide the union a quarterly list of Non-Traditional employees and a description of their assignments.

#### Letter of Agreement No. 10

#### **Exhibit Books**

In 2023 bargaining, both parties discussed the importance of having updated Exhibit A, B, & C books. As a result of these discussions, the changes will be effective January 1, 2024, and the Company shall provide the Union with such updated books no later than April 1, 2024.

#### Exhibit B

#### **GROUP 1**

6T68 Boring Machine Operator, (Tool Room)	(6)
6T70 Precision Tool and Die Maker (Tool Room)	(6)(SC)
5T72 Mill Operator, (Tool Room)	(5)
4T85 Grinder Operator, (Tool Room)	(4)
4T75 Lathe Operator, (Tool Room)	(4)
5T51 Layout Developer (Tool Room)	(5)
4T71 Mill Operator, (Tool Room)	(4)
4T72 Gear Machine Operator, (Tool Room)	(4)
4T27 Tool Hardener (Tool Room)	(4)
4T55 Horizontal Shaper Operator (Tool Room)	(4)
4T66 Vertical Shaper Operator (Tool Room)	(4)
4T64 Planer Operator, (Tool Room)	(4)
4T86 Surface Grinder Operator, (Tool Room)	(4)
3T55 Drill Operator, (Tool Room)	(3)

3T52 Tool Grinding Specialist	(3)	
4T52 Tool Grinding Specialist	(4)	
3T45 Milling Machine Operator (TR)	(3)	
GROUP 2		
6E51 Electrician	(6)(SC)	
5E72 Refrigeration and Ventilation Equip		(5)(SC)
5E64 Carpenter	(5)(SC)	(=)(==)
5T62 Blacksmith	(5)(SC)	
3E51 Maintenance Specialist	(3)	
5E51 Electrician Specialist	(5)(SC)	
•		
GROUP 3		
6E90 Maintenance Mechanic	(6)(SC)	
6E81 Machine Repair Mechanic	(6)(SC)	
5E66 Maintenance Specialist	(5)(SC)	
GROUP 4		
4E68 Machine Repair Mechanic	(4)(SC)	
GROUP 6		
5E63 Tinner	(5)(SC)	
GROUP 11		
4E51 Maintenance Painter	(4)(SC)	
GROUP 12		
5E70 Maintenance Welder	(5)(SC)	
CDOVD 45		
GROUP 17		
5E55 Automotive Mechanic	(5)(SC)	

#### **Exhibit C**

#### NON-INTERCHANGEABLE OCCUPATIONS

(NIO)

- 1. Assembly
- 2. Inspection
- 3. General Machining
- 4. Materials
- 5. Lathes
- 7. Grinding
- 8. Welding
- 9. Facilities Engineering
- 14. Tool Grinding
- 15. Press Shop
- 16. Training
- 21. Gearing

#### NOTE:

SC = Specific Classification — Employee must have held the classification to be downgraded to it.

SJ = Specific Job — Employee must have held the specific job to be downgraded to it.

# NON-INTERCHANGEABLE OCCUPATIONS

(NIO)

#### Labor Grade

NIO 1 - Assembly

3A61 Assembly and Test Specialist (3)

3A60 Product Painter (3)(SC)

3A64 Assembly and Test Specialist (3)(SC)

4A64 Assembly and Test Specialist (4)(SC)

NIO 2 - Inspection

3Q61 Quality Specialist

4Q61 Quality Specialist	(4)		
NIO 3 - General Machining			
3M55 Drill Specialist	(3)		
3M65 Boring Specialist	(3)		
4M65 Boring Specialist	(4)		
3M71 Mill Specialist	(3)		
3M91 Mill, Drill & Bore Machine Specialist		(3)	
4M91 Mill, Drill & Bore Machine Specialist		(4)	
4M92 Manufacturing Systems Operator	(4)	)	
5M92 Manufacturing Systems Operator	(5)	)	
NIO 4 - Materials			
3P51 Materials Specialist	(3)		
3K12 MEO/Crane Operator	(3)		
NIO 5 - Lathes			
3M75 Lathe Specialist	(3)		
4M75 Lathe Specialist	(4)		
NIO 7 - Grinding			
3M85 Grinding Specialist	(3)		
NIO 8 - Welding			
3S61 Fabrication Specialist	(3)		
4S61 Fabrication Specialist	(4)		
NIO 9 Facilities Engineering			
5B78 Utilities Operator	(5)(SC)		
3B55 Machine Lubricator	(3)		
NIO 15 - Press Shop			
3S51 Metalworking Specialist	(3)		
4S51 Metalworking Specialist	(4)		
NIO 16 - Training *			
2X10 Machine Repair Apprentice (Special Rate	es)	(SC)	
2X31 Electrician Apprentice (Special Rates)		(SC)	
2X15 Machinist Apprentice (Special Rates)		(SC)	
2X40 Maintenance Specialist Apprentice (Spec	cial Rates)		(SC)
2X12 Toolmaking Apprentice (Special Rates)		(SC)	

2X13 Machine Shop Trainee (Special Rates)	(SC)	
2X16 Welding Trainee (Special Rates)	(SC)	
2X20 Associate Degree Trainee (Special Rates)	(SC)	
2X19 Cooperative Trainee (Special Rates)		(SC)
2X18 College Graduate Trainee (Special Rates)	(SC)	
2X36 Maintenance Mechanic Apprentice (Special Rate	s)	(SC)
NIO 21 - Gearing		
3M81 Gear Machine Specialist (3)		
4M81 Gear Machine Specialist (4)		

# INTERCHANGEABLE OCCUPATIONS

(IO)

Labor Grade	
2A61 Assembly and Test Specialist	(2)
1E51 Maintenance Laborer	(1)
2E51 Maintenance Specialist	(2)
2H51 Metal Treating Specialist	(2)
1M51 Shop Helper	(1)
2M55 Drill Specialist	(2)
2M65 Boring Specialist	(2)
2M71 Mill Specialist	(2)
2M75 Lathe Specialist	(2)
1P51 Materials Specialist	(1)
2P51 Materials Specialist	(2)
2S61 Fabrication/Metal Working Specialist	(2)